### **ORIGINAL**

### FLORIDA PROGRESS CORPORATION

Douglas E. Wentz Corporate Coursel

March 26, 1998

VIA OVERNIGHT MAIL

Ms. Blanca S. Bayó
Director, Records and Reporting
Florida Public Service Commission
2540 Shumard Oak Boulevard
Tallahasace, Florida 32301

Re: Docket No. 961216-El - Consummation Report.

Dear Ms. Bayó:

Pursuant to the requirements of the Commission's Order No. PSC-96-1521-FOF-EI issued December 16, 1996, as amended by Order No. PSC-97-0925-FOF-EI issued August 4, 1997, I enclose herewith for filing one executed original and three additional conformed copies of a Consummation Report dated March 26, 1998 for Florida Power Corporation (the "Company") in the above-referenced docket.

Please acknowledge your receipt of the Consummation Reports by date-stamping the enclosed copy of this letter and returning it to me using the enclosed, self-addressed and stamped envelope provided for this purpose.

ACK	envelope provided for this purpose.	un vicioso, ani assivano am na
AFA	工	Very truly yours,
APP		Douglas E. Wener
CAF		Douglas E. Wentz
CMU	Exclosures	The second secon
CTR		
EAG	cc: Kenneth E. Armstrong	
LEG	Rodney E. Gaddy Jack Shreve	
LIN	(each with enclo.)	
OPC		
RCH	RECEIVED FILED	
SEC	P:/Credit.Agt/Consum.96	
WAS	FPSC-BUREAU OF RE	CORDS

DOCUMENT NUMBER -DATE

3201 34th Street South 33711 • P.O. Box 14042 • St. Petersburg, Rorldo 33733 • 813/866-4274 • Fax 813/866-4021

### ORIGINAL

### PLORIDA PUBLIC SERVICE COMMISSION TALLAHASSEE, FLORIDA

CONSUMMATION REPORT

TO

**APPLICATION OF** 

FLORIDA POWER CORPORATION

FOR AUTHORITY TO ISSUE AND SELL

**SECURITIES DURING 1997** 

PURSUANT TO FLORIDA STATUTES, SECTION 366.04

AND RULE 25-8, FLORIDA ADMINISTRATIVE CODE

Address communications in connection with this Consummation Report to:

Kenneth E. Armetrong Vice President and General Counsel Florida Power Corporation P.O. Box 14042 St. Petersburg, FL 33733

Dated: March 26, 1998

DOCUMENT NUMBER - DATE

03638 MAR 27 %

FPSC-RECORDS/REPORTING

### FLORIDA PUBLIC SERVICE COMMISSION

APPLICATION OF PLORIDA POWER CORPORATION FOR AUTHORITY TO ISSUE AND SELL SECURITIES DURING 1997 PURSUANT TO FLORIDA STATUTES	))))	DOCKET NO. 961216-EI
SECTION 366.04 AND CHAPTER 25-8, FLORIDA ADMINISTRATIVE CODE	;	

The Applicant, Florida Power Corporation (the "Company"), pursuant to Commission Order No. PSC-96-1521-FOF-EI issued December 16, 1996, as amended by Order No. PSC-97-0925-FOF-EI issued on August 4, 1997 (collectively, the "Order"), hereby files its Consummation Report as required by the terms of such Order and pursuant to Rule 25-8.009, Florida Administrative Code.

cogeneration facility located in Polk County, Florida, for a purchase price of approximately \$450 million (the "Tiger Bay Transaction"). Initially, the Company financed the acquisition with a \$200 million loan under a short-term committed facility, and a \$250 million loan under a long-term committed facility, each with The Chase Manhattan Bank ("Chase"). On July 22, 1997, the Company entered into a Terms Agreement with a group of six underwriters (J.P. Morgan Securities Inc., PaineWebber Incorporated, First Chicago Capital Markets, Inc., Salomon Brothers Inc, Chase Securities Inc. and NationsBanc Capital Markets, Inc.) providing for the sale in an underwritten transaction of \$450 million of Medium-Term Notes (the "MTNs"), and distributed a Prospectus Supplement dated July 22, 1997 relating thereto. The proceeds from the sale of the MTNs were used by the Company to repay, on July 25, 1997, the funds borrowed pursuant to the short and long-term commitment letters with Chase. A special

Consummation Report relating to the Tiger Bay Transaction, with details concerning the loans and MTNs, and copies of the commitment letters, Prospectus Supplement and other related documents, was filed with the Commission in Dockst No. 961216-EI on September 22, 1997, and is incorporated herein by reference.

Apart from the securities issued in connection with the Tiger Boy transaction, the Company did not issue any medium-term notes, first mortgage bonds or other debt or equity securities during calendar year 1997, except for (i) commercial paper and (ii) notes that were delivered to various banks to evidence the extension of the Company's long and short-term revolving credit agreements (although no funds have as yet been borrowed by the Company under those notes and agreements).

The Company regularly issues commercial paper for terms up to but not exceeding 270 days from the date of issuence. The commercial paper is issued pursuent to a Commercial Paper Dealer Agreement dated December 22, 1988 with Merrill Lynch Money Markets Inc. (the "Merrill CP Agreement") and a Letter Agreement dated November 20, 1992 with First Chicago Capital Markets, Inc. (the "First Chicago CP Agreement"), each as amended as discussed below. The commercial paper is sold at a discount, including the underwriting discount of the commercial paper dealer, at a rate comparable to interest rates being paid in the commercial paper market by borrowers of similar creditworthiness. Given the frequency of these sales, it is not practicable to give the details of each issue. However, the Company's 1997 commercial paper activity can be summarised as follows:

### 1997 Commercial Paper Activity (\$ in thousands)

Commercial paper issued: \$4,013,200
Commercial paper metured: \$3,837,500
Average outstanding: \$217,272
Weighted average yield: 5.553%
Weighted average term: 22 days

As back-up for its commercial paper program, the Company previously executed (i) a Credit Agreement A with The Chase Manhattan Bank (National Association) ("Chase") as agent for the landers named therein, dated as of November 26, 1991, as amended, providing for shortterm loans to the Company in the aggregate principal amount not exceeding \$200,000,000 ("Credit Agreement A"), and (ii) a Credit Agreement B with Chase, as agent for the lenders named therein, dated as of November 26, 1991, as amended, providing for long-term loans to the Company in the aggregate principal amount not exceeding \$200,000,000 ("Credit Agreement B"). The Company now has executed a Second Amended and Restated Credit Agreement A dated as of November 18, 1997 and a Second Amended and Restated Credit Agreement B dated as of November 18, 1997, which extend the terms of Credit Agreements A and B to November 17, 1996 and November 30, 2002, respectively, and which increase the amount of the loans that may be outstanding at any time under Credit Agreement A from \$200,000,000 to \$300,000,000. No loans have as yet been made to the Company pursuant to the Credit Agreements. For accounting purposes, the Company classifies monies borrowed under, and commercial paper backed by, Credit Agreement B as long-term debt.

In connection with increasing the amount of loans that may be outstanding under Credit Agreement A, the Company also entered into a Letter Agreement dated November 18, 1997 with Merrill Lynch Money Markets, Inc. and a Letter dated December 4, 1997 with First Chicago Capital Markets, Inc. to amend the Merrill and First Chicago CP Agreements to indicate that the aggregate amount of commercial paper that the Company may have outstanding at any time shall be increased from \$400 to \$500 million.

A statement showing capitalization, pretax interest coverage, and debt interest and preferred stock dividend requirements at December 31, 1997 is attached hereto as Schedule I.

The following exhibits are filed herewith (with the exhibit numbers corresponding to the applicable paragraph number of Chapter 25-8, Rule 25-8.009 of the Florida Administrative Code):

### Exhibit No. Description of Exhibit

- (a)-1 Second Amended and Restated Credit Agreement A dated as of November 18, 1997, between the Company, the Lenders named therein and Chase, as agent for the Lenders.
- (a)-2 Second Amended and Restated Credit Agreement B dated as of November 18, 1997, between the Company, the Lenders named therein and Chase, as agent for the Lenders.
- (a)-3 Commercial Paper Issuer memorandum dated November 18, 1997 of Merrill Lynch Money Markets Inc.
- (a)-4 Commercial Paper Offering Memorandum dated November 18, 1997 of First Chicago Capital Markets, Inc.
- (d)-1 Commercial Paper Dealer Agreement dated December 22, 1988 between the Company and Merrill Lynch Money Markets Inc. (Filed as Exhibit (d)-1 to the Company's Consummation Report dated March 26, 1997, as filed with the Commission in Dockst No. 951229-EI on March 27, 1997, and incorporated herein by reference.)
- (d)-2

  Letter agreement dated November 18, 1997 from the Company to Merrill Lynch

  Money Markets, Inc. regarding increase in maximum amount of Commercial

  Paper outstanding from \$400 to \$500 million.

- (d)-3

  Letter Agreement dated November 20, 1992 between the Company and First Chicago Capital Markets, Inc. relating to the Company's commercial paper. (Filed as Exhibit (d)-2 to the Company's Consummation Report dated March 26, 1997, as filed with the Commission in Docket No. 951229-EI on March 27, 1997, and incorporated herein by reference.)
- (d)-4 Letter dated December 4, 1997 from the Company to First Chicago Trust Company of New York regarding increase in maximum amount of Commercial Paper outstanding from \$400 to \$500 million.

Respectively submitted this 26th day of March, 1998.

FLORIDATION FLORIDATION

Rodney E. Coddy

**Assistant General Counsel** 

P:/Credit.Agt/Consum.98

### SCHEDULE I

Statement showing capitalization, pre-tax interest coverage, and debt interest and preferred stock dividend requirements at December 31, 1997.

### FLORIDA POWER CORPORATION SELECTED FINANCIAL DATA

### SCHEDULE 1

### CAPITALIZATION:

Fleride Power's capitalization at December 31, 1997:

	Interest Rate					
		(in millions)				
First mortgage bands						
Maturing in 1000	0.50%	\$75.0				
Maturing 2002 and 2003	6.50% (a)	200.0				
Moturing 2006	0.00%	80.0				
Maturing 2021 through 2023	7.98% (a)	400.0				
Pollution control refunding revenue bonds						
Moturing 2014 through 2027	6.50% (a)	240.9				
Notes meturing		-				
1007-1006	6.67%	1.5				
1000-2000	6.60% (a)	474.5				
Commercial paper, supported by revelver maturing						
November 30, 2002	5.85% (a)	200.0				
Discount, not of premium, being amerized over term of bands		(5.0)				
Total leng-term dobt		1,748.9				
Notes poyable		179.8				
Total data		1,928.7				
		1,869.7				

### Professed stock:

thout sinking funds, not subject to mandatory

	Obdient Bate	Current Redemption Price	Shares Outstanding	
	4.00% Series	\$104.25	30,000	4.0 7.5
	4.40% Series 4.50% Series	\$102.00 \$101.00	75,000 90,900	7.5 10.0
	4.80% Series	8103.25	30,007	4.0
	4.79% Series	\$102.00	80,000	8.0
Total preferred stock			334,667 (b)	33.5
Common stock equity Total capitalization				1,767.5 \$3,727.7

<sup>(</sup>a) Weighted everage interest rate at December 31, 1907.
(b) Total authorized chares outstanding at December 31, 1907: 336,000.

### FLORIDA POWER CORPORATION SELECTED FINANCIAL DATA

### SCHEDULE 1

### PRE-TAX INTEREST COVERAGE:

Florida Power's pre-tex interest coverage for 1907 was 2.75.

### DEBT INTEREST:

Floride Power's debt interest charges for 1987 were \$117.3 million.

### PREFERRED STOCK DWIDEND REQUIREMENTS:

Floride Power's preferred stock dividend requirements for 1907 were \$1.5 million.

### EXHIBIT (a)-1

Second Amended and Restated Credit Agreement A

	FLORIDA POWER CORPORATION
SECOND A	AMENDED AND RESTATED CREDIT AGREEMENT A
	Deted as of November 18, 1997
Cı	This Agreement amends and restates edit Agreement A Deted as of November 26, 1991
	THE CHASE MANHATTAN BANK

### TABLE OF CONTENTS

This Table of Contents is not part of the Agreement to which it is attached but is inserted for convenience only.

																	P	NE:
Section 1.	Definition	and Acco	untine Me	mers .														. 1
	1.01	Certain De	AND DESCRIPTION OF THE PARTY OF	Contract of the Contract of th														
	1.02	Accounting																
	1.03	Classes an																
	1.03		. ryber o			• •	• •	• •	٠.	٠,	•	٠.	•	• •	•	• •	•	••
Section 2.	Commit	month																74.00
	2.01	Syndicated																
	2.02	Borrowing	s of Syad	icated	Loss													12
	2.03	Money Ma	etes Loss															12
	2.04	Changes o	(Commit	meets .														16
	2.05	Facility Fe																
	2.06	Leading O																
	2.07	Several Ot																
	2.08	Notes	COMPANY OF THE PARTY OF THE PAR				-											
	2.09	Propaymos										•	•	•	•	٠.	•	
		Loss																18
Section 3.	Payment	of Principa	d and late															18
Jenne J.	3.01	Repayment																
	3.02	laterest .																
	3.02					• •		٠.	• •		•	٠.	•		•		•	10
Section 4	Payme	ats; Pro Rate																
	4.01	Paymonia .											. 7					19
	4.02	Pro Rate T	restment															20
	4.03	Computation																
	4.04	Mislaum .	Amounts														. :	20
	4.05	Certain No																
	4.06	Non-Recei	et of Pund	s by th	e As	eat											. :	21
	4.07	Sharing of																
Section 5.	Yield Po	-	llegality .														. !	23
	5.01	Additional																
	5.02		on Types															
	5.03	Illegality .																
	5.04	Treatment																
	5.05	Compenset	No. of the second second second		The Control of the Co								3000		75.00			
Section 6.	Condition	as Precedent									, .						. :	28
	6.01	Initial Loss																
	-						1100	1000		- 7			7.00		-			

	6.02	Initial and	Subsequ	est Lo	-	٠.			 ٠.		 ٠.		 	٠.	28
Section 7.	Represen	stations and	Warrantie	<b>.</b>					 		 		 		. 29
	7.01	Corporate													
	7.02	Financial													
	7.03	Litientice													
	7.04	No Bread													
	7.05	Corporate											 		
	7.06	Approvale													
	7.07	Use of La													
	7.08	ERISA .													
	7.09														-
		Taxes													
	7.10	Investmen													
	7.11	Public Ut	usy Hold	ag Co	-	,	AC		 	٠.		•	 ٠		31
Section 8.	Covena	s of the Co	mpany .						 		 		 		32
	8.01	<b>Financial</b>													
	8.02	Litigation							 		 		 		34
	8.03	Corporate													
	8.04	Probibitio													
	8.05	Use of Pe													
	8.06	ladobteda													
	8.07	Negative													
Section 9.	Events o	( Default .							 						36
Section 10.	The Age	<b></b>									 		 		38
	10.01	Appointm													
	10.02	Reliance t						_	 				 	-	_
	10.03														
	10.04		Louder												
	10.05	The state of the s	etice												
	10.06	Nos-Relie	CONTRACTOR OF THE PARTY OF THE	E F E E									 	17.	
	10.07	Failure to													
	10.00	Parlament.	The second secon	oval of		4. 15. 1									
	10.09	Acres B				-	-								
	10.10	Austice F													
Section 11.	Meete														41
Secure 11.	11.01	Waiver .			A										
	11.02	Control of the Contro			7										
	11.03	the second secon	Esc												
	11.04	A Total	Control of the Contro		3. 7. 0										
	11.05	Swarzen													41.00
	44.00	Name of Street, or other Party of Street, or							 				 		-

11.06	Assignments and Participations
11.07	Survival
11.08	Captions
11.09	Counterparts
11.10	Governing Law; Submission to Jurisdiction
	Waiver of Jury Trial
EXHIBIT A-1	- Form of Note for Syndicated Loans
EXHIBIT A-2	- Form of Note for Money Market Loans
EXHIBIT B	- Form of Opinion of Counsel to the Parent
EXHIBIT C	- Form of Money Market Quote Request
EXHIBIT D	- Form of Money Market Quote
	2 days and

NDED AND RESTATED CREDIT / Š B

princi

# ection 1. Definition and Accounting Matter

5 Ē 2 provisions of this and vice versa):

Type of CO OC M

## · Applicable Margin Total asses

- (i) during each Class I Rating Period.
- (A) with respect to Base Rate Loans, zero,
- (B) with respect to Eurodolfer Losse, 0.19%, and
- (C) with respect to CD Loans, 0.315%,
- during each Class 2 Rating Period.
- (A) with respect to Base Rate Loans, zero,
- B) with respect to Eurodollar Losse, 0.275%, and
- (C) with respect to CD Louis, U.4%, and
- during each Class 3 Rating Period.

- (A) with respect to Base Rate Loans, zero.
- (B) with respect to Eurodollar Losse, 0.45%, and
- (C) with respect to CD Loans, 0.575%.

ach Class 2 Rating Period, 0.10%; and (c)

laterest Period for any CD Loan, the secondary, to the measure 1/100 of 1%) deal Office

Prime Rate for such day.

which bear interest at rates based

:

2

Op Land and faition of "Fixed Base Rate" "Chase" shall mean The Chase Manhattan Book.

"Class" shall have the meaning given to that term in Section 1.03 hereof.

"Class I Rating Pariod" shall mean any period during which the rating of the First Mortgage Bonds (a) by Moody's equals or exceeds "A3" and (b) by S&P equals or exceeds "A-".

"Class 2 Rating Parind" shall mean any period during which the rating of the First Mortgage Bonds (a) by Moody's equals or exceeds "Bas3" and (b) by S&P equals or exceeds "BBB-", and which is not a Class 1 Rating Period.

"Class 3 Rating Pariod" shall mean any period that is neither a Class 1 Rating Period nor a Class 2 Rating Period.

"Code" shall mean the laternal Revenue Code of 1986, as amended from time to time.

"Commitment" shall mean, with respect to each Leader, the obligation of such Leader to make Syndicated Leans pursuant to Section 2.01 hereof in an aggregate amount at any one time outstanding up to but not exceeding the amount set opposite such Leader's name on the signature pages hereof under the caption "Commitment" (as the same may be reduced at any time or from time to time pursuant to Section 2.04 hereof). The original aggregate amount of the Commitments is \$300,000,000.

"Commitment Termination Date" shall mean, with respect to each Lender, November 17, 1996; provided that

- (a) if, (i) not later than 90 days prior to the Commitment Termination Date, determined after giving office to all provious extensions thereof pursuant to this definition (the "Existing Commitment Termination Date"), the Company requests that the Lenders agree to extend the Commitment Termination Date to the 364th day after the Existing Commitment Termination Date (the "Proposal Commitment Termination Date") and (ii) each of the Lenders so agrees in writing on or prior to the Existing Commitment Termination Date (such agreement based on each Lender's credit determination made at such time), then the "Commitment Termination Date" shall be extended, with respect to each Lender, to the Proposed Commitment Termination Date;
- (b) if, pursuant to any such request, some, but not all, of the Lenders agree to so extend the Existing Commitment Termination Date to the Proposed Commitment Termination Date (the Lenders that so agree, the "Commitment Termination Date" shall mean (i) with respect to the Commitment Termination Date and (ii) with respect to the Lenders that are not Commitment Termination Date and (iii) with respect to the Lenders that are not Commitment Termination Date; and

O c) If the Commitment Termination Date as determined above is not a Day, the Commitment Termination Date shall be the next preceding Business

"Committeed Schniding" shall Person (whether now existing or beneather created by (or should have been) committeed accordance with GAAP. Created or sequinal ean, as to any Person, each Subsidiary of such or acquired) the financial statements of which in the financial statements of such Person in

Continue". "Continue of a Plant pursuant to Section 2.09 berroef of a Plant same Type from one lawrest Period to the " and "Continued" shall refer to the continuation of Rase Long of one Type as a Fixed Rase Long of the ment language Pariod.

"Cannet". "Canneting" and "Canneting" shall refer to a conversion pursuant to Section 2.09 hereof of Base Rase Louis into CD Louis or Eurodollar Louis, of CD Louis into Base Louis or Eurodollar Louis, of CD Louis into Base Rase Louis or CD Louis, which may be accompanied by the transfer by a Louiser (at its sole discretion) of a Louis from one Applicable Louising Office to another.

"Default" shall mean an Event of Def of time or both would become an Event of Default. bask or an event which with notice or lapse

P 2" and "3" shall mean lewful money of the United States of America.

"ERISA" shall make the Employee Rationment Income Security Act of 1974, as monded from time to time.

"Elisa Affilias" shall mean any corporation or trade or business which is a member of the same constrolled group of corporations (within the meaning of Section 414(b) of the Code) as the Company or is under common control (within the meaning of Section 414(c) of the Code) with the Company.

"Empirite Loan" shall mean Syndicated Loans the interest rates on which are determined on the basis of rates referred to in clause (a) of the definition of "Fixed Base Rate" in this Section 1.01.

"Event of Datash" shall have the meaning assigned to such term in Section 9

"Existing Industry: shall mean the Industrie dated as of January 1, 1944 between the Company and First Chicago Trust Company of New York, successor Trustee, as amended and supplemented and in effect from time to time.

shall mean the Company's First Mortgage Bonds issued

with respect to any Fixed Rase Loan for any

- 9

sech Loss;

determine such Fined Base Rate on the basis of information timely furnished by the remaining Reference London.

"Final Res" shall mean, for any Fixed Rate Loan for any Interest Period therefor, a rate per annum (rounded upwards, if necessary, to the nearest 1/100 of 1%) determined by the Agent to be equal to the sum of (a) the Fixed base Rate for such interest Period divided by 1 minus the Receive Requirement for such Loan for such interest Period plus (b) if such Loan to a CD Loan, the Assessment Rate for such interest Period.

"Final Rate Loans" shall mean CD Loans, Eurodoller Loans and, for the purposes of the definition of "Fixed Base Rate" in this Section 1.01 and in Section 5 hereof, LIBOR Market Loans.

"GAAP" shall mean generally accepted accounting principles as in effect from time to time.

"Guarante" shall mean any guarantee, endorsoment, contingent agreement to purchase or to furnish funds for the payment or maintenance of, or otherwise to be or become contingently liable under or with respect to, Indebtedness, other obligations, not worth, working capital or earnings of any Person, or a guarantee of the payment of dividends or other distributions upon the stock of any Person, or any agreement to purchase, sell or lease (as lesses or lesser) property, products, materials, supplies or services primarily for the purpose of enabling any Person to make payment of its obligations or any agreement to assure a creditor against less, and including without limitation, causing a bank to open a letter of credit for the benefit of another Person, but excluding endorsements for collection or deposit in the ordinary course of business. The terms "Guarantee" and "Guaranteed" used as a verb shall have a correlative meaning.

"Indebtedness" shall mean, as to any Person: (a) indebtedness created, issued or incurred by such Person for berrowed money (whether by loan or the issuance and sale of debt securities); (b) obligations of such Person to pay the deferred purchase or acquisition price of property, other than trade accounts psyable (other than for borrowed money) arising, and accrued expanses incurred, in the ordinary course of business so long as such trade accounts psyable are paid within 90 days of the date the respective goods are delivered; (c) obligations of such Person in respect of latters of credit or similar instruments issued or accepted by banks and other financial institutions for the account of such Person; (d) Capital Lesse Obligations of such Person; and (e) Indebtedness of others Guaranteed by such Person.

### "Interest Period" shall move:

(a) with respect to any Eurodollar Loan, each period commencing on the date such Eurodollar Loan is made or Converted from a Loan of another Type or the last day of the next preceding interest Period for such Loan and ending on the numerically corresponding day in the first, second, third or sixth calendar month thereafter, as the

Company may select as provided in Section 4.05 hereof, except that each Interest Period which commences on the last Business Day of a calendar month (or on any day for which there is no manufacility corresponding day in the appropriate subsequent calendar month) shall end on the last Business Day of the appropriate subsequent calendar month;

- (b) with respect to any CD Loan, each period commencing on the date such CD Loan is made or Converted from a Loan of another Type or the last day of the next preceding Interest Period for such Loan and ending on the day 30, 60, 90 or 180 days thereafter, as the Company may select as provided in Section 4.05 hereof;
- (c) With respect to any Set Rate Loan, the period commencing on the date such Set Rate Loan is made and ending on any Business Day up to 180 days thereafter, as the Company may select as provided in Section 2.03(b) hereof; and
- (d) With respect to any LIBOR Market Loan, the period commencing on the date such LIBOR Market Loan is made and ending on the sumerically corresponding day in the first, second, third or sinth calendar month thereafter, as the Company may select as provided in Section 2.03(b) hereof, except that each listerest Period which commences on the last Business Day of a calendar month (or any day for which there is no numerically corresponding day in the appropriate subsequent calendar month; shall end on the last Business Day of the appropriate subsequent calendar month.

Notwithstanding the foregoing: (i) so Interest Period with respect to Loans to be made by any Leader may and after such Leader's Commitment Termination Date (as in effect on the first day of such Interest Period); (ii) each Interest Period which would otherwise end on a day which is not a Business Day shall and on the next succeeding Business Day (or, in the case of an Interest Period for Eurodolfar Loans or LIBOR Market Loans, if such next succeeding Business Day falls in the next succeeding calendar month, on the next preceding Business Day); and (iii) notwithstanding clause (i) above, no Interest Period for any Fixed Rate Loans or LIBOR Market Loans shall have a duration of less than one month (in the case of Eurodolfar Loans and LIBOR Market Loans) or 30 days (in the case of CD Loans) and, if the Interest Period for any Fixed Rate Loans or LIBOR Market Loans would otherwise be a shorter period, such Loans shall not be available herounder.

"LIBO Rate" shall mean, for any LIBOR Market Loan, a rate per sessum (rounded upwards, if necessary, to the nearest 1/100 of 1%) determined by the Agent to be equal to the rate of interest specified in clause (a) of the definition of "Fixed Base Rate" in this Section 1.01 for the Interest Period for such Loan divided by 1 minus the Reserve Requirement for such Loan for such Interest Period.

"LIBOR Auction" shall mean a solicitation of Money Market Quotes setting forth Money Market Margins based on the LIBO Rate pursuant to Section 2.03 hereof.

"LIBOR Merket Loans" shall mean Money Market Loans the interest rates on which are determined on the basis of LIBO Rates pursuent to a LIBOR Auction.

"Lies" shall mean, with respect to any asset, any mortgage, lies, plodge, hypothecation, charge, security inserest or encumbrance of any kind in respect of such asset. For purposes of this Agreement, the Purent or any of its Subsidiaries shall be deemed to own subject to a Lies any asset which it has acquired or holds subject to the interest of a vendor or lessor under any conditional sale agreement, capital lease or other title retention agreement relating to such asset.

"Losse" shall mean Money Market Losse and Syndicated Losse.

"Majority Landors" shall mean Londors having at least 66-2/3% of the aggregate amount of the Commitments; provided that, if all of the Commitments shall have terminated, Majority Landors shall mean Londors holding at least 66-2/3% of the aggregate uspaid principal amount of the Longo.

"Margin Stack" shall mean margin stock within the meaning of Regulations U and X.

"Money Market Barrowing" shall have the meaning assigned to such term in Section 2.03(b) hereof.

"Money Market Loans" shall mean the loans provided for by Section 2.03 hereof.

"Money Market Margin" shall have the meaning assigned to such term in Section 2.03(c)(ii)(C) hereof.

"Money Market Chase" shall mean an offer in accordance with Section 2.03(c) hereof by a Londor to make a Money Market Lone with one single specified interest rate.

"Money Market Chain Request" shall have the meaning assigned to such term in Section 2.03(b) hereof.

"Money Market Rese" shall have the meaning assigned to such term in Section 2.03(c)(ii)(D) hereof.

"Montr's" shall mean Mondy's Investors Services, Inc.

"Multiamplayer Plan" shall mean a multiamployer plan defined as such in Section 3(37) of ERISA to which contributions have been made by the Company, the Parent or any ERISA Affiliate and which is covered by Title IV of ERISA.

"1935 Act" shall have the messing given to that term in Section 7.11 hereof.

W W. S. S. man to Section 13 or 15(d) of the Securities Exchange Act of 1934.

notes provided for by Section 2.08 hereof.

Other FEC A C Ì ided and Restated Credit 20

. Bear Ploride Progr ses Corporation, a Florida corporation.

The Party of the P Action of Pals. fit Guaranty Corporation or any entity

ō a of Nove Ė Progress Capital and (b) the Second

ity or political organization or mpany, voluntary association, government (or any agency,

y or my ERISA AMI II Plan') usions do a me of \$30,000,000. ther plan established or maintained by ich is covered by Title IV of ERISA, then established or maintained by the rial liabilities of less than \$50,000,000 actuarial liabilities of all Small Plans

acipal of any Loan or any Note that is not paid when rate per assum during the Loss and the che 'Post-Deft

China Bas' dall man de Ē

Principal Office Ave , New York, New York 10017. . E 0

õ terly Date shall 2 7 M

York (or their Leading Offices, as the case may be). ranty Trust Company of

G D. -Ė of Govern

Č 9

0 7 \$100,000 a

SAP' shall meen Sundard & Poor's Corporation

"Set Rate Auction" shall mean a solicitation of Money Market Quotes setting forth Money Market Rates pursuant to Section 2.03 hereof.

"Set Rate Loans" shall mean Money Market Loans the interest rates on which are determined on the basis of Money Market Rates pursuent to a Set Rate Auction.

"Subsidiery" shall mean, as to any Person, any corporation of which at least a majority of the outstanding shares of stock having by the forms thereof ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether or not at the time stock of any other class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is at the time directly or indirectly owned or controlled by such Person or one or more Subsidieries of such Person or by such Person and one or more Subsidieries of such Person. "Whelly-surned Subsidiery" shall mean any such corporation of which all of such shares, other than directors' qualifying shares, are so owned or controlled.

"Syndiented Leans" shall mean the leans provided for by Section 2.01 hereof.

"Syndicated Massa" shall mean the promissory notes provided for by Section 2.00(a) hereof.

"Total Contribution" shall mean, with respect to any Person, the sum of the value of the common stock, retained earnings, and preferred and preference stock of such Person (in each case, determined in accordance with GAAP) plus all Indebtedness of such Person.

"Type" shall have the meaning given to that term in Section 1.03 hereof.

### 1.02 Accounting Terms and Deserminations.

- (a) Except as otherwise expressly provided herein, all accounting terms used herein shall be interpreted, and all financial statements and certificates and reports as to financial matters required to be delivered to the Londors hereunder shall (unless otherwise disclosed to the Londors in writing) be prepared, in accordance with generally accepted accounting principles applied on a basis consistent with that used in the preparation of the latest financial statements furnished to the Londors hereunder after the date hereof.
- (b) The Company will not change the last day of its fiscal year from December 31 of each year, or the last days of the first three fiscal quarters in each of its fiscal years from March 31, June 30 and September 30 of each year, respectively.
- 1.03 Cleans and Types of Loans. Loans herounder are distinguished by "Class" and by "Type". The "Class" of a Loan (or of a Commitment to make a Loan) refers to whether such Loan is a Money Martest Loan or a Syndicated Loan, each of which constitutes a Class. The "Type" of a Loan refers to whether such Loan is a Base Rate Loan, a CD Loan,

a Eurodoller Loan, a Set Rate Loan, or a LIBOR Market Loan, each of which constitutes a Type. Loans may be identified by both Class and Type.

### Section 2. Commitments.

- 2.01 Syndicated Loans. Each Loader severally agrees, on the terms of this Agreement, to make loans to the Company in Dollars during the period from and including the date hereof to but not including such Londor's Commitment Termination Date in an aggregate principal amount at any one time outstanding up to but not exceeding the amount of such Londor's Commitment as then in office. Subject to the terms of this Agreement, during such period the Company may berrow, rupsy and reborrow the amount of the Commitments by means of flace Rate Loans, CD Loans and Buredoller Loans and may Convert Syndicated Loans of one Type into Syndicated Loans of another Type (as provided in Section 2.09 hereof) or Continue Syndicated Loans of one Type as Syndicated Loans of the same Type; provided that there may be no more than 15 different interest Periods for Syndicated Loans outstanding at the same time.
- 2.02 <u>Becoming of Syndicated Loans</u>. The Company shall give the Agent (which shall promptly antify the Landers) notice of each berrowing hereunder of Syndicated Loans as provided in Section 4.05 hereof. Not later than most New York time on the date specified for each berrowing of Syndicated Loans berrounder, each Loader shall make available the amount of the Syndicated Loan to be made by it on such date to the Agent, at account member NYAO-DI-900-9-000002 maintained by the Agent with Chase at the Principal Office, in immediately available funds. The amount so received by the Agent shall, subject to the terms and conditions of this Agreement, be made available to the Company by depositing the same, in immediately available funds, in an account of, and designated by, the Company maintained at a bank in New York City.

### 2.03 Money Mediat Long.

- (a) In addition to borrowings of Syndicated Loans, the Company may, as set forth in this Section 2.03, request the Londors to make offers to make Money Market Loans to the Company in Dollars. The Londors may, but shall have no obligation to, make such offers and the Company may, but shall have no obligation to, accept any such offers in the manner set forth in this Section 2.03. Money Market Loans may be LIBOR Market Loans or Set Rate Loans, provided that:
  - (i) there may be no more than 15 different interest Periods for both Syndicated Leans and Money Market Loans outstanding at the same time (for which purpose interest Periods described in different lettered clauses of the definition of the term "Interest Periods shall be deemed to be different interest Periods even if they are constrained;

- (ii) the aggregate principal amount of all Money Market Loans, together with the aggregate principal amount of all Syndicated Loans, at any one time outstanding shall not exceed the aggregate amount of the Commissents at such time.
- (b) When the Company wishes to request offers to make Money Market Lones, it shall give the Agent (which shall promptly notify the Londors) notice (a "Money Market Conte Regnet") so as to be received so later than 11:00 a.m. New York time on (x) the fourth Business Day prior to the date of berrowing proposed therein, in the case of a LIBOR Auction or (y) the Business Day next proceding the date of berrowing proposed therein, in the case of a Set Rate Auction (or, in any such case, such other time and date as the Company and the Agent, with the consent of the Majority Londors (and with notice to each Londor prior to the effectiveness of such consent), may agent). The Company may request offers to make Money Market Lones for up to five different interest Periods in a single notice (for which purpose Interest Periods in different lettered classes of the definition of the term "Interest Period" shall be deemed to be different interest Periods even if they are coterminous); provided that the request for each separate interest Period shall be deemed to be a separate Money Market Quote Request for a separate berrowing (a "Manny Market Barrowing"). Back such notice shall be substantially in the form of Exhibit C hereto and shall specify as to each Money Market Borrowing:
  - (i) the proposed date of such borrowing, which shall be a Business
    Day;
  - (ii) the aggregate amount of such Money Market Borrowing, which shall be at least \$10,000,000 (or in larger multiples of \$1,000,000) but shall not cause the limits specified in Section 2.03(a) bereof to be violated;
    - (iii) the duration of the Interest Period applicable thereto;
  - (iv) whether the Money Martest Quotes requested for a particular Interest Period are seeking quotes for LIBOR Martest Loans or Set Rate Loans; and
  - (v) if the Meany Market Quetes requested are seeking quotes for Set Rate Leans, the date on which the Meany Market Quotes are to be submitted if it is before the proposed date of berrowing (the date on which such Meany Market Quotes are to be submitted is called the "Quetation Date").

Except as otherwise provided in this Section 2.03(b), no Money Market Quote Request shall be given within five Business Days (or such other number of days as the Company and the Agent, with the consent of the Majority Londons (and with notice to each London prior to the effectiveness of such consent), may agree) of any other Money Market Quote Request.

(c) (i) Each Londor may submit one or more Money Market Quotes, each containing an offer to make a Money Market Loan in response to any Money

Market Quote Request; provided that, if the Company's request under Section 2.03(b) hereof specified more than one Interest Period, such Lander may make a single submission containing one or more Money Market Quotes for each such Interest Period. Each Money Market Quote must be submitted to the Agent not later than (x) 2:00 p.m. New York time on the fourth Business Doy prior to the proposed date of borrowing, in the case of a Set Rate Auction or (y) 10:00 a.m. New York time on the Quotation Date, in the case of a Set Rate Auction (or, in any such case, such other time and date as the Company and the Agent, with the consent of the Majority Landers (and with notice to each Lander prior to the offictiveness of such counsel), may agree); provided that any Menny Market Quote submitted by Chase (or its Applicable Landing office) may be submitted, and may only be submitted, if Chase (or such Applicable Landing Office) notifies the Company of the terms of the offic commised thereis not later than (x) 1:00 p.m. New York time on the fourth Business Day prior to the proposed date of borrowing, in the case of a LIBOR Auction or (y) 9:45 a.m. New York time on the Quotation Date, in the case of a Set Rate Auction. Subject to Sections 5.02(b), 5.03, 6.02 and 9 hereof, any Menny Market Quote so made shall be irrovocable except with the written connect of the Agent given on the instructions of the Company.

- (ii) Each Meany Market Quote shall be substantially in the form of Exhibit D horses and shall specify:
  - (A) the proposed date of borrowing and the Interest Period therefor;
  - (B) the principal amount of the Money Market Loan for which each such offer is being made, which principal amount shall be at least \$5,000,000 or a larger multiple of \$1,000,000; provided that the aggregate principal amount of all Money Market Loans for which a Londor submits Money Market Quotes (x) may be greater or less than the aggregate Commitments of such Londor but (y) may not exceed the principal amount of the Money Market Borrowing for a particular Interest Period for which offers were requested;
  - (C) in the case of a LIBOR Auction, the margin above or below the applicable LIBO Rate (the "Meany Market Margin") offered for each such Money Market Loan, supressed as a percentage (rounded upwards, if necessary, to the nearest 1/10,000th of 1%) to be added to or subtracted from the applicable LIBO Rate;
  - (D) in the case of a Set Rate Auction, the rate of interest per annum (rounded upwards, if accessary, to the nearest 1/10,000th of 1%) offered for each such Memory Market Lean (the "Menory Merket Rate");
    - (E) the identity of the quoting Londor; and

of all Money

- \$5,000,000 or
- principal amount of each Money Marter

- (ii) the aggregate principal amount of each Money Market Borrowing shall be at least \$10,000,000 (or in larger multiples of \$1,000,000) but shall not cause the limits specified in Section 2.03(a) hereof to be violated; and
- (iii) the Company may not accept any offer where the Agent has advised the Company that such offer fails to comply with Section 2.03(c)(ii) hereof or otherwise fails to comply with the requirements of this Agreement (including, without limitation, Section 2.03(a) hereof).
- (f) Any Londor whose offer to make any Money Market Loan has been accepted shall, not later than noon New York time on the date specified for the making of such Loan, make the amount of such Loan available to the Agent at account number NYAO-DI-900-9-000002 maintained by the Agent with Chase at the Principal Office in immediately available funds, for account of the Company. The amount so received by the Agent shall, subject to the terms and conditions of this Agreement, be made available to the Company on such date by depositing the same, in immediately available funds, in an account of the Company maintained with a bank in New York City designated by the Company.
- (g) Except for the purpose and to the extent expressly stated in Section 2.04(b) hereof, the amount of any Menny Market Loan made by any Londor shall not constitute a utilization of such Londor's Commitment.

### 2.04 Channel of Complement.

- (a) The amount of each Londor's Commitment shall be automatically reduced to zero on such Londor's Commitment Termination Date.
- (b) The Company shall have the right at any time or from time to time (i) so long as no Syndicated Loans or Money Market Loans are outstanding, to terminate the Commitments, and (ii) to reduce the aggregate unused amount of the Commitments (for which purpose use of the Commitments shall be deemed to include the aggregate principal amount of all Money Market Loans); provided that (x) the Company shall give notice of each such termination or reduction as provided in Section 4.05 hereof, and (y) each partial reduction shall be in aggregate amount at least equal to \$10,000,000 and in multiples of \$1,000,000 in excess thereof.
  - (c) The Commitments once terminated or reduced may not be reinstated.
- 2.05 Pacility Res. The Company shall pay to the Agent for account of each Leader a facility fee on the amount of such Leader's Commitment, for the period from and including the date of this Agreement to but not including the earlier of the date such Commitment is terminated or such Leader's Commitment Termination Date, at a rate per annual equal to the Applicable Facility Fee Rate. Accrued facility fees payable to any Leader shall be

- of each Type.

### 2.08 Year

# on 3. Promote of Principal and Interest.

## 3.01 - Danman of Louis

- 0 1 De De
- 40 RESPRIENCE OF AGE 1 Mony Marter Long
- 5
- during much periods as such Loss is a Base Rase Loss, the Base Rase (as see to thes) pine the Applicable Margin (if any);
- Ę tests as such Loss to a Pixed Rate Loss. for each late ad Rate for such Loss for such laterest Period plus
- Loss, the LIBO Ruse for a Messay Marter Margin q otion 2.00 hereof; and quoted by the
- Section 2.03 barro ing such Loan in acc 2000 Loss for the cordsace with

of such Loan to a Loan of another Type (but only on the principal amount so paid, propaid or Converted), except that interest psychie at the Post-Default Rate shall be psychie from time to time on domaid. Promptly after the determination of any interest rate provided for herein or any change therein, the Agent shall give notice thereof to the London to which such interest is psychic and to the Company.

Section 4. Promests: Pro Rata Treatment: Commutations: Etc.

### 4.01 Payments

- (a) Except to the extent otherwise provided herein, all payments of principal, interest and other amounts to be made by the Company under this Agreement and the Notes shall be made in Dollars, in immediately available funds, without deduction, set-off or counterclaim, to the Agent at account number NYAO-DI-900-9-000002 maintained by the Agent with Chase at the Principal Office, not later than 2:00 p.m. New York time on the date on which such payment shall become due (each such payment made after such time on such due date to be deemed to have been made on the next succeeding Business Day).
- (b) Any Londor for whose account any such payment is to be made, may (but shall not be obligated to) debit the amount of any such payment which is not made by such time to any ordinary deposit account of the Company (for purposes of this Section 4.01(b), "ordinary deposit account of the Company" shall not include any account in the name of a Person other than the Company) with such Londor (with notice to the Company).
- (c) The Company shall, at the time of making each payment under this Agreement or any Note, specify to the Agent (which shall promptly notify the intended recipionts) thereof) the Leans or other amounts payable by the Company increased to which such payment is to be applied (and in the event that it falls to so specify, or if an Event of Default has occurred and is continuing, such Leader may apply the amount of such payment received by it from the Agent in such manner as such Leader may determine to be appropriate).
- (d) Each payment received by the Agent under this Agreement or any Note for account of a Londor shall be paid promptly to such Londor, in immediately available funds, for account of such Londor's Applicable Londing Office for the Long in respect of which such payment is made.
- (e) If the dae date of any payment under this Agreement or any Note would otherwise full on a day which is not a Business Day such date shall be extended to the next succeeding Business Day and interest shall be payable for any principal so extended for the period of such extension.
- 4.02 Pro Rate Treatment. Except to the extent otherwise provided herein:

  (a) each borrowing from the Lenders under Section 2.01 hereof shell be made from the Lenders, each payment of facility for under Section 2.05 hereof shall be made for account of the Lenders,

H

K

õ 

such day and, if such recipient(s) shall full promptly to make such payment, the Agent shall be entitled to recover such amount, on demand, from the Payor, together with interest as aforesaid.

### 4.07 Shering of Payments, Etc.

- (a) The Company agrees that, in addition to (and without limitation of) any right of set-off, busher's lien or counterclaim a Londer may otherwise have, each Londer shall be estimated, at its option, to offset balances held by it for account of the Company (for purposes of this Section 4.07(a), "balances held for account of the Company" shall not include any balances held in an account in the same of a Person other than the Company) at any of its offices, in Dollars or in any other currency, against any principal of or interest on any of such Londer's Londo, or any other amount psyable to such Londor berounder, which is not paid when due (regardless of whether such balances are then due to the Company), in which case it shall promptly notify the Company and the Agent thereof, provided that such Londor's failure to give such notice shall not affect the validity of such offset.
- (b) If any Londor shall obtain payment of any principal of or interest on any Lone made by it to the Company under this Agreement through the enercise of any right of set-off, banker's lies or counterclain or similar right or otherwise, and, as a result of such payment, such Londor shall have received a greater percentage of the principal or interest then due becomeder by the Company to such Londor than the percentage received by any other Londors, it shall promptly purchase from such other Londors participations in (or, if and to the extent specified by such Londor, direct interests in) the Londor owing to such other Londors (or in interest due thereon, as the case may be) in such amounts, and make such other adjustments from time to time as shall be equitable, to the end that all the Londors shall share the benefit of such excess payment (not of any exposess which may be incurred by such Londor in obtaining or preserving such excess payment) pro rate in accordance with the uspaid principal of and/or interest on the Londor shall make appropriate adjustments among themselves (by the result of participations sold or otherwise) if such payment is received or must otherwise be restored.
- (c) The Company agrees that any Londor so purchasing a participation (or direct interest) in the Loans made by other Londors (or in interest due thereon, as the case may be) may exercise all rights of ast-off, banker's lien, counterclaim or similar rights with respect to such participation as fully as if such Londor were a direct holder of Loans in the amount of such participation.
- (d) Nothing contained herein shall require any Leader to exercise any such right or shall affect the right of any Leader to exercise, and retain the benefits of exercising, any such right with respect to any other indebtedness or obligation of the Company.
- (e) If, under any applicable bankruptcy, insolvency or other similar law, any Lender receives a secured claim in lieu of a set-off to which this Section 4.07 applies, such Lender shall, to the extent practicable, exercise its rights in respect of such secured claim in a

manner consistent with the rights of the Londors entitled under this Section 4.07 to share in the benefits of any recovery on such secured claim.

### Section 5. Yield Protection and Illegality.

### 5.01 Additional Costs.

- (a) The Company shall pay directly to each Londer from time to time such amounts as such Londer may determine to be necessary to compensate it for any costs which such Londer determines are attributable to its making or maintaining of any Fixed Rate Longe or its obligation to make any Fixed Rate Longe hereunder, or any reduction in any amount receivable by such Londer hereunder in respect of any of such Longe or such obligation (such increases in costs and reductions in amounts receivable being herein called "Additional Costs"), resulting from any Regulatory Change which:
  - (i) subjects any Londor to taxation on, or changes the basis of taxation of, any amounts psychic to such Londor under this Agreement or its Notes in respect of any of such Longo (other than taxes imposed on or measured by the overall not income of such Londor or of its Applicable Londing Office for any of such Longo by the jurisdiction in which such Londor has its principal office or such Applicable Londing office); or
  - (ii) imposes or modifies any reserve, special deposit or similar requirements (other than the Reserve Requirement utilized in the determination of the Fined Rate or LIBO Rate, as the case may be, for such Loan) relating to any extensions of credit or other assets of, or any deposits with or other liabilities of, such Londor (including any of such Loans or any deposits referred to in the definition of "Fixed Base Rate" in Section 1.01 hereof), or any commitment of such Londor (including the Commitments of such Londor hereunder); or
  - (iii) imposes any other condition affecting this Agreement or its Notes (or any of such extensions of credit or liabilities) or its Commissions.

If any Londor requests compensation from the Company under this Section 5.01(a), the Company may, by notice to such Londor (with a copy to the Agent), suspend the obligation of such Londor to make or Continue Longo of the Type with respect to which such compensation is requested until the Regulatory Change giving rise to such request causes to be in effect (in which case the provisions of Section 5.04 hereof shall be applicable).

(b) Without limiting the effect of the provisions of paragraph (a) of this Section 5.01, in the event that, by reason of any Regulatory Change, any Lender either (i) incurs Additional Costs based on or measured by the excess above a specified level of the amount of a category of deposits or other liabilities of such Londor which includes deposits by reference to which the interest rate on Buredeller Longs or CD Longs is determined as provided in this

Agreement or a category of extensions of credit or other assets of such Lender which includes Eurodollar Loans or CD Loans or (ii) becomes subject to restrictions on the amount of such a category of liabilities or assets which it may hold, then, if such Lender so elects by notice to the Company (with a copy to the Agent), the obligation of such Lender to make or Continue Loans of such Type herounder shall be suspended until such Regulatory Change ceases to be in effect (in which case the provisions of Section 5.04 hereof shall be applicable).

(c) Without limiting the effect of the foregoing provisions of this Section 5.01 (but without duplication), the Company shall pay directly to each Lender from time to time on request such amounts as such Lender may determine to be necessary to compensate such Lender (or, without duplication, the bank holding company of which such Lender is a subsidiary) for any costs which it determines are attributable to the meintenance by such Lender (or any Applicable Lending Office or such bank holding company), pursuant to any law or regulation or any interpretation, directive or request (whether or not having the force of law) of any court or governmental or monetary authority

### (i) following any Regulatory Change, or

(ii) implementing any risk-based capital guideline or requirement (whether or not having the force of law and whether or not the failure to comply therewith would be unlowful) horstofore or hereafter issued by any government or governmental or supervisory authority implementing at the national level the Basic Accord (including, without limitation, the Final Risk-Based Capital Guidelines of the Board of Governors of the Federal Reserve System (12 CFR Part 208, Appendix A; 12 CFR Part 225, Appendix A) and the Final Risk-Based Capital Guidelines of the Office of the Comptroller of the Currency (12 CFR Part 3, Appendix A)),

of capital in respect of its Commitment or Loans (such compensation to include, without limitation, an amount equal to any reduction of the rate of return on assets or equity of such Londer (or any Applicable Londing Office or such bank holding company) to a level below that which such Londer (or any Applicable Londing office or such bank holding company) could have achieved with respect to such Londer's Commitment or Loans hereunder but for such law, regulation, interpretation, directive or request). For purposes of this Section 5.01(c), "Basic Accord" shall mean the proposals for risk-based capital framework described by the Basic Committee on Banking Regulations and Supervisory Practices in its paper entitled "International Convergence of Capital Measurement and Capital Standards" dated July 1988, as amended, modified and supplemented and in effect from time to time or any replacement thereof.

(d) Each Londor will notify the Company of any event occurring after the date of this Agreement that will entitle such Londor to compensation under paragraph (a) or (c) of this Section 5.01 as promptly as practicable, but in any event within 45 days, after such Londor obtains actual knowledge thereof; provided, however, that if any Londor fails to give such notice within 45 days after it obtains actual knowledge of such an event, such Londor shall, with respect to compensation payable pursuant to this Section 5.01 in respect of any costs

resulting from such event, only be entitled to payment under this Section 5.01 for costs incurred from and after the date 45 days prior to the date that such Londor does give such notice; and provided, further, that each Londor will designate a different Applicable Londing Office for the Loans of such Lender affected by such event if such designation will avoid the need for, or reduce the amount of, such compensation and will not, in the sole opinion of such Leader, be disadvantageous to such Londer, except that such Londer shall have no obligation to designate an Applicable Lending office located in the United States of America. Each Lender will furnish to the Company a certificate setting forth the basis and amount of each request by such Lender for compensation under paragraph (a) or (c) of this Section 5.01. Determinations and allocations by any Londor for purposes of this Section 5.01 of the offset of any Regulatory Change pursuant to paragraph (a) or (b) of this Section 5.01, or of the effect of capital maintained pursuant to paragraph (c) of this Section 5.01, on its costs or rate of return of maintaining Loans or its obligation to make Loans, or on amounts receivable by it in respect of Loans, and of the amounts required to compensate such Lender under this Section 5.01, shall be conclusive. provided that such determinations and allocations are made on a reasonable basis and in a manner consistent with the determinations and allocations made by such Lender with respect to its other commitments and extensions of credit similarly affected.

- 5.02 <u>Limitation on Types of Loans</u>. Anything herein to the contrary notwithstanding, if, on or prior to the determination of any Fixed Base Rate for any Interest Period:
  - (a) the Agest determines, which determination shall be conclusive provided that it is made on a reasonable basis, that quotations of interest rates for the relevant deposits referred to in the definition of "Fixed Base Rate" in Section 1.01 hereof are not being provided in the relevant amounts or for the relevant maturities for purposes of determining rates of interest for any Type of Fixed Rate Loans as provided herein; or
  - (b) Londors having more than 50% of the aggregate amount of the Commitments determine (or any Londor that has outstanding a Money Market Quote with respect to a LIBOR Market Loan determines), which determination shall be conclusive provided that it is made on a reasonable basis, and notify (or notifies, as the case may be) the Agent that the relevant rates of interest referred to in the definition of "Fixed Base Rate" in Section 1.01 hereof upon the basis of which the rate of interest for Eurodollar Loans or CD Loans (or LIBOR Market Loans, as the case may be) for such Interest Puriod is to be determined are not likely adequately to cover the cost to such Londors (or to such quoting Londor) of making or maintaining such Type of Loans for such Interest Puriod;

then the Agent shall give the Company and each Lender prompt notice thereof, and so long as such condition remains in effect, the Lenders (or such quoting Lender) shall be under no obligation to make additional Leans of such Type, to Continue Leans of such Type or to Convert Leans of any other Type into Leans of such Type and the Company shall, on the last day(s) of the then current interest Period(s) for the outstanding Leans of such Type, either

prepay such Loans or Couvert such Loans into another Type of Loan in accordance with Section 2.09 hereof.

- 5.03 <u>Hearlity</u>. Notwithstanding any other provision of this Agreement, in the event that it becomes uninwful for any Lender or its Applicable Lending Office to honor its obligation to make or maintain Burodollar Loans or LIBOR Market Loans hervander, then such Lender shall promptly notify the Company thereof (with a copy to the Agent) and such Lender's obligation to make Eurodollar Loans shall be suspended until such time as such Lender may again make and maintain Burodollar Loans (in which case the provisions of Section 5.04 hereof shall be applicable), and such Loader shall no longer be obligated to make any LIBOR Market Loan that it has offered to make.
- 5.04 Trustment of Affected Loans. If the obligation of any Lender to make a particular Type of Pixed Rate Loans shall be suspended pursuant to Section 5.01 or 5.03 hereof (Loans of such Type being herein called the "Affected Loans shall be autocastically Converted into Base Rate Loans on the last day(s) of the then current Interest Period(s) for Affected Loans (or, in the case of a Conversion required by Section 5.01(b) or 5.03 hereof, on such earlier date as such Loader may specify to the Company with a copy to the Agent) and, unless and until such Loader gives notice as provided below that the circumstances specified in Section 5.01 or 5.03 hereof which gave rise to such Conversion no longer exist:
  - (a) to the extent that such Londor's Affected Loans have been so Converted,
     all payments and propayments of principal which would otherwise be applied to such Londor's Affected Loans shall be applied instead to its Base Rate Loans;
  - (b) all Loans which would otherwise be made or Continued by such Londer as Loans of the Affected Type shall be made or Continued instead as Base Rate Loans and all Loans of such Loader which would otherwise be Converted into Loans of the Affected Type shall be Converted instead (or shall remain as) Base Rate Loans; and
  - (c) if Loans of other Leaders of the Affected Type are subsequently Converted into Loans of another Type (other than Base Rate Loans), such Leader's Base Rate Loans shall be automatically Converted on the Conversion date for such Loans of the other Leaders into Loans of such other Type to the extent necessary so that, after giving effect thereto, all Loans held by such Leader and the Leaders whose Loans are so Converted are held pro rate (as to principal amounts, Types and Interest Periods) in accordance with their respective Commitments.
- 5.05 Communities. The Company shall pay to the Agent for account of each Lender, upon the request of such Lender through the Agent, such amount or amounts as shall be sufficient (in the reasonable opinion of such Lender) to compensate it for any loss, cost or expense which such Lender determines is attributable to:

- (a) any payment or conversion of a Fixed Rate Loan or a Set Rate Loan made by such Londer for any reason (including, without limitation, the acceleration of the Loans pursuant to Section 9 hereof or the conversion of Loans pursuant to Section 5.04 hereof) on a date other than the last day of the Interest Period for such Loan; or
- (b) any failure by the Company for any reason (including, without limitation, the failure of any of the conditions precedent specified in Section 6 hereof to be satisfied) to borrow a Fixed Rate Loan or a Set Rate Loan (with respect to which, in the case of a Money Market Loan, the Company has accepted a Money Market Quote) from such Leader on the date for such borrowing specified in the relevant notice of borrowing given pursuant to Section 2.02 or 2.03(b) hereof.

Without limiting the effect of the preceding sentence, such componention shall include an amount equal to the excess, if any, of (i) the amount of interest which otherwise would have accrued on the principal amount so paid or converted or not borrowed for the period from the date of such payment, conversion or failure to borrow to the last day of the interest Period for such Loan (or, in the case of a failure to borrow, the Interest Period for such Loan which would have commenced on the date specified for such borrowing) at the applicable rate of interest for such Loan provided for herein over (ii) the interest component of the amount such Leader would have bid in the Loadon interbank market (if such Loan is a Beredeller Loan or a LIBOR Market Loan) or the United States escondary certificate of deposit market (if such Loan is a CD Loan or a Set Rate Loan) for Dollar deposits of leading banks in amounts comparable to such principal amount and with materities comparable to such period (as reasonably determined by such Leader).

## 6. Condition Provident.

- the Noise and (ii)
- 8 C at or treasure 3
- Notes, duty
- Les the Company. As opinion of Kenn General Counsel of the Company, substantia 野田田
- 2

### Section 7. <u>Representations and Warranties</u>. The Company represents and warrants to the London that:

- 7.01 Cornerate Existence. The Company: (a) is a corporation duly organized and validly existing under the lews of the State of Florida; (b) has all requisite corporate power, and has all material governmental licenses, authorizations, consents and approvals necessary to own its assets and carry on its business as now being or as proposed to be conducted; and (c) is qualified to do business in all jurisdictions in which the nature of the business conducted by it makes such qualification necessary and where failure so to qualify would have a material adverse effect on the consolidated financial condition, operations or business taken as a whole of the Company and its Consolidated Subsidiaries.
- 7.02 Financial Constitue. The molited consolidated balance sheet of the Company and its Consolidated Subsidiaries as at December 31, 1996 and the related consolidated statements of income, theretolders' equity and cash flow of the Company and its Consolidated Subsidiaries for the floral year ended on said date, with the opinion thereon of KPMG Peat Marwick LLP, and the unswitted consolidated balance sheet of the Company and its Consolidated Subsidiaries as at September 30, 1997 and the related consolidated statements of income and cash flow of the Company and its Consolidated Subsidiaries for the nine-month period ended on such date, heretofore furnished to each of the Londors, are complete and correct and fairly present the consolidated financial condition of the Company and its Consolidated Subsidiaries as at said dates and the consolidated results of their operations for the flocal year and nine-month period ended on said dates (subject, in the case of such financial statements as at September 30, 1997, to normal year-end sudit adjustments), all in accordance with generally accepted accounting principles and practices applied on a consistent basis (provided that such financial statements may contain condensed features prepared in accordance with Rule 10-01(a)(5) of Securities and Enchange Commission Regulation S-X). Neither the Company nor any of its Subsidiaries had on said dates any contingent liabilities, liabilities for taxes, unusual forward or long-term commitments or unrealized or anticipated losses from any unfavorable commitments, in each case material to the Company and its Consolidated Subsidiaries taken as a whole, except as referred to or reflected or provided for in said balance sheets as at said dates. Since September 30, 1997, there has been no material adverse change in the consolidated financial condition, operations or business taken as a whole of the Company and its Consolidated Subsidiaries from thet set forth in said financial statements as at said date.

- 7.03 <u>Litiention</u>. Except for the matters disclosed in the Company's Annual Report on Form 10-K for the year ended December 31, 1996, and the Company's Quarterly Reports on Form 10-Q for the quarters ended March 31, June 30, and September 30, 1997, there are no legal or arbitral proceedings or any proceedings by or before any governmental or regulatory authority or agency, now pending or (to the knowledge of the Company) threatened against the Company or any of the Company's Subsidiaries which, if adversely determined, could have a material adverse effect on the consolidated fluencial condition, operations or business taken as a whole of the Company and its Consolidated Subsidiaries.
- 7.04 No Breach. None of the execution and delivery of this Agreement and the Notes, the communication of the transactions berein contemplated and compliance with the terms and provisions bereof and thereof will conflict with or result in a breach of, or require any consent under, the articles of incorporation or by-laws of the Company, or any applicable law or regulation, or any order, writ, injunction or decree of any court or governmental authority or agency, or any agreement or instrument to which the Company or any of the Company's Subsidieries is a party or by which any of them is bound or to which any of them is subject, or constitute a default under any such agreement or instrument, or result in the creation or imposition of any Lies upon any of the revenues or assets of the Company or any of the Company's Subsidieries pursuant to the terms of any such agreement or instrument.
- 7.05 Comment Action. The Company has all necessary corporate power and authority to execute, deliver and perform its obligations under this Agreement and the Notes; the execution, delivery and performance by the Company of this Agreement and the Notes have been duly authorized by all necessary corporate action on the part of the Company; and this Agreement has been duly and validly executed and delivered by the Company and each such document constitutes, and each of the Notes when executed by the Company and delivered for value will constitute, the legal, valid and binding obligation of the Company, enforceable in accordance with its terms, except to the extent that enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally.
- 7.06 Agreevals. Other than the approval of the Florida Public Service Commission (which approval has been duly obtained and is in full force and effect), no authorizations, approvals or comments of, and no filings or registrations with, any governmental or regulatory authority or agency are necessary for the execution, delivery or performance by the Company of this Agreement or the Notes, or for the validity or enforceability of any thereof.
- 7.07 <u>Use of Local</u>. The Company is not engaged principally, or as one of its important activities, in the business of extending credit for the purpose, whether immediate, incidental or ultimate, of buying or carrying Margin Stock and no part of the proceeds of any Loan hereunder will be used to buy or carry any Margin Stock.

- 24
- company", or a company "commolise" by 7.50 alled. by as 'a 6 À 7 0 D to Bridge
- 2(0)(11) of 0 2(0)(5) 0 1935 Act or 1935 Act or for seeking to be Parent. No Loan will be or regulation thereunder, for Š y Act of 2

Section 8. <u>Covenants of the Company</u>. The Company agrees that, so long as any of the Commissions are in effect and until payment in full of all Louis berounder, all interest thereon and all other amounts payable by the Company berounder:

### 8.01 Financial Statements. The Company shall deliver to each of the Lenders:

- (a) as seen as available and in any event within 60 days after the end of each quarterly fiscal period of each fiscal year of the Company, consolidated statements of income and cash flow of the Company and its Consolidated Subsidiaries for such period and for the period from the beginning of the respective fiscal year to the end of such period, and the related consolidated balance short as at the end of such period, setting forth in each case in comparative form the corresponding consolidated figures for the corresponding period in the preceding fiscal year, accompanied by a certificate of the Treasurer, an Assistant Treasurer, the Chief Financial Officer or the Controller of the Company, which certificate shall state that said financial statements fairly precent the consolidated financial condition and results of operations of the Company and its Consolidated Subsidiaries in accordance with generally accepted accounting principles, consistantly applied, as at the end of, and for, such period (subject to normal year-end saidt adjustments and provided that such financial statements may contain condensed features prepared in accordance with State 10-01(a)(5) of Securities and Enchange Commission Regulation S-X);
- (b) as seen as available and in any event within 120 days after the end of each flacal year of the Company, consolidated statements of income, shareholders' equity and cash flow of the Company and its Consolidated Subsidiaries for such year and the related consolidated balance short as at the end of such year, setting forth in each case in comparative form the corresponding consolidated figures for the preceding flacal year, and accompanied in the case of said consolidated statements and balance short, by an opinion thereon of independent certified public accountants of recognized national standing, which opinion shall state that said consolidated financial statements fairly present the consolidated financial condition and results of operations of the Company and its Consolidated Subsidiaries as at the end of, and for, such flocal year;
- (c) promptly upon their becoming available, copies of all registration statements and regular periodic reports, if any, which the Company shall have filed with the Securities and Exchange Commission (or any governmental agency substituted therefor) or any national accurities exchange;
- (d) promptly upon the mailing thereof to the shareholders of the Parent generally, copies of all financial statements, reports and proxy statements so mailed;
- (e) if any of the events or conditions specified below with respect to any Plan or Multiemployer Plan shall have occurred or exist, promptly upon filing any required notice thereof with PBGC, a copy of such notice or other report to PBGC and

- į
- tion 4041 of ERISA of a notice of inners to of any Plan;
- 00 e 1 08 4042 of 200
- SIS of Base. 1

The Company will furnish to each Leader, at the time it furnishes each set of flancial statements pursuant to puragraph (a) or (b) shove, a certificate of the Treasurer, an Assistant Treasurer, the Chief Flancial Officer or the Controller of the Company to the effect that no Default has accurred and is continuing (or, if any Default has occurred and is continuing, describing the same in reasonable detail and describing the action that the Company has taken and proposes to take with respect thereto).

- 8.02 Literator. The Company will premptly give to each Londor notice (which notice may be given by the Company through the Parent and may be in the form of the Company's or the Parent's 1934 Act Reports) of all legal or arbitral proceedings, and of all proceedings by or before any governmental or regulatory authority or agency, and any material development in respect of such legal or other proceedings, affecting the Company or any of the Company's Subsidiaries, except proceedings which, if adversely determined, would not have a material adverse offect on the consolidated financial condition, operations or business taken as a whole of the Company and its Consolidated Subsidiaries.
- 8.05 Company Existence. Btc. The Company will: preserve and maintain its corporate existence and all of its material rights, privileges and franchises; comply with the requirements of all applicable laws, rules, regulations and orders of governmental or regulatory authorities if failure to comply with such requirements would materially and adversely affect the consolidated fluoristic condition, operations or business taken as a whole of the Company and its Consolidated Substitleries, except for any such laws, rules, regulations or orders that the Company is contesting in good faith and by proper proceedings; pay and discharge all taxes, assessments and governmental charges or levies imposed on it or on its income or profits or on any of its property prior to the date on which possible attach thereto, except for any such tax, assessment, charge or levy the payment of which is being contested in good faith and by proper proceedings and against which adequate reserves are being maintained; maintain all of its properties used or useful in its business in good working order and condition, ordinary wear and tear excepted; and permit representatives of any Lender or the Agent, during normal business hours, to examine, copy and make extracts from its books and records, to impact its properties, and to discuss its business and affairs with its officers, all to the extent reasonably requested by such Lender or the Agent (as the case may be).
- 8.04 <u>Prohibition of Fundamental Changes</u>. The Company will not enter into any transaction of merger or consolidation or smalgametion, or liquidate, wind up or dissolve itself (or suffer any liquidation or dissolving), except that the Company may merge into the Parent so long as the Parent is the surviving corporation and assumes all of the Company's obligations under this Agreement and the Notes and so long as both prior to such merger and after giving office thereto no Definit shall have occurred and be continuing. The Company will not convey, sell, lease, transfer or otherwise dispose of, in one transaction or a series of transaction, all or any substantial part of its business or assets, whether now owned or hereafter acquired.

- Contalization of the Con -0 ō 1200
- 59
- on Oceanor 31, 1997 in an
- my Lieu existing on any sees of any conparation at the time such an a Sabridiary of the Company and and creamed in communication of
- y asset of any corporation existing at the time such
- existing on any asset prior to the acquisition thereof by the
- wai or refus
- et, distraint or similar of as the essention or red thereby are being

- (h) Lieus incidental to conduct of its business or the ownership of its assets which (i) do not secure Indebtedness and (ii) do not in the aggregate materially detract from the value of its assets or materially impair the use thereof in the operation of its business:
  - (i) the Lieus created by the Existing Indonture; and
- (j) Lieus upon righto-of-way for transmission or distribution line purposes, provided that the Company or one of its Consolidated Subsidiaries, as the case may be has, in the opinion of counsel for the Company, power under emissest domain or similar statutes to condemn and acquire ensembles or righto-of-way sufficient for its purposes over the land covered by the righto-of-way in question or other lands adjacent thereto.
- Section 9. Events of Default. If one or more of the following events (herein called "Events of Default") shall occur and be continuing:
  - (a) The Company shall default in the payment when due of any principal of or interest on any Loan or any other amount psychle by it herounder; or
  - (b) The Company or any of its Subsidiaries shall default in the payment when due of any principal of or inserest on any of its other indebtedness aggregating \$10,000,000 or more; or any event specified in any note, agreement, indenture or other document evidencing or relating to any such indebtedness in the aggregate amount set forth above in this clause (b) shall occur if the effect of such event is to cause, or (with the giving of any notice or the lapse of time or both) to permit the holder or holders of such indebtedness (or a trustee or agent on behalf of such holder or holders) to cause, such indebtedness to become due, or to be propaid in full (whether by redemption, purchase or otherwise), prior to its stated materity; or
  - (c) Any representation, warranty or certification made or deemed made herein (or in any modification or supplement hereto) by the Company, or any certificate furnished to any Londor or the Agent pursuant to the provisions hereof (or thereof), shall prove to have been false or misleading as of the time made or furnished in any material respect; or
  - (d) The Company shall default in the performance of any of its obligations under any of Sections 8.01(f) or 8.04 hereof; or the Company shall default in the performance of any of its other obligations in this Agreement and such default shall continue unramedied for a period of 30 days after action thereof to the Company by the Agent or any Leader (through the Agent); or any "Event of Default" under and as defined in the other FPC Agreement shall be continuing; or

- (e) The Company or any of the Company's Subsidiaries shall admit in writing its instillity to, or be generally unable to, pay its debts as such debts become due; or
- (f) The Company or any of the Company's Subsidiaries shall (i) apply for or consent to the appointment of, or the taking of possession by, a receiver, custodian, trustee or liquidator of itself or of all or a substantial part of its property, (ii) make a general assignment for the benefit of its creditors, (iii) commence a voluntary case under the Bankruptcy Code (as now or hereafter in effect), (iv) file a petition seeking to take advantage of any other law relating to bankruptcy, insolvency, reorganization, winding-up, or composition or readjustment of debts. (v) fail to controvert in a timely and appropriate manner, or acquisece in writing to, any petition filed against it in an involuntary case under the Bankruptcy Code (as now or hereafter in effect), or (vi) take any corporate action for the purpose of effecting any of the foregoing; or
- (g) A proceeding or case shall be commenced, without the application or consent of the Company or any of the Company's Subsidiaries, in any court of competent jurisdiction, seeking (i) its liquidation, reorganization, dissolution or winding-up, or the compensition or readjustment of its debts, (ii) the appointment of a trustee, receiver, custodian, liquidator or the like of the Company or such Subsidiary or of all or any substantial part of its assets, or (iii) similar relief in respect of the Company or such Subsidiary under any law relating to bankruptcy, insolvency, reorganization, winding-up, or composition or adjustment of debts, and such proceeding or case shall continue undiamissed, or an order, judgment or decree approving or ordering any of the foregoing shall be exceed and continue unstayed and in effect, for a period of 60 or more days; or an order for relief against the Company or such Subsidiary shall be extered in an involuntary case under the Bankruptcy Code (as now or hereafter in effect); or
- (h) A final judgment or judgments for the payment of money in excess of \$10,000,000 in the aggregate shall be rendered by a court or courts against the Company and/or any of the Company's Subsidiaries and the same shall not be discharged (or provision shall not be made for such discharge), or a stay of execution thereof shall not be procured, within 30 days from the date of entry thereof and the Company or the relevant Subsidiary shall not, within said period of 30 days, or such longer period during which execution of the same shall have been stayed, appeal therefrom and cause the execution thereof to be stayed during such appeal; or
- (i) An event or condition specified in Section 8.01(e) hereof shall occur or exist with respect to any Plan or Multiemployer Plan and, as a result of such event or condition, together with all other such events or conditions, the Company, the Parent or any ERISA Affiliate shall incur or in the opinion of the Majority Lenders shall be reasonably likely to incur a liability to a Plan, a Multiemployer Plan or PBGC (or any combination of the foregoing) which is, in the determination of the Majority Lenders, material in relation to the consolidated financial condition, operations or business taken

as a whole of the Company and its Consolidated Subsidiaries or of the Parent and its Consolidated Subsidiaries; or

- (j) Any authorization, approval or consent of any governmental or regulatory authority or agency accessary for the execution, delivery or performance by the Company of this Agreement or the Notes, or for the validity or enforceability thereof, shall cease to be in full force and effect; or
- (k) The Company shall cease to be a Wholly-owned Subsidiary of the Parent;

THEREUPON: (1) in the case of an Event of Default other than one referred to in clause (1) or (g) of this Section 9 with respect to the Company, (A) the Agent may and, upon request of the Majority Londors, shall, by notice to the Company, cancel the Commitments and they shall thereupon terminate, and (B) the Agent may and, upon request of the Majority London shall, by notice to the Company declare the principal amount then outstanding of, and the accrued interest on, the Loans and all other amounts payable by the Company berounder and under the Notes (including, without limitation, any amounts payable under Section 5.05 hereof) to be forthwith due and payable, whereupon such amounts shall be immediately due and payable without presentment, domand, present or other formalities of any kind, all of which are hereby expressly waived by the Company; and (2) in the case of the occurrence of an Event of Default ferred to in clause (f) or (g) of this Section 9 with respect to the Company, the Commitments shall automatically be cancelled and the principal amount then outstanding of, and the accrued interest on, the Loans and all other amounts payable by the Company horounder and under the Notes (including, without limitation, any amounts payable under Section 5.05 hereof) shall automatically become immediately due and payable without presentment, domand, protest or other formalities of any kind, all of which are hereby expressly waived by the Company.

### Section 10. The Agent.

appoints and authorizes the Agent to act as its agent herounder with such powers as are specifically delegated to the Agent by the terms of this Agreement, together with such other powers as are reasonably incidental thereto. The Agent (which term as used in this sentence and in Section 10.05 and the first sentence of Section 10.05 hereof shall include reference to its affiliates and its own and its affiliates' officers, directors, employees and agents): (a) shall have no duties or responsibilities except these expressly set forth in this Agreement, and shall not by reason of this Agreement be a trustee for any Lender; (b) shall not be responsible to the Lenders for any recitals, entennest, representations or warranties contained in this Agreement, or in any certificate or other document referred to or provided for in, or received by any of them under, this Agreement, or for the value, validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement, any Note or any other document referred to or provided for herein or for any failure by the Company or any other focument referred to or provided for herein or thereunder; (c) shall not be required to initiate or conduct any litigation or

- pon ay
- or notice

10.05 Indemnification. The Leaders agree to indemnify the Agent (to the extent not reimbursed under Section 11.03 hereof, but without limiting the obligations of the Company under said Section 11.03) ratably in accordance with the aggregate amount of their respective Commitments, for any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind and nature whatsoever which may be imposed on, incurred by or asserted against the Agent in any way relating to or arising out of this Agreement or any other documents contemplated by or referred to herein or the transactions contemplated hereby (including, without limitation, the costs and expenses which the Company is obligated to pay under Section 11.03 hereof but excluding, unless a Default has occurred and is continuing, normal administrative costs and expenses incident to the performance of its agency duties horomater) or the enforcement of any of the terms hereof or of any such other documents, provided that no Leader shall be liable for any of the foregoing to the extent they arise from the green negligance or willful misconduct of the party to be indomnified.

10.06 Non-Beliance on Agent and Other Landers. Each Londer agrees that it has, independently and without reliance on the Agent or any other Londer, and based on such documents and information as it has deemed appropriate, made its own credit analysis of the Company and its Subsidiaries and decision to enter into this Agreement and that it will, independently and without reliance upon the Agent or any other Londer, and based on such documents and information as it shall does appropriate at the time, continue to make its own analysis and decisions in taking or not taking action under this Agreement. The Agent shall not be required to keep itself informat as to the performance or observance by the Company of this Agreement or any other document referred to or provided for berein or to impact the properties or books of the Company or any of its Subsidiaries. Encept for notices, reports and other documents and information expressly required to be furnished to the Londers by the Agent herounder, the Agent shall not have any duty or responsibility to provide any Lender with any credit or other information concerning the affairs, flasnoial condition or business of the Company or any of its Subsidiaries (or any of their affiliates) which may come into the possession of the Agent or any of its affiliates.

10.07 Failure to Act. Except for action expressly required of the Agent hereunder, the Agent shall in all cases be fully justified in failing or refusing to act hereunder unless it shall receive further assurances to its satisfaction from the Lenders of their indemnification obligations under Section 10.05 hereof against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action.

10.08 <u>Resignation or Removal of Agest</u>. Subject to the appointment and acceptance of a successor Agest as provided below, the Agest may resign at any time by giving notice thereof to the Lenders and the Company, and the Agest may be removed at any time with or without cause by the Majority Lenders. Upon any such resignation or removal, the Majority Lenders shall have the right to appoint a successor Agest. If no successor Agest shall have been so appointed by the Majority Lenders and shall have accepted such appointment within 30 days after the retiring Agest's giving of notice of resignation or the Majority Lenders' removal of the retiring Agest, then the retiring Agest may, on behalf of the Lenders, appoint a successor

Agent, which shall be a bank which has an office in New York, New York and (unless it is a Leader) a combined capital and surplus of at least \$200,000,000. Upon the acceptance of any appointment as Agent horounder by a successor Agent, such successor Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Agent, and the retiring Agent shall be discharged from its duties and obligations berounder. After any retiring Agent's resignation or removal horounder as Agent, the provisions of this Section 10.08 shall continue in effect for its benefit in respect of any actions taken or omitted to be taken by it while it was acting as the Agent.

10.09 Agency Fig. So long as the Commitments are in effect and until payment in full of the principal of and interest on the Louis and all other amounts payable by the Company horounder, the Company will pay to the Agent an agency fee in an amount previously agend, payable annually in advance on the Quarterly Date falling on or nearest to December 1 in each year. Such fee, once paid, shall be non-refundable. The appointment of a successor Agent under Section 10.08 hereof shall not increase or otherwise modify the Company's obligations under this Section 10.09.

10.10 <u>Auction For.</u> The Company agrees to pay to the Agent an auction fee in an amount previously agreed, payable on each day that the Company delivers a Money Market Quete Request.

### Section 11. Miscellegeous.

11.01 <u>Waiver</u>. No failure an the part of the Agent or any Londor to exercise and no delay in exercising, and no course of dealing with respect to, any right, power or privilege under this Agreement or any Note shall operate as a waiver thereof, nor shall any single or partial exercise of any right, power or privilege under this Agreement or any Note proclude any other or further exercise thereof or the exercise of any other right, power or privilege. The remedies provided herein are cumulative and not exclusive of any remedies provided by law.

(including, without limitation, any modifications of, or waivers or consents under, this Agreement) shall be given or made by telex, telecopy, telegraph, cable or in writing (or, with respect to notices given pursuant to Section 2.03 hereof, by telephone, confirmed in writing by telex by the close of business on the day the notice is given) and telexed, telecopied, telegraphed, cabled, mailed or delivered (or telephoned, as the case may be) to the intended recipient at the "Address for Netices" specified below its name on the signature pages hereof; or, as to any party, at such other address as shall be designated by such party in a notice to each other party. Except as otherwise provided in this Agreement, all such communications shall be designed to have been duty given when transmitted by telex or telecopier, delivered to the telegraph or cable office or personally delivered or, in the case of a mailed notice, upon receipt, in each case given or addressed as aforesaid.

### 11.03 Errosse, Frc.

- (a) The Company agrees to pay or reimburse each of the Lenders and the Agent for all reasonable out-of-pocket costs and expenses of the Agent (including, without limitation, the reasonable fees and expenses of counsel to the Agent), in connection with the negotiation, preparation, execution and delivery of this Agreement and the Notes and the making of the Loans bereunder.
- (b) The Company agrees to pay or reimburse each of the Lenders and the Agent for (i) all reasonable out-of-pocket costs and expenses of the Agent (including reasonable counsel's fees) in connection with any amendment, prodification or waiver of any of the terms of this Agreement or any of the Notes and (ii) all costs and expenses of the Lenders and the Agent (including reasonable counsels fees for each Lender and for the Agent) in connection with any Default and any enforcement or collection proceedings resulting therefrom.
- (c) The Company agrees to pay all transfer, stamp, documentary or other similar tense, assessments or charges levied by any governmental or revenue authority in respect of this Agreement or any of the Notes or any other document referred to herein.
- Agreement, any provision of this Agreement may be amended or modified only by an instrument in writing signed by the Company, the Agent and the Majority Londors, or by the Company and the Agent acting with the consent of the Majority Londors, and any provision of this Agreement may be waived by the Majority Londors or by the Agent acting with the consent of the Majority Londors; provided that no amendment, modification or waiver shall, unless by an instrument signed by all of the Londors or by the Agent acting with the consent of all of the Londors: (i) extend the term, or extend the time or waive any requirement for the reduction or termination, of, or increase, the Commitment (except to the extent contemplated by the definition of "Commitment Termination Dute" in Section 1.01 hereof), (ii) extend the date fixed for the payment of principal of or interest on any Loan or any fees or other amounts payable hereunder, (iii) reduce the amount of any payment of principal thereof or the rate at which interest is payable thereon or any fee is payable hereunder, (iv) alter the terms of Section 8.06 or 11.06(a) hereof or of this Section 11.04, or (v) amend the definition of the term "Majority Lenders"; and provided, further, that any amendment of Section 10 hereof shall also require the consent of the Agent.
- 11.05 Successors and Assigns. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns.

### 11.06 Assissment and Participations.

(a) The Company may not assign its rights or obligations herounder or under the Notes without the prior consent of all of the Leaders and the Agent.

No Leader may assign any of its Loans, its Notes, or its Commitment in whole or in part without the prior consent of the Company and the Agent (such consent of the Agent not to be unreasonably withheld); provided that (i) each such assignment by a Lender of its Syndicated Loans or its Commitment shall be made so that the assignee holds the same pro-rate portions of the Syndicated Loans and the Commitments; and (ii) each such assignment by a Leader of its Syndicated Loans and Commitment shall be made concurrently with an assignment by it to the same assignee of the same pro rate portion of its outstanding "Syndicated Losss' and its "Commitment" under and as defined in the Other FPC Agreement. Upon execution and delivery by the assignee to the Company and the Agent of an instrument in writing pursuant to which such assignee agrees to become a "Lender" herounder (if not already a Lender) having the Commitment(s) and Leans specified in such instrument, and upon consent thereto by the Company and the Agent to the extent required above, the assignee shall have, to the extent of such assignment (unless otherwise provided in such assignment with the consent of the Company and the Agent), the obligations, rights and benefits of a Lender herounder holding the Commisment(s) and Loans (or portions thereof) assigned to it (in addition to the Commitment(s) and Loans, if any, theretofore held by such assignee) and the assigning Londor shall, to the entent of such assignment, be released from the Commitment(s) (or portion(s) thereof) so essigned. In connection with any assignment pursuant to this Section 11.06(b), the Company agrees to issue replacement Notes in the State of New York, in the appropriate principal amounts and psychic to the appropriate Londors, upon the request of any assigning Londor or assignment Londor. Upon each such assignment the assigning Londor shall pay the Agent an assignment fee of \$3,500.

A Londor may sell or agree to sell a participation in all or any part of any Loan held by it or Loans made or to be made by it. In the event of such a participation, no such participant shall have any rights or benefits under this Agreement or any Note (the participant's rights against such Londor in respect of such participation to be those set forth in the agreement (the "Participation Agreement") executed by such Londor in favor of the participant). The granting of a participation by a Londor horounder shall not in any way release such Londor from any of its obligations under this Agreement. All amounts payable by the Company to any Londor under Section 5 hereof shall be determined as if such Londor had not ligations under this Agreement. All amounts payable by the sold or agreed to sell any participations in such Loan and as if such Londor were funding all of such Loss in the same way that it is fending the portion of such Loss in which no participations have been sold. In no event shall a Londor that sells a participation be obligated to the participant under the Participation Agreement to take or refrain from taking any action hereunder or under such Lender's Notes except that such Lender may agree in the Participation Agreement that it will not, without the consent of the participant, agree to (i) the extension of any date fixed for the payment of principal of or interest on the related Loan or Loans or any portion of any fees payable to the participant, (ii) the reduction of any payment of principal thereof, or (iii) the reduction of the rate at which either interest is payable thereon or (if the participant is entitled to any part thereof) facility fee is psyable herounder to a level below the rate at which the participant is entitled to receive interest or facility fee (as the case may be) in respect of such participation.

- (d) Anything in this Section 11.06 to the contrary notwithstanding, any Lender may assign and pledge all or any portion of its Loans and its Notes to any Federal Reserve Bank as collected accurity pursuant to Regulation A of the Board of Governors of the Federal Reserve System and any Operating Circular issued by such Federal Reserve Bank. No such assignment shall release the assigning Londer from its obligations bereunder.
- (e) A Londor may furnish any information concerning the Company, the Parent or any of their respective Subsidiaries in the possession of such Londor from time to time to assignees and participants and, with notice to the Company, to prospective participants and, with the consent of the Company, to prospective assignees.
  - (D If:
  - (i) any Lender makes a demand for payment under Section 5.01 hereof.
  - (ii) Loans of any Londor are Converted into Base Rate Loans pursuant to Section 5.04 bereof, or
  - (iii) any Londor does not become a Consuming Londor (as that term is used in the definition of "Commitment Termination Date" in Section 1.01 hereof), does not become a "Consenting Londor" under (and as defined in) the Other FPC Agreement or either of the PCH Agreements, in each case within the required pariods set forth in the relevant definitions thereof;

then, in the case of clause (i) above, within 60 days after the Company makes the payment demanded; in the case of clause (ii) above, within 60 days after the last of the relevant Conversions; and, in the case of clause (iii) above, at any time thereafter, the Company may, so long as no Definit shall be continuing and subject to the consent of the Agent (which consent shall not be unreasonably withhold), demand that such Londor assign, pursuant to this Section 11.06 and documentation reasonably acceptable to such Londor, all (but not less than all) of such Londor's Commitment and outstanding Loans hereunder to one or more banks or flasacial institutions designated by the Company, for a purchase price not less than the principal amount of such outstanding Loans, accrued interest thereon and all other amounts payable by the Company to such Londor hereunder (including amounts payable under Section 5.05 hereof) and under the Notes hold by such Londor, such assignment to take place no later than 30 days after the Company's demand.

- 11.07 <u>Survival</u>. The obligations of the Company under Sections 5.01, 5.05 and 11.03 hereof shall survive the repayment-of the Loans and the termination of the Commitments.
- 11.08 <u>Captions</u>. The table of contents and captions and section headings appearing herein are included solely for convenience of reference and are not intended to affect the interpretation of any provision of this Agreement.

- 11.09 Counterparts. This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same instrument and any of the parties hereto may execute this Agreement by signing any such counterpart.
- 11.10 Governing Law: Submission to Jurisdiction. This Agreement and the Notes shall be governed by, and construed in accordance with, the law of the State of New York. The Company hereby submits to the nonexclusive jurisdiction of the United States District Court for the Senthern District of New York and of any New York state court sitting in New York City for the purposes of all legal proceedings arising out of or relating to this Agreement or the transactions contemplated hereby. The Company irrevocably waives, to the fullest extent permitted by law, any objection which it may now or hereafter have to the laying of the venue of any such proceeding brought in such a court and any claim that any such proceeding brought in such a court and any claim that any such proceeding brought in an inconvenient forum.
- 11.11 Water of lary Itial. EACH OF THE COMPANY, THE AGENT AND THE LENGTH BETTENY BREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARMING OUT OF OR RELATING TO THE AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

IN WITNESS WHEREOF, the parties bareto have caused this Agreement to be duly executed as of the day and year first above written.

FLORIDA POWER CORPORATION

or Hamela a. Thou

Passols A. Sasri Assistant Treasurer

Address for Notices:

Post Office Box 14042

St. Potersburg, Florida 33733

Telecopier No.:

813/866-4021

Telephone No.:

813/866-4400

Attention: Mr. Konneth E. McDonald

Sworm to and subscribed before me on this 17th day of November, 1997, in the State of New York, County of New York.

Many Malle Mayor

CAROL A. MAYERS
NOTARY PUBLIC. State of New York
No. 0104n5004316
Qualified in Brons County
Commence Lapares Seet. 2, 1999

p: lendt approvate.

\$50,225,000

### THE CHASE MANHATTAN BANK

Tide: PRIAL V. PARRIELL

Leading Office for all Loans (other than Eurodolfer Luans):

The Chase Manhattan Bank 270 Park Avenue New York, New York 10017-2070

### Leading Office for Burodollar Losse:

The Chase Manhattan Bank Cayman Islands, British West Indies Breach c/o The Chase Manhattan Bank Our Chase Manhattan Plans New York, New York 10081

### Address for Notices:

The Chase Manhettan Bank Global Power & Environmental Group 270 Park Avenue New York, New York 10017-2070

Telecopier No.: 212/270-3089

Telephone No.: 212/270-7653

Attention: Mr. Paul V. Farrell

\$43,325,000

MORGAN GUARANTY TRUST COMPANY OF NEW YORK

Dy \_

Philip S. Dojess Vice President

Loading Office for all Loans (other than Eurodolfar Loans):

Morgan Guaranty Trust Company of New York 60 Wall Street New York, New York 10260-0060

### Leading Office for Buredollar Loans:

Morgan Guaranty Trust Company of New York Nesses, Bahamas c/o J. P. Morgan Services, Inc. Bure-Lean Securities Unit 902 Martes Street Wilmington, Delaware 19801

### Address for Notices:

Morgan Guaranty Trust Company of New York 60 Wall Street New York, New York 10260-0060

Telecopier No.: 212/648-5014

Telephone No.: 212/648-8454

Attention: Mr. Phillip S. Detjons

Commission \$43,325,000

NATIONSBANK, N.A.

Tide: SVP

Leading Office for all Loans: Nationallank, N.A. 101 North Tryon Street Charlotte, NC 28295 Atts: Judy Dudley, Credit Services

Address for Notices:

Nationallank, N.A. 21st Floor 600 Peachtree Street, N.E. Atlanta, GA 30308-2213

Telecopier No.: 404/607-6465

Telephone No.: 404/607-5561

Attention: Mr. McKie M. Trotter II

S43,325,000

FIRST UNION NATIONAL BANK

y\_//

Title:

MICHAEL J. KOLOSOWSKY

Leading Office for all Loans

First Union National Bank
One First Union Center
301 South College Street
Charlotte, North Carolina 28288-0745

Address for Notices:

First Union National Bank One First Union Conter 301 South College Street Charlotte, North Carolina 28288-0745

Telecopier No.

704/374-2802

Telephone No.:

704/383-7686

Amenica: Mr. David M. Johnson

S33,300,000

SUNTRUST BANK, TAMPA BAY

Title: SVC

Loading Office for all Loans:

SunTrust Bank, Tampa Bay 300 1st Avenue South St. Petersburg, Florida 33701

Address for Notices:

SunTrust Bank, Tampa Bay 300 1st Avenue South St. Petersburg, Florida 33701

Telecopier No.:

813/892-4810

Telephone No.:

813/892-4958

Amention: Ms. Brigitta A. Lawton

Sworn to and subscribed before me on this 17th day of November, 1997, in the State of New York, County of New York.

ry Public

CAROL A MAYERS MOTARY PUBLIC, State of New York No. 91MASSEA316 Qualities in Broke County S32,000,000

### THE FIRST NATIONAL BANK OF CHICAGO

William > Bowley

Tide: AUTHORIZED AGENT

### Leading Office for all Loans:

The First National Bank of Chicago One First National Plans Suite 0363 Chicago, Illinois 60670-0363

### Address for Notices:

The First National Bank of Chicago One First National Plans Suite 0363 Chicago, Illinois 60670-0363

Telecopier No.: 312/732-3055

Telephone No.: 312/732-9781

Attention: Mr. William N. Benks

\$32,000,000

PNC BANK, NATIONAL ASSOCIATION

Title:

Leading Office for all Losse:

PNC Bank, National Association One PNC Plans 3rd Floor 249 - 5th Avenue Pinsburgh, PA 15222-2707

Address for Notices:

PNC Bank, National Association One PNC Plans 3of Plans 249 - 5th Avenue Planburgh, PA 15222-2707

Telecopier No.: 412/762-2571

Telephone No.: 412/762-2540

Assestion: Mr. Christopher N. Moravec

S22,500,000

BARNETT BANK, N.A., PINELLAS COUNTY

Title: SYP

Leading Office for all Loans:

Barnett Bank, N.A., Pinellas County 200 Central Avenue Suite 1800 St. Petersburg, Florida 33701

Address for Notices:

Barnett Bank, N.A., Pinellas County 200 Control Avenue Suite 1800 St. Petersburg, Florida 33701

Telecopier No.:

813/892-1545

Telephone No.:

813/892-1518

Attention: Mr. Michael S. Crowe

Sworn to and subscribed before me on this 17th day of November, 1997, in the State of New York, County of New York.

MOTARY PUBLIC, State of few Yark No. 016A5565316 Qualitation or Brown County THE CHASE MANHATTAN BANK as Agent

By And

PAUL V. PARRELL VICE PRESENT

Address for Notices to Chase as Agent:

The Chase Manhettan Bank
1 Chase Manhettan Plaza - 8th Floor
New York, New York 10061

Telepopier No.:

212/552-7490

Telephone No.

212/552-7943

Assession: Mr. Municum Appeare

# [Form of Note for Syndicated Longs]

### PROMISSORY NOTE

	1	•
	1	
NEW YORK, NEW YO		_

FOR VALUE RECEIV ED, FLORIDA POWER CORPORATION, a Flo

at Period (if applicable) of

the "Credit

The Credit Agreement provides for the acceleration of the maturity of this Note upon the occurrence of certain events and for prepayments of Louis upon the terms and conditions specified therein.

Except as permitted by Section 11.06(b) of the Credit Agreement, this Note may not be assigned by the Lender to any other Person.

This Note shall be governed by, and construed in accordance with, the law of the State of New York.

FLORIDA POWER CORPORATION

By		
Title:		

### SCHEDULE OF LOANS

This Note Schodule describes Loans made, Continued or Converted under the withindescribed Credit Agreement to the Company, on the dates, in the principal amounts, of the Types, bearing interest at the rates and maturing on the dates set forth below, subject to the payments and prepayments of principal set forth below:

Date Made							
Cont-	Principal			Duration	Date and		
inued	Amount	Type		of	Amount	Uspeid	
or Con-	of		Interest	Laterest	Paid or	<b>Principal</b>	Notation
verted	Loss	Loss	_Batt	Period	Prepaid	Amount	Made by

### [Form of Note for Money Market Loans]

### **PROMISSORY NOTE**

New York, New York

The date, amount, Type, interest rate and maturity date of each Money Market Loan made by the Londor to the Company, and each payment made on account of the principal thereof, shall be recorded by the Londor on its books and, prior to any transfer of this Note, endorsed by the Londor on the schedule attached hereto or any continuation thereof, provided that any failure by the Londor to make any such endorsement shall not affect the obligations of the Company hereunder.

This Note is one of the Notes referred to in the Second Amended and Restated Credit Agreement A (as modified and supplemented and in effect from time to time, the "Credit Agreement") deted as of November 18, 1997 between the Company, the lenders named therein (including the Lender) and The Chase Manhattan Bank, as Agent, and evidences the Company's obligation to repay the Money Market Loans made by the Lender thereunder and interest thereon. Capitalized terms used in this Note have the respective meanings assigned to them in the Credit Agreement.

The Credit Agreement provides for the acceleration of the maturity of this Note upon the occurrence of certain events and for prepayments of Loans upon the terms and conditions specified therein.

Except as permitted by Section 11.06(b) of the Credit Agreement, this Note may not be assigned by the Lender to any other Person.

This Note shall be governed by, and construed in accordance with, the law of the State of New York.

FLORIDA POWER CORPORATION

7	L Par		
Title:			

### SCHEDULE OF LOANS

This Note Schedule describes Loans made under the within-described Credit Agreement to the Company, on the dates, in the principal amounts, of the Types, bearing interest at the rates and maturing on the dates set forth below, subject to the payments and propayments of principal set forth below:

	Principal				Date and		
Date	Amount	Type		Maturity	Amount	Umpaid	
of	of	of	Interest	Date of	Paid or	Principal	<b>Notation</b>
Loss	Long	Loss	_lan_	Loss	Present	Amount	Made by

### [Form of Opinion of Counsel to the Parent]

. 199

To: the Lenders party to the Credit Agreement referred to below and The Chase Manhattan Bank, as Agent

### Ladies and Gentlemen:

I am Vice President and General Counsel of Florida Power Corporation (the "Company"), a wholly owned subsidiary of Florida Progress Corporation (the "Parent"), and am rendering this opinion in connection with the Credit Agreement A dated as of November 26, 1991 (the "Original Agreement"), between the Company, the leaders named therein and The Chase Manhattan Bank, as Agent, as amended and restated by the Second Amended and Restated Credit Agreement A dated as of November 18, 1997 (the Original Agreement, as amended and restated, being hereinafter referred to as the "Credit Agreement") providing for loans to be made by said lenders to the Company in an aggregate principal amount not exceeding \$300,000,000. I have represented the Company in connection with the negotiation of the Credit Agreement. Terms defined in the Credit Agreement are used herein as defined therein.

In rendering the opinion expressed below, I have examined the originals or conformed copies of such corporate records, agreements and instruments of the Company and the Parent, certificates of public officials and of officers of the Company and the Parent, and such other documents and records, and such matters of law, as I have deemed appropriate as a basis for the opinions hereinafter expressed.

### Based upon the foregoing, I am of the opinion that:

- The Company is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Florida and has the necessary corporate power to make and perform the Credit Agreement and the Notes (collectively, the "Credit Documents") and to borrow under the Credit Agreement.
- 2. The making and performance by the Company of the Credit Agreement and the borrowings thereunder have been duly authorized by all necessary corporate action, and do not and will not violate any provision of law or regulation or any provision of its articles or by-laws or result in the breach of, or constitute a default or require any consent under, or result in the creation of any Lieu upon any of its properties, revenues or assets pursuant to, any indenture or other agreement or instrument to which the Company or any of its

Subsidiaries is a party or by which the Company or any of its Subsidiaries or their respective properties may be bound.

- The Credit Agreement has been duly executed and delivered by the Company and constitutes, and the Notes when executed and delivered for value will constitute, legal, valid and binding obligations of the Company, enforceable in accordance with their respective terms, except as such enforceability may be limited by (a) bankruptcy, insolvency, reorganization, moratorium or other similar laws of general applicability affecting the enforcement of creditors' rights and (b) the application of general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law), and except that no opinion is expressed as to (i) Section 4.07(c) of the Credit Agreement or (ii) the effect of the law of any jurisdiction (other than the State of Florida) wherein any Lender (including any of its Applicable Lending Offices) may be located which limits rates of interest which may be charged or collected by such Leader. I express no opinion as to (i) whether a Federal or state court outside of the State of New York would give effect to the choice of New York law provided for in the Credit Agreement and the Notes, (ii) the second sentence of Section 11.10 of the Credit Agreement, insofar as such sentence relates to the subject matter jurisdiction of the United States District Court for the Southern District of New York to adjudicate any controversy related to the Credit Agreement or the Notes. (iii) the waiver of inconvenient forum set forth in Section 11.10 of the Credit Agreement with respect to proceedings in the United States District Court for the Southern District of New York, or (iv) Section 11.11 of the Credit Agreement.
- 4. In connection with the above, I wish to point out that provisions of the Credit Agreement that permit the Agent or any Londor to take action or make determinations, or to benefit from indomnities and similar undertakings of the Company, may be subject to a requirement that such action or inaction by the Agent or a Londor which may give rise to a request for payment under such an undertaking be taken or not taken, on a reasonable basis and in good faith.
- 5. Except for the matters disclosed (i) under the heading "Legal Proceedings" in part I, Item 3 of the Company's Annual Report on Form 10-K for the year ended December 31, 1996, and (ii) under the heading "Legal Proceedings" in Part II, Item 1 of the Company's Quarterly Reports on Form 10-Q for the quarters ended March 31, June 30, and September 30, 1997, there are no legal or arbitral proceedings, and no proceedings by or before any governmental or regulatory authority or agency, pending or (to my knowledge) threatened against or affecting the Company or any of the Company's Subsidiaries, or any properties or rights of the Company or any of the Company's Subsidiaries, which, if adversely determined, would have a material adverse effect on the compolidated financial condition, operations or business taken as a whole of the Company and its Consolidated Subsidiaries.
- Other than the approval of the Florida Public Service Commission (which
  approval has been duly obtained and is in full force and effect), no authorizations, consents,
  approvals, licenses, filings or registrations with, any governmental or regulatory authority

0 belivery or performance by the

1935 Act of on Form U-3A-2 under I 72 Miss boldi 0 y virtue of the filling of an exert 3(a)(1) of the 1935 Act. Suci or with all applicable rules and regulations or the 1935 Act and was filed on March 3, all of the requireme aing of Section 2(a)(7) of the all of the requirements of the licable rules and regulations Such Form U-3A-2

e 1935 Act o a 3(a)(1) of the 1935 Act. Commission pursuant to Rule 6 7. d by such filing may be terminated by the Dule 6 under the 1935 Act thirty days a Commission by registered mail to the exists as to whether or not the Parest is may arise should the Socurities and e 1935 Act) I am not aware of any y registered mail to the fiber or not the Parent is of the 1935 Act or any afforded by Rule 2 under he of the Parent to file opticable section of the for by the 1935 Act erest of investors or Excha

None of the L the Company in the transactions contemplated by the Credit be embject to regulation by any governmental authority as an "a "public utility company", a "holding company" or a stilling, of any of the foregoing under the 1935 Act.

pition as to any mattern governed to and the Federal low of the Unite r of the State of Florida and I do not herein intend to express on according by any laws other than the law of the State of # of A

Very truly yours.

### [Form of Money Market Quote Request]

To:	The Chase Manhattan Bank, as Agent	1

From: Floride Power Corporation, Inc.

Re: Money Market Quote Request

Pursuant to Section 2.03 of the Second Amended and Restated Credit Agreement A (the "Credit Agreement") deted as of November 18, 1997, as amended or restated from time to time, between Florida Power Corporation, the leaders named therein and The Chase Manhattan Bank, as Agent, we hereby give notice that we request Money Market Quotes for the following proposed Money Market Borrowing(s):

Borrowing Quotation Interest

Date Date(\*1) Amount(\*2) Type(\*3) Period(\*4)

Terms used herein have the meanings assigned to them in the Credit Agreement.

FLORIDA POWER CORPORATION

By\_\_\_\_\_\_Title:

<sup>\*</sup> All numbered footnotes appear on the last page of this Exhibit.

- [1] For use if a Money Market Rate in a Set Rate Auction is requested to be submitted before the Borrowing Date.
- [2] Each amount must be \$10,000,000 or a larger multiple of \$1,000,000.
- [3] Insert either "Margin" (in the case of LIBOR Market Loans) or "Rate" (in the case of Set Rate Loans).
- [4] One, two, three or six months, in the case of a LIBOR Market Loan or, in the case of a Set Rate Loan, a period of up to 180 days after the making of such Set Rate Loan and ending on a Business Day.

### [Form of Money Market Quote]

To:	The	Change	Manhettan	Beek.	-	Anne
	-	the latest spine			-	

Attention:

Re: Money Market Quote to

Florida Power Corporation (the "Borrower")

This Money Market Quote is given in accordance with Section 2.03(c) of the Second Amended and Restated Credit Agreement A (the "<u>Credit Agreement</u>") dated as of November 18, 1997, as amended or restated from time to time, between Florida Power Corporation, the leaders named therein and The Chase Manhettan Bank, as Agest. Terms defined in the Credit Agreement are used berein as defined therein.

In response to the Berrower's invitation dated \_\_\_\_\_\_, 19\_\_, we hereby make the following Money Market Quote(s) on the following terms:

- 1. Quoting Londor:
- 2. Person to contact at Quoting Lender:
- We hereby offer to make Money Market Loan(s) in the following principal amount(s), for the following Interest Period(s) and at the following rate(s):

Borrowing	Quotation		Interest		
		Amount[*2]	Type(*3)	Period[*4]	Rate(*5)

4. The maximum aggregate principal amount of all Money Market Loans:

All numbered footnotes appear on the last page of this Exhibit.

We understand and agree that the offer(s) set forth above, subject to the satisfaction of the applicable conditions set forth in the Credit Agreement, irrevocably obligates us to make the Money Market Loan(s) for which any offer(s) (is/are) accepted, in whole or in part (subject to the third sentence of Section 2.03(e) of the Credit Agreement).

Very truly yours,

[Name of Leader]

Dv			
Aut	porized o	fficer	

Dated:		

- [1] As specified in the related Money Market Quote Request.
- [2] The principal amount bid for each Interest Period may not exceed the principal amount requested. Bids must be made for at least \$5,000,000 or a larger multiple of \$1,000,000.
- [3] Indicate "Margin" (in the case of LIBOR Market Loans) or "Rate" (in the case of Set Rate Loans).
- [4] One, two, three or six months, in the case of a LIBOR Market Loan or, in the case of a Set Rate Loan, a period of up to 180 days after the making of such Set Rate Loan and ending on a Business Day, as specified in the related Money Market Quote Request.
- [5] For a LIBOR Market Loan, specify margin over or under the London interbank offered rate determined for the applicable Interest Period. Specify percentage (rounded to the morest 1/10,000th of 1%) and specify whether "PLUS" or "MINUS"; For a Set Rate Loan, specify rate of interest per annum (rounded to the morest 1/10,000th of 1%).

### EXHIBIT (a)-2

Second Amended and Restated Credit Agreement B

	FLORIDA POWER CORPORATION
SECOND	AMENDED AND RESTATED CREDIT AGREEMENT I
	Dated as of November 18, 1997
, Cr	This Agreement amends and restates redit Agreement B Duted as of November 26, 1991
	THE CHASE MANHATTAN BANK as Agent
•••••	***************************************

### TABLE OF CONTENTS

This Table of Contents is not part of the Agreement to which it is attached but is inserted for convenience only.

		Pag
Section 1.	Definitions	and Accounting Matters
	1.01	Certain Defined Terms
	1.02	Accounting Terms and Determinations
	1.03	Classes and Types of Loans
Section 2.	Commission	<b> </b>
	2.01	Syndicated Loans
	2.02	Borrowings of Syndicated Loans
	2.03	Money Market Losss
	2.04	Changes of Commitments
	2.05	Facility Fee
	2.06	Leading Offices
	2.07	Several Obligations; Remedies Independent
	2.08	Notes
	2.09	Prepayments and Conversions on Continuations
		of Leass
Section 3.	Payments of	Principal and Interest
J.	3.01	Repayment of Loans
	3.02	Interest
	3.02	
Section 4.	Payments; I	Pro Rata Treatment; Computations; Etc
	4.01	Payments
	4.02	Pro Rata Treatment
	4.03	Computations
	4.04	Minimum Amounts
	4.05	Certain Notices
	4.06	Non-Receipt of Funds by the Agent
	4.07	Sharing of Payments, Etc
	4.07	Salay or royalas, Ex.
Section 5.	<b>Yield Prote</b>	tion and Illegality
	5.01	Additional Costs
	5.02	Limitation on Types of Loans
	5.03	Illegality
	5.04	Treatment of Affected Loans
	5.05	Compensation
Section 6.	Conditions	Precedent
Section 0.	6.01	Initial Loss
	6.02	Initial and Subsequent Loans
	0.02	

Section 7.	Representati	cas and Warranties	20
	7.01	Corporate Existence	
	7.02	Financial Condition	2
	7.03	Litigation	2
	7.04	No Breach	30
	7.05	Corporate Action	3
	7.06	Approvals	
	7.07	Use of Loans	
	7.08	ERISA	
	7.09	Texas	
	7.10	Investment Company Act	
	7.11	Public Utility Holding Company Act	
Section 8.	Covenants of	f the Company	31
	8.01	Financial Statements	
	8.02	Litigation	
	8.03	Corporate Existence, Etc	
	8.04	Prohibition of Fundamental Changes	
	8.05	Use of Proceeds	
	8.06	Indobtedness to Capitalization Ratio	
	8.07	Negative Pledge	35
Section 9.	Events of D	test	36
Section 10.	The Agent		38
	10.01	Appointment, Powers and Immunities	
	10.02	Relience by Agent	
	10.03	Defaults	39
	10.04	Rights as a Londor	
	10.05	Indomnification	40
	10.06	Non-Reliance on Agent and Other Leaders	40
	10.07	Failure to Act	40
	10.06	Resignation or Removal of Agent	40
	10.09	Agency Fee	41
Section 11.	Miscellanson	•	41
	11.01	Waiver	41
	11.02	Netices	41
	11.03	Expenses, Etc	42
	11.04	Amendments, Etc	
	11.05	Successors and Assigns	
	11.06	Assignments and Participations	
	11.07	Survival	44
	11.06	Castions	44

11.09	Counts	spects .													 45
11.10	Govern	ing Law	; Sub	missi	08 K	) Ju	ris	dic	tio	8 .					45
11.11	Waive	of Jury	Triel	٠.		٠.	٠.			٠.				, .	 45
EXHIBIT A-I		Form of	Note	for	Symd	ice	bed	ما	44						
<b>EXHIBIT A-2</b>		Form of	Note	for	Mon	ey	Ma	rke	t L	.00	8				
EXHIBIT B		Form of	Opia	ice c	f Co	Na.	el I	10	be	P		1			
EXHIBIT C		Form of	Mon	ey M	arke	Q	uot	e R	leq	ue	et				
EXHIBIT D		Form of	Mon	y M	arke	t Q	uot								

THIS SECOND AMENDED AND RESTATED CREDIT AGREEMENT B dated as of November 18, 1997 between: FLORIDA POWER CORPORATION, a corporation duly organized and validity existing under the laws of the State of Florida (the "Company"); each of the lenders that is a signatory hereto or which, pursuant to Section 11.06(b) hereof, shall become a "Lender" hereunder (individually, a "Lender" and, collectively, the "Lenders"); and THE CHASE MANHATTAN BANK, as agent for the Lenders (in such capacity, together with its successors in such capacity, the "Agent"); amends and restates the Credit Agreement B dated as of November 26, 1991, between the Company, each of the Lenders and the Agent.

The Company has requested that the Lenders make loans to it in an aggregate principal amount not exceeding \$200,000,000 at any one time outstanding and the Lenders are prepared to make such loans upon the terms hereof. Accordingly, the parties hereto agree as follows:

### Section 1. Definitions and Accounting Matters.

1.01 Certain Defined Terms. As used herein, the following terms shall have the following meanings (all terms defined in this Section 1.01 or in other provisions of this Agreement in the singular to have the same meanings when used in the plural and vice versa):

"Applicable Landing Office" shall mean, for each Lender and for each Type of Loan, the "Lending Office" of such Lender (or of an affiliate of such Lender) designated for such Type of Loan on the signature pages hereof or such other office of such Lender (or of an affiliate of such Lender) as such Lender may from time to time specify to the Agest and the Company as the office by which its Loans of such Type are to be made and maintained.

### "Applicable Margie" shall mose:

- (i) during each Class 1 Rating Period.
  - (A) with respect to Base Rate Loans, zero,
  - (B) with respect to Eurodollar Loans, 0.17%, and
  - (C) with respect to CD Loans, 0.295%,
- (ii) during each Class 2 Rating Period,
  - (A) with respect to Base Rate Loans, zero,
  - (B) with respect to Eurodollar Loans, 0.225%,
  - (C) with respect to CD Loans, 0.35%, and
- (iii) during each Class 3 Rating Period.

- (A) with respect to Base Rate Loans, zero,
- (B) with respect to Eurodollar Loans, 0.40%, and
- (C) with respect to CD Loans, 0.525%.

cach Class during each Applicable Facility For Base 1 Rating Period, 0.08%; (b) du h Class 3 Rating Period, 0.20%. g" shall mean, a rate per assum equal to (a) during uring each Class 2 Rating Period, 0.15%; and (c)

ayable by Ca To Dall D na, for any RET Ç laserest Period for any CD Loan, the secessary, to the meanest 1/100 of 1%) poration (or any success TIL Office successor) for deposit g such laterest Period,

such day plass 1/2 of 1% ly day, the higher of (a) the Fo iting from a cha teral Funds Rate for Each change in any

upon the I m' stall man Syndic ed Losse which bear interest at rates based

6 K Cky is Dollar . a payment or propayment of princip r Loss or a LIBOR Market Loss or ates to the giving of notices or quotes yment, prepayment or Interest are carried out in the London rcial banks are not authorized est of pris

Person to pay rest or to use) real and/or paccounted for as a cap 2 p Õ No. 13 of the hase of (or other a ce with GAA Person, the obligations of such GAAP (inclu

determined on the basis in this Section 1.01. referred to in clause (b) of the det effaition of "Fixed Base Rate" "Chane" shall mean The Chase Manhattan Bank.

"Class" shall have the meaning given to that term in Section 1.03 hereof.

"Class I Rating Pariod" shall mean any period during which the rating of the First Mortgage Bonds (a) by Moody's equals or exceeds "A3" and (b) by S&P equals or exceeds "A-".

"Class 2 Rating Period" shall mean any period during which the rating of the First Mortgage Bonds (a) by Moody's equals or exceeds "Baa3" and (b) by S&P equals or exceeds "BBB-", and which is not a Class 1 Rating Period.

"Class 3 Rating Pariod" shall mean any period that is neither a Class 1 Rating Period nor a Class 2 Rating Period.

"Code" shall mean the Internal Revenue Code of 1986, as amended from time to time.

"Commitment" shall mean, with respect to each Lender, the obligation of such Lender to make Syndicated Loans pursuant to Section 2.01 hereof in an aggregate amount at any one time outstanding up to but not exceeding the amount set opposite such Lender's name on the signature pages hereof under the caption "Commitment" (as the same may be reduced at any time or from time to time pursuant to Section 2.04 hereof). The original aggregate amount of the Commitments is \$200,000,000.

"Commitment Termination Date" shall mean, with respect to each Lender, November 30, 2002; provided that

- (a) if, (i) not later than 39 months, but no more than 40 months, prior to the Commitment Termination Date, determined after giving effect to all previous extensions thereof pursuant to this definition (the "Existing Commitment Termination Date"), the Company requests that the Landers agree to extend the Commitment Termination Date to the December 31 falling two years after the Existing Commitment Termination Date (the "Proposed Commitment Termination Date") and (ii) each of the Lenders so agrees in writing prior to the Existing Commitment Termination Date, then the "Commitment Termination Date" shall be extended, with respect to each Lender, to the Proposed Commitment Termination Date;
- (b) if, pursuant to any such request, some, but not all, of the Lenders agree to so extend the Existing Commitment Termination Date to the Proposed Commitment Termination Date (the Lenders that so agree, the "Commitment Lenders"), the "Commitment Termination Date" shall mean (i) with respect to the Consenting Lenders, the Proposed Commitment Termination Date and (ii) with respect to the Lenders that are not Consenting Lenders, the Existing Commitment Termination Date; and

(c) if the Commitment Termination Date as determined above is not a business Day, the Termination Date shall be the next preceding Business Day.

hall be (or shoul whether now existing or here (or should have been) com-ne with GAAP. <u>lititud Subsidiary</u> shall mean, as to any Person, each Subsidiary of such existing or hereafter created or acquired) the financial statements of which new been) consolidated with the financial statements of such Person in

"Continue". "Continuetos pursuant to Section 2.09 hereof of a Fixed same Type from one lanerest Period to the " and "Continued" shall refer to the continuation if Rate Loan of one Type as a Fixed Rate Loan of the sext laterest Period.

into Base Rate Loans or Eurodoller Loans, or of Eurodoller Loans into Base Rate Loans or C. Loans, which may be accompanied by the transfer by a Loader (at its sole discretion) of a Loans one Applicable Loaning Office to another. to Section 2.09 harmof of Bus B F. F. Loss isso CD Los mad' shall refer to a conversion pu as or Eurodollar Loans, of CD Loans LORS OF CD

"Default" shall mean an Event of Default or an event which with notice or lapse of time or both would become an Event of Default.

"Dallan" and "S" shall mean lewful money of the United States of America.

mended from time to time. I mean the Employee Ratirement Income Security Act of 1974, as

member of the same country the Code) as the Company of the Code) with the Com "ERISA Affiliate" shall mean any corporation or trade or business which is a same controlled group of corporations (within the meaning of Section 414(b) of the Company or is under common control (within the meaning of Section 414(c))

"Eurodollar Long" shall mean Syndicated Longs the interest rates on which are sed on the basis of rates referred to in clause (a) of the definition of "Fixed Base Rate" ection 1.01.

"Event of Default" shall have the meaning assigned to such term in Section 9

the Company "Existing Industries" shall mean the Indenture dated as of January 1, 1944 between and First Chicago Trust Company of New York, successor Trustee, as amended used and in effect from time to time.

"Endered Funds Bate" shall mean, for any day, the rate per amoun (rounded upwards, if necessary, to the nearest 1/100 of 1%) equal to the weighted average of the rates

Company's First Mongage Boads issued

with respect to any Fixed Rate Loan for any

"Final Rate" shall mean, for any Fixed Rate Loan for any Interest Period therefor, a rate per annum (rounded upwards, if necessary, to the nearest 1/100 of 1%) determined by the Agent to be equal to the sum of (a) the Fixed Base Rate for such Interest Period divided by 1 misses the Reserve Requirement for such Loan for such Interest Period plus (b) if such Loan is a CD Loan, the Assessment Rate for such Interest Period.

"Fixed Rest Loans" shall mean CD Loans, Eurodollar Loans and, for the purposes of the definition of "Fixed Base Rate" in this Section 1.01 and in Section 5 hereof, LIBOR Market Loans.

"GAAP" shall mean generally accepted accounting principles as in effect from time to time.

"Guarantee" shall mean any guarantee, conformment, contingent agreement to purchase or to furnish funds for the payment or maintenance of, or otherwise to be or become contingently liable under or with respect to, Indobtedness, other obligations, net worth, working capital or carnings of any Person, or a guarantee of the payment of dividends or other distributions upon the stock of any Person, or any agreement to purchase, sell or lease (as leases or lessor) property, products, materials, supplies or services primarily for the purpose of enabling any Person to make payment of its obligations or any agreement to assure a creditor against loss, and including without limitation, causing a bank to open a letter of credit for the benefit of another Person, but excluding endorsements for collection or deposit in the ordinary course of business. The terms "Guarantee" and "Guaranteed" used as a verb shall have a correlative meaning.

"Indebtedness" shall mean, as to any Person: (a) indebtedness created, issued or incurred by such Person for borrowed money (whether by loan or the issuance and sale of debt securities); (b) obligations of such Person to pay the deferred purchase or acquisition price of property, other than trade accounts psyable (other than for borrowed money) arising, and accrued expenses incurred, in the ordinary course of business so long as such trade accounts psyable are paid within 90 days of the date the respective goods are delivered; (c) obligations of such Person in respect of letters of credit or similar instruments issued or accepted by banks and other financial institutions for the account of such Person; (d) Capital Lease Obligations of such Person; and (e) Indebtedness of others Guaranteed by such Person.

### "Interest Period" shall moon:

(a) with respect to any Eurodeller Loan, each period commencing on the date such Eurodeller Loan is made or Converted from a Loan of another Type or the last day of the next preceding laterest Period for such Loan and ending on the numerically corresponding day in the first, second, third or sixth calender month thereafter, as the Company may select as provided in Section 4.05 hereof, except that each Interest Period which commences on the last Business Day of a calender month (or on any day for which

there is no numerically corresponding day in the appropriate subsequent calendar month) shall end on the last Business Day of the appropriate subsequent calendar month;

- (b) with respect to any CD Loss, each period commencing on the date such CD Loss is made or Converted from a Loss of another Type or the last day of the next preceding Interest Period for such Loss and ending on the day 30, 60, 90 or 180 days thereafter, as the Company may select as provided in Section 4.05 hereof;
- (c) With respect to any Set Rate Loan, the period commencing on the date such Set Rate Loan is made and ending on any Business Day up to 180 days thereafter, as the Company may select as provided in Section 2.03(b) hereof; and
- (d) With respect to any LIBOR Market Loan, the period commencing on the date such LIBOR Market Loan is made and ending on the mamerically corresponding day in the first, second, third or sixth calendar month thereafter, as the Company may select as provided in Section 2.03(b) hereof, except that each laterest Period which commences on the last Business Day of a calendar month (or any day for which there is no numerically corresponding day in the appropriate subsequent calendar month) shall end on the last Business Day of the appropriate subsequent calendar month.

Notwithstanding the foregoing: (i) so Interest Period with respect to Loans to be made by any Leader may end after such Loader's Commitment Termination Date (as in effect on the first day of such Interest Period); (ii) each Interest Period which would otherwise end on a day which is not A Business Day shall end on the next succeeding Business Day (or, in the case of an Interest Period for Eurodolfar Loans or LIBOR Market Loans, if such next succeeding Business Day falls in the next succeeding calendar mouth, on the next preceding Business Day); and (iii) notwithstanding clause (i) above, so Interest Period for any Fixed Rate Loans or LIBOR Market Loans shall have a duration of less than one mouth (in the case of Eurodolfar Loans and LIBOR Market Loans) or 30 days (in the case of CD Loans) and, if the Interest Period for any Fixed Rate Loans or LIBOR Market Loans would otherwise be a shorter period, such Loans shall not be available horounder.

"LIBO Rate" shall meen, for any LIBOR Market Loan, a rate per annum (rounded upwards, if necessary, to the mearest 1/100 of 1%) determined by the Agent to be equal to the rate of interest specified in clause (a) of the definition of "Fixed Base Rate" in this Section 1.01 for the Interest Period for such Loan divided by 1 minus the Reserve Requirement for such Loan for such Interest Period.

"LIBOR Auction" shall mean a solicitation of Money Market Quotes setting forth Money Market Margins based on the LIBO Rate pursuant to Section 2.03 hereof.

"LIBOR Market Loans" shall mean Money Market Loans the interest rates on which are determined on the basis of LIBO Rates pursuant to a LIBOR Auction.

"Lim" shall mean, with respect to any asset, any mortgage, lien, pledge, hypothecation, charge, security interest or encumbrance of any kind in respect of such asset. For purposes of this Agreement, the Parent or any of its Subsidiaries shall be deemed to own subject to a Lien any asset which it has acquired or holds subject to the interest of a vendor or lessor under any conditional sale agreement, capital lease or other title retention agreement relating to such asset.

"Longs" shall mean Money Market Loans and Syndicated Loans.

"Majority Landers" shall mean Lenders having at least 66-2/3% of the aggregate amount of the Commitments; provided that, if all of the Commitments shall have terminated, Majority Lenders shall mean Lenders holding at least 66-2/3% of the aggregate unpaid principal amount of the Long.

"Margin Stock" shall mean margin stock within the meaning of Regulations U and X.

"Money Market Barrowing" shall have the meaning assigned to such term in Section 2.03(b) hereof.

"Money Market Longs" shall mean the loans provided for by Section 2.03 hereof.

"Money Market Margin" shall have the meaning assigned to such term in Section 2.03(c)(ii)(C) hereof.

"Money Market Chote" shall mean an offer in accordance with Section 2.03(c) hereof by a Lender to make a Money Market Loan with one single specified interest rate.

"Money Market Ounce Request" shall have the meaning assigned to such term in Section 2.03(b) hereof.

"Money Machet Rese" shall have the meaning assigned to such term in Section 2.03(c)(ii)(D) hereof.

"Mondy's" shall mean Mondy's Investors Services, Inc.

"Multiemployer Plea" shall mean a Multiemployer plan defined as such in Section 3(37) of ERISA to which contributions have been made by the Company, the Parent or any ERISA Affiliate and which is covered by Title IV of ERISA.

"1935 Act" shall have the meaning given to that term in Section 7.11 hereof.

"1934 Act Reports" shall mean all periodic reports filed with the Securities and Exchange Commission pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934, as amended.

"Notes" shall mean the promissory notes provided for by Section 2.06 hereof.

"Other FPC Agraement" shall mean the Second Amended and Restated Credit
Agreement A dated as of November 18, 1997 between the Company, the Lenders and Chase,
as agent for the Lenders thereunder, as the same may be amended and supplemented and in
effect from time to time.

"Pacest" shall mean Floride Progress Corporation, a Floride corporation.

"PRGC" shall mean the Pension Benefit Guaranty Corporation or any entity succeeding to any or all of its functions under ERISA.

"PCH Agreements" shall mean, collectively, (a) the Second Amended and Restated Credit Agreement A deted as of November 18, 1997 between Progress Capital Holdings, Inc., the Lenders and Chase, as agent for the Lenders thereunder and (b) the Second Amended and Restated Credit Agreement B dated as of November 18, 1997 between Progress Capital Holdings, Inc., the Lenders and Chase, as agent for the Lenders thereunder, as each of said agreements may be amended and supplemented and in offect from time to time.

"Person" shall mean any individual, corporation, company, voluntary association, partnership, joint venture, trust, unincorporated organization or government (or any agency, instrumentality or political subdivision thereof).

"Plan" shall mean an employee benefit or other plan established or maintained by the Company, the Parent or any ERISA Affiliate and which is covered by Title IV of ERISA, other than (a) a Multiemployer Plan and (b) any such plan established or maintained by the Company or any ERISA Affiliate that has assets and actuarial liabilities of less than \$50,000,000 (a "Small Plan") unless the aggregate assets or aggregate actuarial liabilities of all Small Plans is in excess of \$50,000,000.

"<u>Post-Default Rese</u>" shall mean, in respect of any principal of any Loan or any other amount psyable by the Company under this Agreement or any Note that is not paid when due (whether at stated maturity, by acceleration or otherwise), a rate per annum during the period from and including the due date to but excluding the date on which such amount is paid in full equal to 1% above the Base Rate as in effect from time to time (provided that, if the amount so in default is principal of a Fixed Rate Loan or a Money Market Loan and the due date thereof is a day other than the last day of an Interest Period therefor, the "Post-Default Rate" for such principal shall be, for the period from and including such due date to but excluding the last day of such Interest Period, 1% above the interest rate for such Loan as provided in Section 3.02 hereof and, thereafter, 'the rate provided for above in this definition).

Principal Office as its prime commercial mercial leading noed by

"Bringing Office" shall mean the principal office of the ed at 270 Part Avenue, New York, New York 10017. Ş and Chase

the first of which a shall mean the first day of January. April, July and October hall be the first such day after the date of this Agreement; x a Duniness Day, then such Quarterly Date shall be the next

"Reference Leading" shall mean Chase and Morgan Guaranty Trust Company of New York (or their Applicable Leading offices, as the case may be).

Regulation D. Regul System (or any succ cies U and Regulation X of the B 100 and of Gover oversors of the Fed and from time to time. . respectively.

once of law) by say count or ter any United Sta 90 9 to a class of bas lew or m ect to any Leader, any chan orthy cha (whether or not having the ch date of any Leader of or

York City with or LIBOR Mar (b) in the case 20 00 other reserve more. Withou of CD La Loan, the average maximum rate at or emergency reserves) are required agulation D by manufer banks of the E M. LIBO 8 as or CD Loans (as the case may be) riod for any Fixed serves (including a sest shall include any in Regulation D) or ount of \$100,000 or

"S&P" shall mean Standard & Poor's Corporation

"Set Rate Auction" shall mean a solicitation of Money Market Quotes setting forth Money Market Rates pursuant to Section 2.03 hereof.

"Set Rate Loans" shall mean Money Market Loans the interest rates on which are determined on the basis of Money Market Rates pursuant to a Set Rate Auction.

"Subsidiery" shall mean, as to any Person, any corporation of which at least a majority of the outstanding shares of stock having by the terms thereof ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether or not at the time stock of any other class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is at the time directly or indirectly owned or controlled by such Person or one or more Subsidiaries of such Person or by such Person and one or more Subsidiaries of such Person. "Wholly-owned Subsidiary" shall mean any such corporation of which all of such shares, other than directors' qualifying shares, are so owned or controlled.

"Syndicated Losse" shall mean the losses provided for by Section 2.01 hereof.

"Syndicated Notes" shall mean the promissory notes provided for by Section 2.08(a) hereof.

"Total Conitalization" shall mean, with respect to any Person, the sum of the value of the common stock, retained earnings, and preferred and preference stock of such Person (in each case, determined in accordance with GAAP) plus all Indebtedness of such Person.

"Type" shall have the messing given to that term in Section 1.03 hereof.

### 1.02 Accounting Terms and Determinations.

- (a) Except as otherwise expressly provided herein, all accounting terms used herein shall be interpreted, and all financial statements and certificates and reports as to financial matters required to be delivered to the Londors hereunder shall (unless otherwise disclosed to the Lenders in writing) be prepared, in accordance with generally accepted accounting principles applied on a basis consistent with that used in the preparation of the latest financial statements furnished to the Londors hereunder after the date hereof.
- (b) The Company will not change the last day of its fiscal year from December 31 of each year, or the last days of the first three fiscal quarters in each of its fiscal years from March 31, June 30 and September 30 of each year, respectively.
- 1.03 Classes and Types of Loss. Loss hereunder are distinguished by "Class" and by "Type". The "Class" of a Loss (or of a Commitment to make a Loss) refers to whether such Loss is a Money Market Loss or a Syndicated Loss, each of which constitutes a Class. The "Type" of a Loss refers to whether such Loss is a Base Rate Loss, a CD Loss, a

Eurodollar Loan, a Set Rate Loan, or a LIBOR Market Loan, each of which constitutes a Type Loans may be identified by both Class and Type.

## Section 2. Commitment

- siness Day of 1992 to but not S Co Comp 2.01 00 5 th C 2 mow, repay and reborrow the as of another Type (as provided in S Type as Syndicated Loans of the ser rest laterest Periods for Syndicated I in effect. Subject to the terms of this Assessment g. Each Lender severally agrees, on the terms of this pany in Dollars during the period from and including the including such Lender's Commitment Termination Date ad reborrow the amoun he serms of this Agreement, during the amount of the Commitments by i may Convert Syndicated Loans ided in Section 2.09 hereof) or of the same Type; provided that diag at the
- (which shall promptly noti Loans as provided in Sect specified for each borrowin ified for each borrowing of Symble mount of the Symblessed Lenn to ber NYAO-DI-900-9-000002 main in New York City. ety evaluate frants. The amount so receives of this Agreement, be made evaluate rely evaluate frants, in an eccount of, and caion 4.05 hereof. increased Louise bertweeder. The best and the Agent. The best by the Agent with Chase at the Principal Office.

  The Agent with Chase at the Principal Office.

  The Agent with Chase at the Principal Office.

  The Agent shall, subject to the terms

  The anale svalights to the Company by depositing the same.

  The Agent with Chase at the Principal Office. active of each The Company shall give the Agent borrowing herwarder of Syndicated a acces New York time on the date ler, each Lender shall make available

## 2.03 Monty Marias Long.

- nay be LIBOR Ma shall have so o In addition to borrowings of Syndicated Loans, the Company may, on or is Day of 1992, as set forth in this Section 2.03, request the Lenders to Money Market Loans to the Company in Dollars. The Lenders may, but e such offers and the Company may, but a manner set forth in this Section 2.03. hall have no obligation Money Market Loans
- which purpose laterest Per leftation of the term "later feriods even if they are con may be no more than 15 different interest Periods for both it Money Market Louis constanding at the same time (for set Periods described in different interest clauses of the "lawrest Period" shall be deemed to be different interest

- principal amount of all Money Market Loans, cipal amount of all Syndicated Loans, at any one igate amount of the Com
- Money Market Q Auction or (y) th My are comme Day prior of the Major 9 Company 2 Ĩ inhes to request offers to make Money promptly notify the Lenders) notice (a no later than 11:00 a.m. New York time for each separate for a separate borrowing (a "Money the form of Exhibit C hereto ent latered clauses of the est laterest Periods even if to laterest Period shall be date as the Company a " in the case of a LIBOR York time on (x) Martes
- P ed date of such borrowing. which shall be a
- the aggregate amount of such Money Market Borrowing, which least \$10,000,000 (or in larger multiples of \$1,000,000) but shall not insite specified in Section 2.03(a) hereof to be violated;
- iii) the duration of the Interest Period applicable thereto;
- (iv) whether the Money Market Quotes requested for a particular t Period are seeking quotes for LIBOR Market Longs or Set Rate Longs;
- 3 if the Mossy Market Quotes requested are seeking quotes for Set the date on which the Mossy Market Quotes are to be submitted if the proposed date of borrowing (the date on which such Mossy see are to be submitted is called the "Quotation Date").

Except as otherwise provided in this Section 2.03(b), no Money Martest Quote Request shall be given within five Business Days (or such other number of days as the Company and the Agent, with the consent of the Majority Landers (and with notice to each Lander prior to the effectiveness of such consent), may agree) of any other Money Martest Quote Request.

- ry Market Quotes 5.02(b), 5.03. .... D
- Estable D 18 Martes Quote shall be sa stially in the form of
- 3 date of borrowing and Interest Period therefor:
- for which each \$5,000,000 or ê H but (y) (X) 2 ē
- 1/10,000 **E** 3 0.5 of a Linox Micable LIBO below the
- tion, the rate of interest per annum: 1/10,000th of 1%) offered for each
- (E) the identity of the quoting Lander; and

(F) the maximum aggregate principal amount of all Money Market Longs for which such offer is being made.

Unless otherwise agreed by the Agent and the Company, no Money Market Quote shall contain qualifying, conditional or similar language or propose terms other than or in addition to those set forth in the applicable Money Market Quote Request and, in particular, no Money Market Quote may be conditioned upon acceptance by the Company of all (or some specified minimum) of the principal amount of the Money Market Loan for which such Money Market Quote is being made.

- (d) The Agent shall (x) in the case of a Set Rate Auction, as promptly as practicable after the Money Market Quote is submitted (but in any event not later than 10:15 a.m. New York time on the Quotation Date) or (y) in the case of a LIBOR Auction, by 4:00 p.m. New York time on the day a Money Market Quote is submitted, notify the Company of the terms (i) of any Money Market Quote submitted by a Lender that is in accordance with Section 2.03(c) lessed and (ii) of any Money Market Quote submitted by such Lender with respect to the same Money Market Quote Request. Any such subsequent Money Market Quote shall be disregarded by the Agent unless such subsequent Money Market Quote is submitted solely to correct a manifest error in such former Money Market Quote. The Agent's notice to the Company shall specify (A) the aggregate principal amount of the Money Market Borrowing for which offers have been received and (B) the respective principal amounts and Money Market Margins or Money Market Rates, as the case may be, so offered by each Lender (identifying the Lender that made each Money Market Quote).
- (e) Not later than 11:00 a.m. New York time on (x) the third Business Day prior to the proposed date of borrowing, in the case of a LIBOR Auction or (y) the Quotation Date, in the case of a Set Rate Auction (or, in any such case, such other time and date as the Company and the Agent, with the consent of the Majority Lenders (and with notice to each Lender prior to the Money Market Quote Request for which such change is to be effective), may agree), the Company shall notify the Agent of its acceptance or nonacceptance of the offers so notified to it pursuant to Section 2.03(d) hereof (and the failure of the Company to give such notice by such time shall constitute nonacceptance) and the Agent shall promptly notify each affected Lender. In the case of acceptance, such notice shall specify the aggregate principal amount of offers for each Interest Period that are accepted. The Company may accept any Money Market Quote in whole or in part (provided that any Money Market Quote accepted in part shall be at least \$5,000,000 or in larger multiples of \$1,000,000); provided that:
  - the aggregate principal amount of each Money Market Borrowing may not exceed the applicable amount set forth in the related Money Market Quote Request;
  - (ii) the aggregate principal amount of each Money Market Borrowing shall be at least \$10,000,000 (or in larger multiples of \$1,000,000) but shall not cause the limits specified in Section 2.03(a) hereof to be violated; and

- the Company may not accept any offer where the Agent has advised the lat such offer falls to comply with Section 2.03(c)(ii) harvof or otherwise falls vith the requirements of this Agreement (including, without limitation, Section
- Loan, make the at 9-000002 maintain ROF York City ( ) ( ) ( ) 0 ety available frank. ē scipal Office in immediately available ived by the Agent shall, subject to the N ST SCOOLST SE 8 ç eter Loan has been to making of such ther NYAO-DI-900listely available
- of my ma Except for the purpose and to the extent expressly stand in Section 2.04(b) Martex Loan made by any Londor shall not constitute a

## 2.04 Canad Commission

- (a) The set to zero on such Lender's Co noust of each Leader's Com D sest shall be automatically reduced
- purpose use of the Commit termination or redu long as no Symd ments and (ii) to reduce the age 3 Louis): pa 5 rovided that (x) i un ai bas 000,000,000 on Warter Loans are outstanding, to terminate the 8 te the aggregate principal amount of may shall give notice of each such f, and (y) each partial reduction shall in multiples of \$1,000,000 in excess C cipal amount of of each such eas (for which
- 3 THE COMM MAN CACH STREET minated or reduced may not be reinstated.
- 9 ē The Company shall pay to the Agent for account of each it of such Leader's Commitment, for the period from and set including the earlier of the date such Commitment is fees payable to any Lende stion Date, at a rate per an e shall be payable on terminated and such man equal to the
- 28 Landing Offices. The Longs of each Type made by each Lender shall be at such Lander's Applicable Lending Office for Longs of such Type.

### 2.08 Name

- 3 by each
- mas made by any sory mote of the

# Section 3. Promests of Principal and Interest.

3.01 Repayment of Long

- Leader the principal made by such Leads The Company hereby promises to pay to the Agent for account of each of each Syndicated Loan made by such Leader, and each Syndicated Loan for shall matter, on such Leader's Commitment Termination Date.
- (b) The Company harrby pro-that makes any Money Market Loan the last day of the Imment Period for such M geny hareby promises to pay to the Agest for account of each Market Loss the principal amount of such Money Market Loss Period for such Money Market Loss.
- ind from and including 2 The Company hereby promises to pay to the Agent for account unpaid principal amount of each Lonn made by such Lender for the date of such Lonn to but excluding the date such Lonn shall
- trum time to time) plus the Applicable Margin (if any); Base Rate (as
- faring such periods as such Loan is a Fixed Rate Loan, for each interest thereto, the Fixed Rate for such Loan for such interest Period plus the
- (c) if such Loss is a LIBOR Market Loss, the LIBO Rate for laserest Period therefor plus (or misses) the Mossy Market Margin for making such Loss is accordance with Section 2.03 hereof; and quoted by the
- laterest Period then Section 2.03 hereof. if such Loss is a Set Rate Loss.
  I therefor quoted by the Lender of n, the Set Rate for such Loan for the making such Loan in accordance with

by such Leader, and on my wheld by such Leader to or for and (iii) in the case of any Loan, upon the payment or propayment the of such Loan to a Loan of another Type (but only on the principal and Converted), except that imment payable at the Post-Default Rase shall time on demand. Promptly after the desermination of any inservet rates. Company hereby promises to pay to the Agent for account incide Post-Defhult Rate on any principal of any Loan made moust payable by the Company hereunder or under the Notes and of such Leader, which shall not be paid in full when due thereties or otherwise), for the period from and including the the date the same is paid in full. Accrued interest on each following the first day of such Interest Period. rvice), for the period from an er is paid in full. Accrued in a., quarterly on the Quarterly Dases, (ii)
Loss, on the last day of each interest
as 90 days (in the case of a CD Loss or
dollar Loss or a LIBOR Market Loss). ipal amount so paid, prep so shall be payable from ti rest rate provided for her applied or

De Comme give actice thereof to the Landers to which such

# p

## 4.01 Parameter

- Dolla al Office, as 3 NYAO-D 2000 New York char Day. nest and the Notes shall to on which such
- (b) shall not be obligate than the Comp y ordinary deposit assessment of the Company) with sect der (with Š account any such payment is to be made, may (but I any such payment which is not made by such time any (for purposes of this Section 4.01(b), "ordinary of a Person
- Agreement or any None, specify to the recipiests) thereof) the Losse or other assect payment is to be applied (and in the Default has occurred and is constanting, a received by it from the Agent in such many 0 The Con to Leader may apply the the Company hereunder to which to so specify, or if an Event of nime to be appropriate). payment under this notify the intended berreader to which at of such payme
- Each payment received by the Agent under this Agreement or any Note or shall be paid promptly to such Londer, in immediately available funds, ender's Applicable Londing office for the Long in respect of which such
- (e) If the due date of any payment under this Agreement or any Note would otherwise full on a day which is not a Deviness Day such date shall be extended to the next succeeding Decimes Day and interest shall be payable for any principal so extended for the period of such extended.
- 8 (e) to mide To maker Section 2.01 horses of the amount of the Comection 2.01 harms hand shall be made made for account of the Lenders.

  made for account of the Lenders.

  made for account of the Lenders.

  according to the amounts of their
  according to Syndicated Loans of reia: (a)

- **y**) 000 8 Loans shall be computed on the basis of a year first day but excluding the last day) occurring is of a year of 365 ny but exch
- to \$10,000,000 as • 3
- 8 0 on 2.03 he

Termination or reduction of the Commission

3

or Convenient into

Me day

Committee of Landson for the Committee of Landson for Landson for Landson for Landson Landson

I

Committee in Commi

3

available to the intended recipient(s) on such date and, if the Payor has not in fact made the Required Payment to the Agent, the recipient(s) of such payment shall, on demand, repay to the Agent the amount so made available together with interest thereon in respect of each day during the period commencing on the date such amount was so made available by the Agent until the date the Agent recovers such amount at a rate per annum equal to the Federal Funds Rate for such day and, if such recipient(s) shall fail promptly to make such payment, the Agent shall be entitled to recover such amount, on demand, from the Payor, together with interest as aforesaid.

## 4.07 Shering of Payments, Etc.

- (a) The Company agrees that, in addition to (and without limitation of) any right of set-off, banker's lien or counterclaim a Lender may otherwise have, each Lender shall be entitled, at its option, to offset balances held by it for account of the Company (for purposes of this Section 4.07(a), "balances held for account of the Company" shall not include any balances held in an account in the name of a Person other than the Company) at any of its offices, in Dollars or in any other curvacy, against any principal of or interest on any of such Lender's Lonns, or any other amount payable to such Lender herounder, which is not paid when due (regardless of whether such balances are then due to the Company), in which case it shall promptly notify the Company and the Agent thereof, provided that such Lender's failure to give such notice shall not affect the validity of such offset.
- Loan made by it to the Company under this Agreement through the enercise of any right of setoff, banker's lieu or counterclaim or similar right or otherwise, and, as a result of such
  payment, such Lender shall have received a greater percentage of the principal or interest then
  due hereunder by the Company to such Lender than the percentage received by any other
  Lenders, it shall promptly purchase from such other Lenders participations in (or, if and to the
  extent specified by such Lender, direct interests in) the Loans owing to such other Lenders (or
  in interest due thereon, as the case may be) in such amounts, and make such other adjustments
  from time to time as shall be equitable, to the end that all the Lenders shall share the benefit of
  such excess payment (not of any expenses which may be incurred by such Lender in obtaining
  or preserving such excess payment) pro ruts in accordance with the uspaid principal of and/or
  interest on the Loans owing to each of the Lenders. To such end all the Lenders shall make
  appropriate adjustments among themselves (by the resale of participations sold or otherwise) if
  such payment is receinded or must otherwise be restored.
- (c) The Company agrees that any Lender so purchasing a participation (or direct interest) in the Loans made by other Lenders (or in interest due thereon, as the case may be) may exercise all rights of set-off, banker's lien, counterclaim or similar rights with respect to such participation as fully as if such Lender were a direct holder of Loans in the amount of such participation.

- (d) Nothing contained herein shall require any Lender to exercise any such right or shall affect the right of any Lender to exercise, and retain the benefits of exercising, any such right with respect to any other indebtedness or obligation of the Company.
- (e) If, under any applicable bankruptcy, insolvency or other similar law, any Lender receives a secured claim in lieu of a set-off to which this Section 4.07 applies, such Lender shall, to the extent practicable, exercise its rights in respect of such secured claim in a manner consistent with the rights of the Lenders entitled under this Section 4.07 to share in the benefits of any recovery on such secured claim.

## Section 5. Yield Protection and Illegality.

### 5.01 Additional Costs.

- (a) The Company shall pay directly to each Londer from time to time such amounts as such Londer may determine to be necessary to compensate it for any costs which such Londer determines are attributable to its making or maintaining of any Fixed Rate Lones or its obligation to make any Fixed Rate Lones horounder, or any reduction in any amount receivable by such Londer horounder in respect of any of such Lones or such obligation (such increases in costs and reductions in amounts receivable being Lerein called "Additional Costs"), resulting from any Regulatory Change which:
  - (i) subjects any Lender to taxation on, or changes the basis of taxation of, any amounts payable to such Lender under this Agreement or its Notes in respect of any of such Loans (other than taxes imposed on or measured by the overall not income of such Lender or of its Applicable Lending Office for any of such Loans by the jurisdiction in which such Lender has its principal office or such Applicable Lending Office); or
  - (ii) imposes or modifies any reserve, special deposit or similar requirements (other than the Reserve Requirement utilized in the determination of the Fixed Rate or LIBO Rate, as the case may be, for such Loan) relating to any extensions of credit or other assets of, or any deposits with or other liabilities of, such Loader (including any of such Loans or any deposits referred to in the definition of "Fixed Base Rate" in Section 1.01 hereof), or any commitment of such Loader (including the Commitments of such Lorder hereunder); or
  - (iii) imposes any other condition affecting this Agreement or its Notes (or any of such extensions of credit or liabilities) or its Commitments.

If any Lender requests compensation from the Company under this Section 5.01(a), the Company may, by notice to such Lender (with a copy to the Agent), suspend the obligation of such Lender to make or Continue Loans of the Type with respect to which such compensation

is requested-until the Regulatory Change giving rise to such request ceases to be in effect (in which case the provisions of Section 5.04 hereof shall be applicable).

- (b) Without limiting the effect of the provisions of paragraph (a) of this Section 5.01, in the event that, by reason of any Regulatory Change, any Lender either (i) incurs Additional Costs based on or measured by the excess above a specified level of the amount of a category of deposits or other liabilities of such Lender which includes deposits by reference to which the interest rate on Eurodollar Loans or CD Loans is determined as provided in this Agreement or a category of extensions of credit or other assets of such Lender which includes Eurodollar Loans or CD Loans or (ii) becomes subject to restrictions on the amount of such a category of liabilities or assets which it may hold, then, if such Lender so elects by notice to the Company (with a copy .to the Agent), the obligation of such Lender to make or Continue Loans of such Type herounder shall be suspended until such Regulatory Change ceases to be in effect (in which case the provisions of Section 5.04 hereof shall be applicable).
- (c) Without limiting the effect of the foregoing provisions of this Section 5.01 (but without duplication), the Company shall pay directly to each Lender from time to time on request such amounts as such Lender may determine to be necessary to compensate such Lender (or, without duplication, the bank holding company of which such Lender is a subsidiary) for any costs which it determines are attributable to the maintenance by such Lender (or any Applicable Lending Office or such bank holding company), pursuant to any law or regulation or any interpretation, directive or request (whether or not having the force of law) of any court or governmental or mountary authority
  - (i) following any Regulatory Change, or
  - (ii) implementing any risk-based capital guideline or requirement (whether or not having the force of law and whether or not the failure to comply therewith would be unlawful) heretofore or hereafter issued by any government or governmental or supervisory authority implementing at the national level the Basile Accord (including, without limitation, the Final Risk-Based Capital Guidelines of the Board of Governors of the Federal Reserve System (12 CFR Part 208 'Appendix A; 12 CFR Part 225, Appendix A) and the Final Risk-Based Capital Guidelines of the Office of the Comptroller of the Currency (12 CFR Part 3, Appendix A)),

of capital in respect of its Commitment or Loans (such Compensation to include, without limitation, an amount equal to any reduction of the rate of return on assets or equity of such Lender (or any Applicable Lending Office or such bank holding company) to a level below that which such Lender (or any Applicable Lending office or such bank holding company) could have achieved with respect to such Lender's Commitment or Loans herounder but for such law, regulation, interpretation, directive or request). For purposes of this Section 5.01(c), "Basic Accord" shall mean the proposals for risk-based capital framework described by the Basic Committee on Banking Regulations and Supervisory Practices in its paper entitled "International

of Capital M rement and Capital Standards" dated July 1968, as an in effect from time to time or any replacement thereof.

- Agreement 5.01 3 ler will notify the Co of cre my of any event occ I May Le 3.01 for co 5.01, shall be 9 ing after the date spin (a) or (c) of ured from and provided. CHAIVE, 9
- 3.8 2 2 NO. to the determination of any Fixed Base Rate for any Interest
- on of 'Fixed Be tion shall be conclusive usts or for the relevant for any Type of Fixed se Rate" in Section
- etermination ad notify (or Market De

e may be) for such laserest Period is to be determined are not to cover the cost to such Lenders (or to such quoting Lender) staining such Type of Longs for such laserest Period; her Louis or CD Louis (or LIBOR Market

Loss of any other 8 all give the Company ted for the or Louis of such Leaders (or such quotie and each Less such Type, her Type of Loss is accor for prompt notice thereof, and so long as such quoting Lensier) shall be under no to Continue Louis of such Type or to ting Louis of such Type, either prepay Loans of such 1379-

- 5.8 200 Ī er or its Applicab of (with g any other provision of this Agreeme or its Applicable Lending Office to a copy to the Ages e the provisions of Section 5.04 hereof obligated to make any LIBOR Market -Ē
- ender may specify to the Con gives notice as provided below which gave rise to such Conv he case of a Comw Š 2 F - A 1 . A. so longer exist: a copy to the Agent) and, unless and until such Lender a 5.01(b) or 5.03 hereof, on such earlier date as such F If the obligation of any Lender to make a led pursuant to Section 5.01 or 5.03 hereof Lenne" and such Type being herein called shall be automatically Converted into Base specified in Section 5.01 or 5.03 hereof riod(s) for Affected Loans (or, is
- all payments Lender's Affi 3 the estimat that such Lander's Affected Loans have been so Converted, propayments of principal which would otherwise be applied to such Loans shall be applied instead to its Base Rate Loans;
- 3 A TO COMP 2 would otherwise be made or Continued by such Lender I be made or Continued instead as Base Rate Loans in would otherwise be Converted into Loans of the messed (or shall remain as) Base Rate Loans; and
- 0 if Louis of a Type (other than Base Rate Los ly Converted an the Convertion er Leaders of the-Afficial Type are subsequently Converted ther than Base Rate Louis), such Leader's Base Rate Louis read as the Conversion date for such Louis of the other

r and the Lenders whose Louis are so Co in, Types and Inserest Periods) in accordan at ascessary so that, after giving effect are so Conversed

- 8 P ermines is estrao The Com Marie 5: 12 ty shall pay to the Agent for account of each the Agent, such amount or amounts as shall 10 S COM nate it for any loss, cost or
- ext or conversion of a Fixed Rase Loan or a Set Rase Loan made reason (including, without limitation, the acceleration of the te last day of the las mod or the conve ion of Longs pursuant to Section 5.04 serest Period for such Long; or
- on the date for such borrowing to Section 2.02 or 2.03(b) B. Be Ca ne by the Co spany for any reason (includio scadest specified in Section 6 is the relev may Martin Q at actice of borrowing gives

Loan) or the Units or a Set Day ig the office of the precedu or Do shell include an amount for such Loan (or. ader would have LIBOR Market is a CD Loss BOCO S of suc 8

## Section 6. Conditions Procedure.

- 6.01 <u>laited Loss</u>. The obligation of any Leader to make its initial Loan sides to the receipt by the Agent of the following documents, each of which shall to the Agent in form and substance:
- laws of the Company of and all corporate action taken by see and borrowings by the Company Certified copies of the articles of incorporation and by-corporate action taken by the Company approving this arrowings by the Company hereunder (including, without

1 setting forth the resolutions of the Board of Directors of the spect of the transactions contemplated hereby).

- my in respect of each of the any to the contrary). achusively rely or givin
- 6.02 Officer's Ca er of the Con 788 A certificate of a vice president or treas the effect set forth in the first semence of in the first sentence of Section
- The Notes, duty com stated and executed and delivered
- torn of Exhibit 3 harms e Co and General Counsel of the Company, substantially in
- PSC Agencyal. A copy of the order of the Florida Public Service therizing the execution and delivery by the Company of the Notes and its
- Ģ Such other docu ments as the Agent or any Lender may
- 700: (i) BO ditions

Section 7. Warne The Com pany represents and

- 7.01 Corporate Existence. The Company: (a) is a corporation duly organized and validly existing under the laws of the State of Florida; (b) has all requisite corporate power, and has all material governmental licenses, authorizations, consents and approvals necessary to own its assets and carry on its business as now being or as proposed to be conducted; and (c) is qualified to do business in all jurisdictions in which the nature of the business conducted by it makes such qualification necessary and where failure so to qualify would have a material adverse effect on the consolidated financial condition, operations or business taken as a whole of the Company and its Consolidated Subsidiaries.
- 7.02 Financial Condition. The audited consolidated balance sheet of the Company and its Consolidated Subsidiaries as at December 31, 1996 and the related consolidated statements of income, shareholders' equity and cash flow of the Company and its Consolidated Subsidiaries for the flecal year ended on said date, with the opinion thereon of KPMG Peat Marwick LLP, and the unsudited consolidated balance sheet of the Company and its Consolidated Subsidiaries as at September 30, 1997 and the related consolidated statements of income and cash flow of the Company and its Consolidated Subsidiaries for the nine-month period ended on such date, heretofore furnished to each of the Lenders, are complete and correct and fairly present the consolidated financial condition of the Company and its Consolidated Subsidiaries as at said dates and the consolidated results of their operations for the fiscal year and nine-mouth period ended on said dates (subject, in the case of such financial statements as at September 30, 1997, to normal year-end audit adjustments), all in accordance with generally accepted accounting principles and practices applied on a consistent basis (provided that such financial statements may contain condensed footnotes prepared in accordance with Rule 10-01(a)(5) of Securities and Exchange Commission Regulation S-X). Neither the Company nor any of its Subsidiaries had on said dates any contingent liabilities, liabilities for taxes, unusual forward or long-term commitments or unrealized or anticipated losses from any unfavorable commitments, in each case material to the Company and its Consolidated Subsidiaries taken as a whole, except as referred to or reflected or provided for in said balance sheets as at said dates. per 30, 1997, there has been no material adverse change in the consolidated financial condition, operations or business taken as a whole of the Company and its Consolidated Subsidiaries from that set forth in said financial statements as at said date.
- 7.03 Litigation. Except for the matters disclosed in the Company's Annual Report on Form 10-K for the year ended December 31, 1996, and the Company's Quarterly Reports on Form 10-Q for the quarters ended March 31, June 30, and September 30, 1997, there are no legal or arbitral proceedings or any proceedings by or before any governmental or regulatory authority or agency, now pending or (to the knowledge of the Company) threatened against the Company or any of the Company's Subsidiaries which, if adversely determined, could have a material adverse effect on the consolidated financial condition, operations or business taken as a whole of the Company and its Consolidated Subsidiaries.
- 7.04 No Breach. None of the execution and delivery of this Agreement and the Notes, the consumention of the transactions herein contemplated and compliance with the terms and provisions hereof and thereof will conflict with or result in a breach of, or require any

t to the terms of any such agreem on or by-laws of the Comp ment, or result in the creation es of the Company or any of R OF IN ay, or any applicable law or governmental authority 2

- STATE OF sary corporate power and greement and the Notes; f the Company; an cat and the Notes have spany and each such my and delivered for
- 7.8 als. Other than the approval of the Florida Public Service a. approvals or consents of, and so filings or registrations with, ry ambarity or approxy are measurery for the execution, delivery or my of this Agreement or the Notes, or for the validity or
- 7.07 Use of Losse. The Company is not engaged principally, or as one of its tivities, in the business of extending credit for the purpose, whether immediate, ultimate, of buying or carrying Margin Stock and so part of the proceeds of any for will be used to buy or carry any Margin Stock.
- 7.0 BISA and the ERISA Affiliates have fulfilled their ding standards of ERISA and the Code with respect maserial respects with the presently applicable act incurred any liability to the PBGC or any Plan contributions in the ordinary course of business).

- Subsidiaries have been examined and closed through the fiscal year of the Parent and its December 31, 1983. The Parent and its Subsidiaries have filed all United States Federal income tax returns and all other material tax returns which are required to be filed by them and have paid all taxes due purment to such returns or purment to any assessment received by the Parent or any of its Subsidiaries. The charges, accruals and reserves on the books of the Parent its Subsidiaries in respect of taxes and other governmental charges are, in the opinion of the Company, adoquate. If the Parent is a member of an affiliated group of corporations filing consolidated returns for United States Federal income tax numbers is in it. of such group.
- or a company "controlled" by a Company Act of 1940, as asse 7.5 dad by as 'break an "investment company", within the meaning of the Investment
- Company' within the meaning of Section 2(a)(5) of the Public Utility Holding Company Act of 1935 (the "1935 Act"). The Company is not a "holding company" within the meaning of Section 2(a)(7) of the 1935 Act. The Company is not a "holding company" within the meaning of Section 2(a)(1) of the 1935 Act. The Company is an "affiliate" within the meaning of Section 2(a)(8) of the 1935 Act. The Purent is a "holding company" of the Purent within the meaning of Section 2(a)(8) of the 1935 Act. The Purent is a "holding company" within the meaning of Section 2(a)(7) of the 1935 Act, but is entitled to and currently claims the benefits of an exemption from the requirements of the 1935 Act (other than Section 9(a)(2) thereof) pursuant to Rule 2 under Section 3(a)(1) of the 1935 Act. The Purent has filed with the Securities and Exchange Commission all documents that are necessary to maintain such exemption from the India force and effect. Such exemption in full force and effect and such exemption from all requirements of the 1935 Act relating to the Company's status as a "subsidiary company" of the Purent. Neither the Company or the Purent has received any notification from the Securities and Exchange Commission under Rule 6 under the 1935 Act with respect to its status under the 1935 Act. Except for proceedings that may arise abould the Securities and Exchange Commission with a basis for secking to regulate the Company as a holding company under the 1935 Act or for secking to revoke the reamption under the 1935 Act or for secking to revoke the reamption of the purvletions of the 1935 Act or any rule or regulation thereunder, for purposes of Section 26(c) thereof.

Section 8. Covernants of the Company. The Company agrees that, so long as any of the Commitments are in effect and until payment in full of all Loans hereunder, all interest thereon and all other amounts payable by the Company hereunder: Section 8.

## 8.01 Financial Statements. The Company shall deliver to each of the Leaders:

- (a) as soon as available and in any event within 60 days after the end of each quarterly fiscal period of each fiscal year of the Company, consolidated statements of income and cash flow of the Company and its Consolidated Subsidiaries for such period and for the period from the beginning of the respective fiscal year to the end of such period, and the related consolidated balance sheet as at the end of such period, setting forth in each case in comparative form the corresponding consolidated figures for the corresponding period in the preceding fiscal year, accompanied by a certificate of the Treasurer, an Assistant Treasurer, the Chief Financial officer or the Controller of the Company, which certificate shall state that said financial statements fairly present the consolidated financial condition and results of operations of the Company and its Consolidated Subsidiaries in accordance with generally accepted accounting principles, consistently applied, as at the end of, and for, such period (subject to normal year-end audit adjustments and provided that such financial statements may contain condensed footnotes prepared in accordance with Rule 10-01(a)(5) of Securities and Exchange Commission Regulation S-X);
- (b) as soon as available and in any event within 120 days after the end of each fiscal year of the Company, consolidated statements of income, shareholders' equity and cash flow of the Company and its Consolidated Subsidiaries for such year and the related consolidated balance short as at the end of such year, setting forth in each case in comparative form the corresponding consolidated figures for the preceding fiscal year, and accompanied, in the case of said consolidated statements and balance short, by an opinion thereon of independent certified public accountants of recognized national standing, which opinion shall state that said consolidated financial statements fairly present the consolidated financial condition and results of operations of the Company and its Consolidated Subsidiaries as at the end of, and for, such fiscal year;
- (c) promptly upon their becoming available, copies of all registration statements and regular periodic reports, if any, which the Company shall have filed with the Securities and Exchange Commission (or any governmental agency substituted therefor) or any national securities exchange;
- (d) promptly upon the mailing thereof to the shareholders of the Parent generally, copies of all financial statements, reports and proxy statements so mailed;
- (e) if any of the events or conditions specified below with respect to any Plan or Multiemployer Plan shall have occurred or exist, promptly upon filing any required notice thereof with PBGC, a copy of such notice or other report to PBGC and a statement signed by a senior financial officer of the Company setting forth details respecting such event or condition and the action, if any, which the Company or an ERISA Affiliate proposes to take with respect thereto:

- ERISA shall be a reportable event regardless of the iss accordance with Section 412(d) of the Code); ortable event, as defined in Section 4043(b) of ERISA and thereunder, with respect to a Plan, as to which PBGC has sed the requirement of Section 4043(a) of ERISA that it be ing standard of Section 412 of the Code or Section 302 ce of such even ace of any waivers
- the filing under Section 4041 of ERUSA of a notice of intent to see any Plan or the termination of any Plan;
- months by the Cor the lastitution by PBGC of proceed ion of, or the appointment of a trustee to administer, any the Company or any ERISA Affiliate of a notice from a such action has been taken by PBGC with respect to such gs under Section 4042 of
- Plan, or the manips by the Company
  Multimaployer Plan that it is in re-4041A of ERISA; and 8 0 lets or partial withdrawal by the Company, the Parent or r Section 4301 or 4304 of ERISA from a Multien be Company or any ERISA Affiliate of notice ization or insolvency pursuant to Section 8 00 10 emin
- ERISA, which proof estination of a proceeding by a fiduciary of any Multiemployer tempany or any ERISA Affiliate to enforce Section 515 of seeding is not dismissed within 30 days;
- (f) promptly after the Company knows or has has occurred, a notice of such Default describing the same in a such notice is a "Notice of Default" and, together with such 2 200 Con y knows or has reason to know that any Default the same in reasonable detail, specifying that has taken and proposes to take with such notice or as soon thereafter
- (g) from time to time such other information regarding the business, affairs or financial condition of the Company or any of the Company's Subsidiaries (including, without limitation, any Plan or Multimaphoyer Plan and any reports or other inf...mation required to be filed under ERISA) as any Lander or the Agent may reasonably request.

Default has occurred and is continuing (or, if any 0 ef Financial Officer eraph (a) or (b) above, a certificate of the Treasurer, an Assistant er or the Contro certificate or other of the Con Default la of the Treasurer, an Assistant Company to the effect that no as occurred and is continuing.

describing the same in reasonable detail and describing the action that the Company has taken and proposes to take with respect thereto).

- 8.02 <u>Litiention</u>. The Company will promptly give to each Lender rotice (which notice may be given by the Company through the Parent and may be in the form of the Company's or the Parent's 1934 Act Reports) of all legal or arbitral proceedings, and of all proceedings by or before any governmental or regulatory authority or agency, and any material development in respect of such legal or other proceedings, affecting the Company or any of the Company's Subsidiaries, except proceedings which, if adversely determined, would not have a material adverse effect on the consolidated financial condition, operations or business taken as a whole of the Company and its Consolidated Subsidiaries.
- 8.03 Cornorate Existence. Etc. The Company will: preserve and maintain its corporate existence and all of its material rights, privileges and franchises; comply with the requirements of all applicable laws, rules, regulations and orders of governmental or regulatory authorities if failure to comply with such requirements would materially and adversely affect the consolidated financial condition, operations or business taken as a whole of the Company and its Consolidated Subsidiaries, except for any such laws, rules, regulations or orders that the Company is contesting in good faith and by proper proceedings; pay and discharge all taxes, assessments and governmental charges or levies imposed on it or on its income or profits or on any of its property prior to the date on which possities attach thereto, except for any such tax, assessment, charge or levy the payment of which is being contested in good faith and by proper proceedings and against which adequate reserves are being maintained; maintain all of its properties used or useful in its business in good working order and condition, ordinary wear and tear excepted; and permit representatives of any Lender or the Agent, during normal business hours, to examine, copy and make extracts from its books and records, to inspect its properties, and to discuss its business and affairs with its officers, all to the extent reasonably requested by such Lender or the Agent (as the case may be).
- 8.04 <u>Prohibition of Fundamental Changes</u>. The Company will not enter into any transaction of merger or consolidation or amalgamation, or liquidate, wind up or dissolve itself (or suffer any liquidation or dissolution), except that the Company may merge into the Parent so long as the Parent is the surviving corporation and assumes all of the Company's obligations under this Agreement and the Notes and so long as both prior to such merger and after giving effect thereto no Default shall have occurred and be continuing. The Company will not convey, sell, lease, transfer or otherwise dispose of, in one transaction or a series of transactions, all or any substantial part of its business or assets, whether now owned or hereafter acquired.
- 8.05 Use of Proceeds. The Company will use the proceeds of the Loans hereunder for its general corporate purposes (in compliance with all applicable legal and regulatory requirements); provided that neither the Agent nor any Lender shall have any responsibility as to the use of any of such proceeds.

- 8.06 Indebtates to Capitalization Ratio. The Company will not permit the ratio of (a) all Indebtates of the Company and its Consolidated Subsidiaries to (b) the Total Capitalization of the Company and its Consolidated Subsidiaries to exceed .65 to 1 at any time.
- Subsidiary to, cres es, whether now own 9 or hereafter acquired, except for: ŀ to exist any Lies on any of its property, assets or The Comp my will not, and will not permit any
- as secured by the Existing Indent Liens existing on the date hereof securing Indebtedness, other than secured by the Existing Indeasure, outstanding on October 31, 1997 in an eigel amount not in excess of \$240,865,000;
- such event; 3 say Lies existing on any asset of any corporation at the time such accesses a Subsidiary of the Company and not created in contemplation of
- (c) any Lies on any asset securing Indobtedness incurred or assumed for the purpose of financing all or any part of the cost of acquiring such asset, provided the such Lies attaches to such asset concurrently with or within 90 days after the acquisition or assumed for the
- (d) any Lies on any asset of any corporation existing at the time such corporation is manipul into or is compositioned with the Company or one of its Subsidiaries and not creased in communication of such event;
- (c) any Lien existing Company or one of its Subsidiar g on any asset prior to the acquisition thereof by the ries and not creased in commemplation of such acquisition;
- (f) any Lien arising out of the refinancing, extension, renewal or refunding of any Indebtedness secured by any Lien permitted by any of the foregoing clauses of this Section, provided that such Indebtedness is not increased and is not secured by any asy Lies orisis
- (g) any Lien arising pursuant to any order of attachment, distraint or similar legal process arising in commention with court proceedings so long as the execution or other emforcement thereof is effectively stayed and the claims secured thereby are being consessed in good faith by appropriate proceedings;
- (h) Lieus incidental to conduct of its business or the ownership of its assets which (i) do not secure Indobsedness and (ii) do not in the approprie materially detract from the value of its assets or materially impair the use thereof in the operation of its
- (i) the Liens created by the Existing Indenture; and

- statutes to condemn and acquire examinate or rig over the land covered by the rights-of-way in qu by for transmission or distribution line purposes; its Consolidated Subsidiaries, as the case may be its Consolidated Subsidiaries, as the case may be tion or other lands adjaces
- Section 9. Events of Default. If one or more of the following events (herein called "Events of Default") shall occur and be continuing:
- (a) The Company shall default in the payment when due of any principal of or immerest on any Loan or any other amount payable by it hereunder; or
- b) The Company or any of its Subsidiaries at when the of any principal of or imment on any of its ing \$10,000,000 or more; or any event specified in any or other document evidencing or relating to any a game amount set forth above in this clause (b) shall on its to came, or (with the giving of any notice or the contraction of the cont ptics or the lapse of time or disess (or a truese or agest additionass to become due, or otherwise), prior to occur if the effect of
- (c) Any representation, warranty or certification made or deemed made herein (or in any modification or supplement hereto) by the Company, or any certificate furnished to any Londor or the Agent pursuant to the provisions hereof (or thereof), shall prove to have been false or mideading as of the time made or furnished in any material respect; or
- Detail (d) The Company shall default in the performance of any of its biguiness under any of Sections 8.01(f) or 8.04 hereof; or the Company shall that in the performance of any of its other obligations in this Agreement and the default shall continue unremedied for a period of 30 days after notice thereof the Company by the Agent or any Lender (through the Agent); or any "Event Default" under and as default in the Other FPC Agreement shall be
- (e) The Company or any of the Company's Subsidiaries shall admit in the immbility to, or be generally unable to, pay its debts as such debts a due; or
- The Company or any of the Company's Subsidiaries shall (i) apply at to the appointment of, or the taking of possession by, a receiver.

custodien, trustee or liquidator of itself or of all or a substantial part of its property, (ii) make a general assignment for the benefit of its creditors, (iii) commence a voluntary case under the Bankruptcy Code (as now or hereafter in effect), (iv) file a petition seeking to take advantage of any other law relating to bankruptcy, insolvency, reorganization, winding-up, or composition or readjustment of debts, (v) fail to controvert in a timely and appropriate manner, or acquiesce in writing to, any petition filed against it in an involuntary case under the Bankruptcy Code (as now or hereafter in effect), or (vi) take any corporate action for the purpose of effecting any of the foregoing; or

- (g) A proceeding or case shall be commenced, without the application or consent of the Company or any of the Company's Subsidiaries, in any court of competent jurisdiction seeking (i) its liquidation, reorganization, dissolution or winding-up, or the composition or readjustment of its debts, (ii) the appointment of a trustee, receiver, custodian, liquidator or the like of the Company or such Subsidiary or of all or any substantial part of its assets, or (iii) similar relief in respect of the Company or such Subsidiary under any law relating to bankruptcy, insolvency, reorganization, winding-up, or composition or adjustment of debts, and such proceeding or case shall continue undiamised, or an order, judgment or decree approving or ordering any of the foregoing shall be entered and continue unstayed and in effect, for a period of 60 or more days; or an order for relief against the Company or such Subsidiary shall be entered in an involuntary case under the Bankruptcy Code (as now or hereafter in effect); or
- (h) A final judgment or judgments for the payment of money in excess of \$10,000,000 in the aggregate shall be rendered by a court or courts against the Company and/or any of the Company's Subsidiaries and the same shall not be discharged (or provision shall not be made for such discharge), or a stay of execution thereof shall not be procured, within 30 days from the date of entry thereof and the Company or the relevant Subsidiary shall not, within said period of 30 days, or such longer period during which execution of the same shall have been stayed, appeal therefrom and cause the execution thereof to be stayed during such appeal; or
- (i) An event or condition specified in Section 8.01(e) hereof shall occur or exist with respect to any Plan or Multiemployer Fian and, as a result of such event or condition, together with all other such events or conditions, the Company, the Parent or any ERISA Affiliate shall incur or in the opinion of the Majority Landers shall be reasonably likely to incur a liability to a Plan, a Multiemployer Plan or PBGC (or any combination of the foregoing) which is, in the determination of the Majority Lenders, material in relation to the consolidated financial condition, operations or business taken as a whole of the Company and its Consolidated Subsidiaries;

- (j) Any authorization, approval or consent of any Governmental or regulatory authority or agency necessary for the execution, delivery or performance by the Company of this Agreement or the Notes, or for the validity or enforceability thereof, shall, after the date of the receipt by the Agent of the order referred to in Section 6.01(g) hereof, cease to be in full force and effect; or
- (k) The Company shall cease to be a Wholly-owned Subsidiary of the Parent;

THEREUPON: (1) in the case of an Event of Default other than one referred to in clause (f) or (g) of this Section 9 with respect to the Company, (A) the Agent may and, upon request of the Majority Lenders, shall, by notice to the Company, cancel the Commitments and they shall thereupon terminate, and (B) the Agent may and, upon request of the Majority Lenders shall, by notice to the Company declare the principal amount then outstanding of, and the accrued interest on, the Leans and all other amounts payable by the Company hereunder and under the Notes (including, without limitation, any amounts payable under Section 5.05 hereof) to be forthwith due and payable, whereupon such amounts shall be immediately due and payable without presentment, domand, present or other formalities of any kind, all of which are hereby expressly varived by the Company; and (2) in the case of the occurrence of an Event of Default referred to in clause (f) or (g) of this Section 9 with respect to the Company, the Commitments shall automatically be canceled and the principal amount then outstanding of, and the accrued interest on, the Leans and all other amounts payable by the Company hereunder and under the Notes (including, without limitation, any amounts payable under Section 5.05 hereof) shall automatically become immediately due and payable without presentment, demand, protest or other formalities of any kind, all of which are hereby expressly waived by the Company.

## Section 10. The Asset.

10.01 Assointment. Powers and Immunities. Each Lender hereby irrevocably appoints and suthorises the Agent to act as its agent hereunder with such powers as are specifically delegated to the Agent by the terms of this Agreement, together with such other powers as are reasonably incidental thereto. The Agent (which term as used in this sentence and in Section 10.05 and the first sentence of Section 10.06 hereof shall include reference to its affiliates and its own and its affiliates' officers, directors, employees and agents): (a) shall have no duties or responsibilities except those expressly set forth in this Agreement, and shall not by reason of this Agreement be a trustee for any Lender; (b) shall not be responsible to the Lenders for any recitals, statements, representations or warranties contained in this Agreement, or in any certificate or other document referred to or provided for in, or received by any of them under, this Agreement, or for the value, validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement, any Note or any other document referred to or provided for herein or for any failure by the Company or any other Person to perform any of its obligations hereunder or thereunder; (c) shall not be required to initiate or conduct any litigation or collection proceedings hereunder; and (d) shall not be responsible for any action taken or omitted

IS COMMENT OF the of my Note ferred to or provided willful misconduct. y it in good faith.
If for all purposes
all have been filed heat or transfer. 

高の 見のな thereof by telephone, telex, telegram have been signed or sent by or on and statements of legal counsel. es be fully protected in estructions signed by the my action taken or failure

ed.D 10.00 pect to such De until the Agent shall have ed to) take such action, or shall deem advisable in the age or notice of set on Loans or

10.04 herefor to any Lender) accept d of banking, trust or other m) as if it were not acting as et and the Loans made r herounder shall have ut having-to accoust ideration from the N SMCC

if or of any such

independently and without reliance on the A documents and information on it has decision independently and without reliance upon the Adocuments and decisions in this or any and an information on it shall does a sandysis and decisions in using or any string a sandysis and decisions in using or any string a sandysis and decisions in using or any of its Seek to required to large least full and any of its Seek to books of the Company or any of its Seek to books of the Agent shall not have any day of the Agent shall not have any day of credit or other information consuming the Company or any of its Seekitisties (or any Company or any of its Seekitisties (or any constraints). is for action, reports and other ad to the Lenders by the Agent to provide any Lender with any a condition or business of the mes) which may come into the A CO BE

10.07 Except for action expressly required of the Agent to the fully justified in falling or refluing to act hereunder reaces to its satisfaction from the Lenders of their tion 10.05 hereof against any and all liability and expense a of taking or continuing to take any such action.

\$ \$ F 0.0 e within 30 days

Agent, which shall be a bank which has an office in New York, New York and (unless it is a Lender) a combined capital and surplus of at least \$200,000,000. Upon the acceptance of any appointment as Agent horounder by a successor Agent, such successor Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Agent, and the retiring Agent shall be discharged from its duties and obligations hereunder. After any retiring Agent's resignation or removal hereunder as Agent, the provisions of this Section 10.08 shall continue in effect for its benefit in respect of any actions taken or omitted to be taken by it while it was acting as the Agent.

10.09 Agency Fee. So long as the Commitments are in effect and until payment in full of the principal of and interest on the Loans and all other amounts payable by the Company horounder, the Company will pay to the Agent an agency fee in an amount previously agreed, payable annually in advance on the Quarterly Date falling on or nearest to December 1 in each year. Such fee, once paid, shall be non-refundable. The appointment of a successor Agent under Section 10.08 hereof shall not increase or otherwise modify the Company's obligations under this Section 10.09.

10.10 <u>Auction For</u>. The Company agrees to pay to the Agent an auction fee in an amount previously agreed, payable on each day that the Company delivers a Money Market Quote Request.

### Section 11. Miscellaneous.

11.01 Waiver. No failure on the part of the Agent or any Londer to exercise and no delay in exercising, and no course of dealing with respect to, any right, power or privilege under this Agreement or any Note shall operate as a waiver thereof, nor shall any single or partial exercise of any right, power or privilege under this Agreement or any Note preclude any other or further exercise thereof or the exercise of any other right, power or privilege. The remedies provided become are cumulative and not exclusive of any remedies provided by law.

(including, without limitation, any modifications of, or waivers or consents under,. this Agreement) shall be given or made by telex, telecopy, telegraph, cable or in writing (or, with respect to notices given pursuant to Section 2.03 hereof, by telephone, confirmed in writing by telex by the close of business on the day the notice is given) and telexed, telecopied, telegraphed, cabled, mailed or delivered (or telephoned, as the case may be) to the intended recipient at the "Address for Notices" specified below its name on the signature pages hereof; or, as to any party, at such other address as shall be designated by such party in a notice to each other party. Except as otherwise provided in this Agreement, all such communications shall be deemed to have been duly given when transmitted by telex or telecopier, delivered to the telegraph or cable office or personally delivered or, in the case of a mailed notice, upon receipt, in each case given or addressed as aforesaid.

## 11.03. Example Br.

- 5 ad the making \*\*\*
- expenses of the Agent (including reason, modification or waiver of any of the t for the Agent) in co wrse each of the Leaders and the s of the Leaders and t
- 0 7 8 None or my over atal or revenue authority in respect it referred to herein. P. docume lary or other
- reof or of this = 11.04; OF (V) of principal thereof or the rate at which interest is made; (iv) after the terms of Section 8.06 or 11.06(a) and the definition of the term "Majority Lendon"; and Section 10 hereof shall also require the consent of the 07 04 any provision of this Agreem expressly provided in this odified only by an instrument reduction or termination. an of the which interest is 8.06 or 11.06(a)
- 11.03 Sans hereto and their respective successors and permitted assigns.

# 11.06 Assignment and Participations

3 The Company may not seeign its rights or obligation the prior comment of all of the Lenders and the Agent.

Mer of

all or any part of any participation, no such

- may assign and pit as collaseral securi 3 Anything in this Section 11.06 to the contrary notwithstanding, any Lender tige all or any portion of its Louis and its Notes to any Federal Reserve Bank y pursuant to Regulation A of the Board of Governors of the Federal Reserve CO No such assign
- Parest or as and participants deir respect A Leader ay, to pro I, with notice to the Company, to prospective participants and information concerning the Company, the possession of such Lender from time to time
- 3
- 8 any Leader makes a demand for payment under Section 5.01 hereof,
- (ii) Loss of any Lander are Conversed into Base Rate Loans pursuant to Section 5.04 hereof, or
- riber of the PCH Ap 0.0 8 of 'Commission Termination Day' nex Termination Date" in Section 1.01 hereof), does not of under (and as defined in) the Other FFC Agreement or to each case within the required periods set forth in the ting Lender (as that term is used in

ender's Con all sox b y, for a p ect to the come to the comment of the Agen ch Lender assign, pursuan th Lender, all (but not less to take place no later than 30 days after price not less than the principal amount and all other amounts payable by the s payable under Section 5.05 hereof) and ter to one or more banks or financial Company akes the payment it of the relevant e Company may, at (which consent any may.

- 11-03 hereof shall su 11.07 Servine. The of The obligations of the Company under Sections 5.01, 5.05 and spayment of the Loans and the termination of the Commitments.
- 11.08 Captions. The table of contents in are included solely for convenience of retion of any provision of this Agreement. and captions and section headings ference and are not intended to affect

- 11.09 Counterparts. This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same instrument and any of the parties hereto may execute this Agreement by signing any such counterpart.
- 11.10 Governing Law: Submission to Jurisdiction. This Agreement and the Notes shall be governed by, and construed in accordance with, the law of the State of New York. The Company hereby submits to the nonexclusive jurisdiction of the United States District Court for the Southern District of New York and of any New York state court sitting in New York City for the purposes of all legal proceedings arising out of or relating to this Agreement or the transactions consumplated hereby. The Company irrevocably waives, to the fullest extent permitted by law, any objection which it may now or hereafter have to the laying of the venue of any such proceeding brought in such a court and any claim that any such proceeding brought in such a court has been brought in an inconvenient forum.
- 11.11 Waiver of lany Irial. EACH OF THE COMPANY, THE AGENT AND THE LENDERS HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

## FLORIDA POWER CORPORATION

By: Hamela (1 )a.
Title: Pamela A. Saari
Assistant Tressurer

Address for Notices:

Post Office Box 14042 St. Petersburg, Florida 33733

Telecopier No.: (813) 866-4021

Telephone No.: (813) 866-4400

Attention: Mr. Kenneth E. McDonald

Sworn to and subscribed before me on this 17th day of November, 1997, in the State of New York, County of New York.

Notary Public

CAROL A MAYERS
MOTARY PUBLIC State of New York
No 01MA5084316
Qualified in Bronz County
Commission Expires Sept. 2, 1999

p:\credit.eg/pressib.f

S33,675,000

## THE CHASE MANHATTAN BANK

, All

Title: PAUL V. FARRELL VICE PRESIDENT

Lending Office for all Loans (other than Eurodollar Loans):

The Chase Manhattan Bank 270 Park Avenue New York, New York 10017-2070

Leading Office for Eurodollar Loans:

The Chase Manhettan Bank Cayman Islands, British West Indies Branch c/o The Chase Manhettan Bank One Chase Manhettan Plats New York, New York 10081

## Address for Notices:

The Chase Mashattan Bank Global Power & Environmental Group 270 Park Avenue New York, New York 10017-2070

Telecopier No.: 212/270-3089

Telephone No.: 212/270-7653

Attention: Mr. Paul V. Farrell

Commitment \$28,875,000

MORGAN GUARANTY TRUST COMPANY OF NEW YORK

Title:

Philip S. Derjens Vice President

Leading Office for all Loans (other than Eurodollar Loans):

Morgan Guaranty Trust Company of New York 60 Wall Street New York, New York 10260-0060

Leading Office for Eurodollar Loans:

Morgan Gueranty Trust Company of New York Nesses, Bahamas c/o J. P. Morgan Services, Inc. Euro-Loan Securities Unit 902 Martest Street Wilmington, Deloware 19801

Address for Notices:

Morgan Guaranty Trust Company of New York 60 Wall Street New York, New York 10260-0060

Telecopier No.:

212/648-5014

Telephone No.:

212/648-8454

Attention: Mr. Phillip S. Detjens

Commitment \$28,875,000

NATIONSBANK, N.A.

By Ticle: SV

Leading Office for all Loans:

NationsBank, N.A. 101 North Tryon Street Charlotte, NC 28255 Attn: Judy Dudley Credit Services

Address for Notices:

NationsBank, N.A. 21st Floor 600 Peachtree Street, N.E. Atlanta, GA 30308-2213

Telecopier No.:

404/607-6465

Telephone No.:

404/607-5561

Attention: Mr. McKie M. Trotter II

S28,875,000

FIRST UNION NATIONAL BANK

Title: WOUND AND ORDERS

MICHAEL J. KOLOSOWSKY

Leading Office for all Loans

First Union National Bank
One First Union Center
301 South College Street
Charlotte, North Carolina 28288-0745

Address for Notices:

First Union National Bank
One First Union Center
301 South College Street
Charlotte, North Carolina 28288-0745

Telecopier No.:

704/374-2802

Telephone No.:

704/143-7686

Attention: Mr. David M. Johnson

S22,200,000

SUNTRUST BANK, TAMPA BAY

Dy Appl

Leading Office for all Loans:

SunTrust Bank, Tampa Bay 300 1st Avenue South St. Petersburg, Florida 33701

Address for Noticus:

SunTrust Beak, Tampa Boy 300 Let Avenue South St. Petersburg, Florida 33701

Telecopier No.:

813/892-4810

Telephone No.:

813/892-4958

Attention: Ms. Brigitta A. Lawton

Sworn to and subscribed before me on this 17th day of November, 1997, in the State of New York, County of New York.

Notary Public

CAROL A MAYERS
NOTARY PUBLIC. State of New York
No. 01MASOB4316
Qualified in Brans County
Commission Expires Sept. 2, 1999

S21,250,000

## THE FIRST NATIONAL BANK OF CHICAGO

Tide: AUTHORIZED AGENT

Leading Office for all Loans:

The First National Bank of Chicago One First National Plaza Suite 0363 Chicago, Illinois 60670-0363

Address for Notices:

The First National Bank of Chicago One First National Plaza Suite 0363 Chicago, Illinois 60670-0363

Telecopier No.:

312/732-3055

Telephone No.:

312/732-9781

Attention: Mr. William N. Benks

S21,250,000

PNC BANK, NATIONAL ASSOCIATION

By Cotton Uhrana

Leading Office for all Loans:

PNC Bank, National Association One PNC Plaza 3rd Floor 249 - 5th Avenue Pinsburgh, PA 15222-2707

Address for Notices:

PNC Bank, National Association One PNC Plans 3rd Floor 249 - Sth Avenue Pinsburgh, PA 15222-2707

Telecopier No.:

412/762-2571

Telephone No.:

412/762-2540

Attention: Mr. Christopher N. Moravec

THE CHASE MANHATTAN BANK

Tiele:

PAUL V. FARRELL

Address for Notices to Chase as Agent:

The Chase Manhattan Bank 1 Chase Manhattan Plaza - 8th Floor New York, New York 10081

Telecopier No.:

212/552-7490

Telephone No.:

212/552-7943

Attention: Mr. Municum Appenne

## BARNETT BANK, N.A., PINELLAS COUNTY

Title: Sold

Leading Office for all Loans:

Bernett Bank, N.A., Pinelles County 200 Central Avenue Suite 1800 St. Petersburg, Florida 33701

Address for Notices:

Bernstt Bank, N.A., Pinelles County 200 Central Avenue Suite 1800 St. Petersburg, Florida 33701

Telecopier No.:

813/892-1545

Telephone No.:

813/892-1518

Attention: Mr. Michael S. Crowe

Sworm to and subscribed before me on this 17th day of November, 1997, in the State of New York, County of New York.

Notary Public

CARRY A MAYERS

AND THE PARTY FOR THE PARTY

# [Form of Note for Syndicated Louis]

## PROMISSORY NOTE

	1	•
	1	
	F۱	
- 1	9	
3	M	
	*	
-	Z	
- 1		
:	< 3	Ř
;	11	

ED, FLORIDA POWER CORPORATION, a Fix 

set Period (if applicable) of ior to any transfer stimustion thereof,

and Restated Credit time, the 'Credit ders manned therein to repay the

The Credit Agreement provides for the acceleration of the maturity of this Note upon the occurrence of certain events and for propayments of Lones upon the terms and conditions specified therein.

Except as permitted by Section 11.06(b) of the Credit Agreement, this Note may not be assigned by the Lender to any other Person.

This Note shall be governed by, and construed in accordance with, the law of the State of New York.

Dv:			
	 	 	٠
Title:			

# SCHEDULE OF LOANS

This Note Schedule describes Loans made, Continued or Converted under the withindescribed Credit Agreement to the Company, on the dates, in the principal amounts, of the Types, bearing interest at the rates and maturing on the dates set forth below, subject to the payments and prepayments of principal set forth below:

inued or Con-	of	Laterest	Duration of Interest	Date and Amount Paid or	Uspaid Principal	Notation
verted	Loss	Pare	Period	Expeld	Amount	Made by

# [Form of Note for Money Market Louis

# PROMISSORY NOTE

New York, New York

FOR VALUE RECEIVED, FLOR NIDA POWER CORPORATION, a Florida corporation provided for by the Credit

ã time of each Money Market Loan made on account of the principal account of the princip my transfer of this No

ö Š ġ

The Credit Age events and for propayments of Louis upon the terms and conditions

Except as permissed by Section 11.06(b) of the Credit Agree of by the Leader to any other Person. Mat. this Note may not be

This Note shall be governed by, and construed in accordance with, the law of the Sizte of New York.

FLORIDA POWER CORPORATION

Dy:\_\_\_\_\_\_ Tide:

# SCHEDULE OF LOANS

This Note Schedule describes Loans made under the within-described Credit Agreement to the Company, on the dates, in the principal amounts, of the Types, bearing interest at the rates and maturing on the dates set forth below, subject to the payments and propayments of principal set forth below:

	Principal					Date and	
				Materity	Amount	Unpaid	
of	of	of	Lateron	Date of	Amount Paid or	Unpaid Principal	Nomice
Loss					Proposid	Amount	

# [Form of Opinion of Counsel to the Purent]

			討
I	below	Credit	
}	L		ł
I	0		S AME
	ï	I	į
	I	8	

# Ladies and Gentlemen

ZE F

# Based upon the foregoing. I am of the opinion that:

- 3. The Credit Agreement has been duly executed and delivered by the Company and constitutes, and the Notes when executed and delivered for value will constitute, legal, valid and binding obligations of the Company, enforceable in accordance with their respective terms, except as such enforceability may be limited by (a) bankruptcy, insolvency, reorganization, moretorium or other similar laws of general applicability coment of creditors' rights and (b) the application of general principles affecting the onfo of equity (regardless of whother such enforceability is considered in a proceeding in equity or at law), and except that no opinion is expressed as to (i) Section 4.07(c) of the Credit Agreement or (ii) the effect of the law of any jurisdiction (other than the State of Florida) wherein any Lander (including any of its Applicable Leading offices) may be located which limits rates of interest which may be charged or collected by such Leader. I express no opinion as to (i) whether a Federal or state court outside of the State of New York would give effect to the choice of New York law provided for in the Credit Agreement and the Notes, (ii) the second seasonce of Section 11.10 of the Credit Agreement, insofter as such sentence relates to the subject matter jurisdiction of the United States District Court for the Southern District of New York to adjudicate any controversy related to the Credit Agreement or the Notes, (iii) the waiver of inconvenient forum set forth in Section 11.10 of the Credit Agreemen: with respect to proceedings in the United States District Court for the Southern District of New York, or (iv) Section 11.11 of the Credit Agreement.
- 4. In connection with the above, I wish to point out that provisions of the Credit Agreement that permit the Agent or any Londor to take action or make determinations, or to benefit from indomnities and similar undertakings of the Company, may be subject to a requirement that such action or inaction by the Agent or a Londor which may give rise to a request for payment under such an undertaking be taken or not taken, on a reasonable basis and in good faith.
- 5. Except for the matters disclosed (i) under the heading "Legal Proceedings" in Part I, Item 3 of the Company's Annual Report on Form 10-K for the year ended December 31, 1996, and (ii) under the heading "Legal Proceedings" in Part II, Item 1 of the Company's Quarterly Reports on Form 10-Q for the quarters ended March 31, June 30, and September 30, 1997, there are no legal or arbitral proceedings, and no proceedings by or before any governmental or regulatory authority or agency, pending or (to my knowledge) threatened against or affecting the Company or any of the Company's Subsidiaries, or any properties or rights of the Company or any of the Company's Subsidiaries, which, if adversely determined, would have a material adverse effect on the compalidated fluoratic condition, operations or business taken as a whole of the Company and its Consolidated Subsidiaries.
- Other than the approval of the Floride Public Service Commission (which approval has been duty obtained and is in full force and effect), no authorizations, consents, approvals, licenses, filings or registrations with, any governmental or

delivery or

1935 Act and ton 2(a)(7) of the

2

other than the law of the State of Florida

Very entry yours.

# [Form of Money Market Quote Request]

۰

To:	The C	has Madata	Bank, as Agent		
From:	Florid	a Power Corpo	ration, Inc.		
Re:	Mone	Market Queen	Request		
Agranas betwees as Ages	Florida l	ed as of Nove Nower Corporat	mber 18, 1997, dies, the lenders as see that we reque	us amonded or s med therein and	it Agreement B (the " <u>Cr</u> setated from time to ti The Chase Manhattan Be t Quotes for the follow
Borrowie Date	•	Question Date[*]]	Ammily	Type (*3)	laterest Pariod(°4)
1	(erass un	d bords bove		ped to them in d	o Credit Agreement.
			By: Tisle:		

All mumbered footnotes appear on the last page of this Exhibit

- [1] For use if a Money Market Rate in a Set Rate Auction is requested to be submitted before the Borrowing Date.
- [2] Each amount must be \$10,000,000 or a larger multiple of \$1,000,000.
- [3] Insert either "Margin" (in the case of LIBOR Market Loans) or "Rate" (in the case of Set Rate Loans).
- [4] One, two, three or six months, in the case of a LIBOR Market Loan or, in the case of a Set Rate Lean, a period of up to 180 days after the making of such Set Rate Loan and ending on a Business Day.

# [Form of Money Market Quote]

To: The chase Manhattan Bank, as Agent

Attention:

Re: Money Market Quote to

Florida Power Corporation (the "Rocrower")

This Money Market Quote is given in accordance with Section 2.03(c) of the Second Amended and Research Credit Agreement B (the "Credit Agreement") dated as of November 18, 1997, as assented or restated from time to time, untween Florida Power Corporation, the leaders masted therein and The Chase Manhattan Bank, as Agest. Torms defined in the Credit Agreement are used barein as defined therein.

In response to the Berrower's invitation dated \_\_\_\_\_\_, 19\_\_, we hereby make the following Meany Market Quete(s) on the following terms:

- 1. Quoting Londor:
- 2. Person to contact at Queting Londor:
- We hereby offer to make Money Market Loan(s) in the following principal amount(s), for the following laterest Period(s) and at the following rate(s):

Borrowing Quotation Interest

Date Desc[\*1] Amount(\*2] Type(\*3) Period(\*4) Resc[\*5]

4. The maximum aggregate principal amount of all Money Market Loss:

All numbered footnotes appear on the last page of this Exhibit.

We understand and agree that the offer(s) set forth above, subject to the satisfaction of the applicable conditions set forth in the Credit Agreement, irrevocably obligates us to make the Money Market Loan(s) for which any offer(s) (is/are) accepted, in whole or in part (subject to the third sentence of Section 2.03(e) of the Credit Agreement).

Very truly yours,	
(Name of Leader)	
Dy:	
By:	

Deted: \_\_\_\_\_\_\_

- [1] As specified in the related Meany Market Quote Request.
- [2] The principal amount bid for each Interest Period may not exceed the principal amount requested. Bids must be made for at least \$5,000,000 or a larger multiple of \$1,000,000.00.
- [3] Indicate "Margin" (in the case of LIBOR Market Loans) or "Rate" (in the case of Set Rate Loans).
- [4] One, two, three or six months, in the case of a LIBOR Market Loan or, in the case of a Set Rate Loan, a period of up to 180 days after the making of such Set Rate Loan and ending on a Business Day, as specified in the related Money Market Quote Request.
- [5] For a LIBOR Market Lean, specify margin over or under the London interbank offered rate determined for the applicable Interest Period. Specify percentage (rounded to the nearest 1/10,000th of 1%) and specify whether "PLUS" or "MINUS". For a Set Rate Lean, specify rate of interest per ansum (rounded to the nearest 1/10,000th of 1%).

# EXHIBIT (a)-3

Merrill Lynch Commercial Paper Issuer memorandum

MerrillLynch Money Markets Inc

# Commercial Paper Issuer

PROGRAM REPORT

November 18, 1997

# Florida Power Corporation \$500,000,000

COMMERCIAL PAPER NOTES

# CREDIT BATTRICE

Standard & Poor's Ratings Group Mondy's Investors Service, Inc. **Duff & Photos Rating Service** 

# Commercial Proor

P-I D-1+

# TERMS OF COMMERCIAL PAPER NOTES

Florida Power Corporation (the "Company") is a subsidiary of Florida Progress Corporation. The company is an operating public utility engaged in the generation, purchase, transmission, distribution and sale of electricity within Florids.

Authorized to a maximum outstanding of US\$500,000,000.

Unsecured notes (the "Notes"), ranking pari gazes with Florida Power Corporation's other unsubordinated and unsecured indebtedness.

The Notes are exempt from registration under the Securities Act of

1933, as amended, pursuant to Section 3(a)(3), and cannot be resold

unless registered or an exemption from registration is available.

Offering Price:

Per less a discount representing an interest factor or, if interest bearing,

uch resings are only accurate as of the date hereof, as they have been obtained with the understand of & Poor's Ratings Group, Mondy's Investors Service and Duff & Photos Rating Service would come the credit of the Company and make finure adjustments to such ratings to the actest warranted. The changed, supercoded or withdraws, and therefore, a prospective purchaser should check the current.

Minimum of \$100,000.

Motorities:

Up to 270 days from date of issue.

The Notes will not be redeemed prior to meturity or be subject to voluntary

propeyment.

Form:

Each Note will be evidenced by (i) a note certificate issued in bearer form or (ii) one of two master notes (interest bearing or discount) registered in the name of the nomines of The Depository Trust Company ("DTC"). Each master note (the "Book-Entry Notes") will be deposited with the lessing and Paying Agent as subcustodies for DTC or its successor. DTC will record, by appropriate entries on its book-entry registration and transfer system, the respective amounts payable in respect of Book-Entry Notes. Paymonts by DTC participants to nurchasers for whom a DTC participant is acting as agent in respect of Book-latry Notes will be governed by the standing instructions and customery practices under which securities are held at DTC through DTC participants.

Unless otherwise agreed to, some day basis, in immediately available funds.

Isoning & Paying Agent:

Chare Manhettan Beats I Chare Manhettan Plans

Lloyd Bogs 212-552-1876

## USE OF PROCEEDS

The proceeds from the sale of the Notes will be used for current transactions.

## BANK FACILITIES

It is the Company's policy to maintain unused bank lines sufficient to back up 100% of Commercial Paper Notes ding. The Company will maintain these unused bank lines at all times except under certain conditions referred to in the Company's agreement with its bank(s).

AVAILABLE INFORMATION

Florids Power Corporation has made financial and other information readily evailable on Disclosure Inc. and Moody's/Decutronics Information Service, Inc. The Company is subject to the informational requirements of the Securities Exchange Act of 1934, as amended, and, in accordance therewith, files reports and other information with the Securities and Exchange Commission (the "Commission"). Such reports and other information may be inspected without charge at the public reference facilities maintained by the Commission at 450 Fifth Street, NW, Washington, D.C. 20549, and at the Regional Offices of the Commission. Copies thereof may be obtained from the Commission upon payment of the prescribed flow. If evallable, such reports and other information may also be accessed through the Commission's electronic data gathering, analysis and retrieval system ("EDGAR") via electronic means, including the Commission's web site on the Internet (http://www.sec.gov). Florida Power Corporation will provide without charge to each purchaser of the Notes, upon oral or written request, a copy of any and all documents filed with the Commission and any and all publicly evallable financial information. Requests ests filed with the Commission and any and all publicly evallable financial information. Requests and to: Investor Services, PO Box 33028, St. Potersburg, Fl. 33733 (813) 866-4247. Any other questions can be directed to Marrill Lynch - Instinctional Marketing at (212) 449-0233.

# EXHIBIT (a)-4

First Chicago Commercial Paper Offering Memorandum

# DEST RATINGS

3201 34th St. South St. Petersburg, FL 33711

	SAP	Mandy's
Commercial Paper	A-1+	P-1
Sr. Unsecured Debt	A+	Al
Senior Secured Debt	AA-	Aa3

November 18, 1997

# FLORIDA POWER CORPORATION 3(a)(3) COMMERCIAL PAPER PROGRAM 5000,000,000

Florida Power Corporation (the "Company") issues short-term notes (the "Notes") through First Chicago Capital Markets, Inc. as a Dealer ("PCCM" or the "Dealer"). The Chase Manhattan Bank in New York acts as Issuing Agent and Paying Agent with respect to the Notes.

The Notes are exempt from registration under the Securities Act of 1933 of the United States of America. The Notes will be placed with instinutional investors by the Dealer. The Notes have maturities of not more than 270 days, are issued in denominations of not less than \$100,000 and in increments of \$1,000 in excess thereof and are generally sold on a discounsed basis. Proceeds from the sale of the Notes will be used for current transactions.

CONTACT:

Edward G. Austin Managing Director (312) 732-7324

A subsidiars of FIRST CHICAGO NBD CORPORATION

# Florida Power Composition

Florida Power Corporation was incorporated in Florida in 1899 and is an operating public utility engaged in the generation, purchase, transmission, distribution and sale of electricity within Florida. The Company became a whelly owned subsidiary of Florida Progress Corporation in March 1982, as a result of a corporate restructuring.

The Company provided electric service during 1996 to an average of 1,300,000 customers, in a service area covering 20,000 square miles in central and northern Florida and along the west coast of the state. The service area includes the densely populated areas around Orlando, as well as the cities of St. Petersburg and Clearwater. Florida Power's 1996 earnings per share were up 5.7 percent over 1995 primarily due to continued customer growth. Residential customer growth of about 2 percent in 1996 continues to have the most significant effect on Florida Power's earnings growth. Contributing to Florida Power's 1996 earnings growth were lower interest and preferred dividend charges for 1996, compared with 1995. Lower debt balances resulting from improved cash flow and the redemption of preferred stock lowered these costs by \$10 million. Important industries in the territory include phosphase and rock mining and processing, electronics design and manufacturing, and citrus and other food processing. Other important commercial activities are tourism, health care, construction and agriculture.

# Book Lines

Florida Power's interim financing needs are funded primarily through its commercial paper program. The utility has a \$300 million 364-day revolving bank credit facility and a \$200 million five-year facility, which are used to back up commercial paper.

## General

This Offering Circular includes the following documents which are incorporated herein by reference: (i) the latest Annual Report on Form 10-K of the Company, as filed with the Securities and Exchange Commission, and (ii) every subsequent Quarterly Report of the Company on Form 10-Q or Current Report on Form 8-K as so filed. Copies of the foregoing documents are available from the Dealer, One First National Plaza, Mail Suite 0033, Chicago, Illinois 60670-0033, Amention: Edward G. Austin, telephone (312) 732-7324.

The attached Offering Memorandum was proposed from information provided by Florida Power Corporation. FCCM makes no representation or warranty, express or implied, as to the accuracy or completeness of any of the information contained in this Offering Memorandum or in any document for the time being incorporated herein by reference and FCCM accepts no responsibility for the accuracy or completeness thereof. This Offering Memorandum is not intended to provide the sole basis of any credit or other evaluation. A potential purchaser of the Notes should determine for itself the relevance of the information contained in the Offering Memorandum and its decision whether or not to purchase Notes should be based on such investigation as it doesns necessary.

This Offering Memorandum is for information purposes only and does not constitute an offer or invitation by or on behalf of Plerida Power Corporation, PCCM or any other person to tender for or subscribe or purchase any Notes. No person has been authorized to give any information or to make any representation not contained in this Offering Memorandum and, if given or made, any such information or representation must not be relied upon as having been authorized.

The First National Bank of Chicago and other banking affiliates of FCCM may participate on a regular basis in various general financing and banking transactions and services for Florida Power Corporation. Proceeds from the cales of the Notes may be used to repay indebtedness of Florida Power Corporation to any leading affiliate of FCCM.

FCCM, a wholly owned subsidiary of First Chicago NBD Corporation, is a separate entity from the lending affiliates. Unless otherwise disclosed, securities sold, offered or recommended by FCCM are not deposits.

are not insured by the Federal Deposit Insurance Corporation, are not guaranteed by any leading affiliate of FCCM, and are not otherwise an obligation or responsibility of any leading affiliate of FCCM.

	12/31/36	12/31/95
	(la	Millions)
ASSETS		
Property, Flast and Equipment:		1221212121
Electric utility plant in service and held for future use	\$5,965.6	\$5,867.5
Less: Accumulated depreciation	2,335.8	2,179.7 165.2
Accumulated decommissioning for nuclear plant Accumulated dismostlement for fossil plants	193.3	104.4
	3,316.9	3,418.2
Construction work in progress	140.3	131.8
Nuclear fast, not of amortisation of \$356.7 in 1996 and \$348.7 in 1994	39.9	59.1
Not electric utility plant	3,517.1	3,609.1
Other property, ast	13.3	23.0
	3.530.4	3.632.1
Current Ameli:		
Cash and equivalents	-	0.8
Accounts receivable, less reserve of \$4.1 in 1996 and \$5.2 in 1995 Inventories, at average cost:	174.7	200.7
Feel	47.2	40.8
Materials and supplies	95.4	101.3
Underrecovery of fast cost	82.6	0.3
Deferred insome taxes	35.6	32.3
Other	_6.2	
	441.7	380.1
Other Assets:		
Nuclear plant decommissioning fund	207.8	161.1
Unamortized debt expense, being amortized over term of debt	25.0	27.5
Other		- 84.1
Tead		272.7
Total Assets	\$4,264.0	\$4,284.9

	12/31/96	12/31/95
	(la Mi	llions)
CAPITALIZATION AND LIABILITIES		
Capitalization:		
Common stack	\$ 1,004.4	\$ 992.9
Revised comings	821.1	761.1
Cumulative professed stock without sinking funds	33.5	113.5
Cumulative preferred stack with sinking funds		25.0
Long-term debt	1.296.4 3.155.4	1.279.1 3.171.6
Current Liabilities:		
Accounts payable	115 5	89.8
Accounts payable to associated companies Customers' deposits	21.2 81.7	24.8 85.3
facome tease, currently payable	10.4	8.9
Accrued other tasse	10.0	12.3
Accrued interest	34.6	32.9
Other	47.3	65.1
	320.9	319.1
Notes payable	4.1	
Current portion of long-torm date	21.3	30.6
Total	<u>21.3</u> 346.3	349.7
Deferred Credits and Other Liabilities:		
Deferred income taxes	472.3	483.8
Unamortized investment tax credits	92.8	100.9
Other postrotiroment benefit costs	96.5	81.5
Other	_100.7	97.4
	762.3	763.6
Total Capitalization and Liabilities	\$4,264.0	\$4,284.9

	December 31,		
	1996	1996	1994
(In Millions)			
Operating Berenses	\$2,393.6	\$2,271.7	\$2,000.5
Operating Expresses:			
Operation: Fuel word in generation	409.7	431.3	425.6
Purchased power	531.6	436.5	294.6
Energy Conservation Cost Recovery	62.6	84.0	104.3
Operations and maintenance	413.4	393.7	412.2
Depreciation	324.2	293.7	261.5
Taxes, other then income tease	183.6	176.2	162.8
Income taxes	_135.8	_129.5	_114.7
Total Operating Exposures	2,060.9	1,944.9	1,775.7
Operating Service	_332.7	_326.8	_304.8
Other Income and Deductions:			
Allowance for equity funds used during			
coentraction	4.6	3.8	6.1 _(6.5)
Miscollessous other income, not	_ <u>(3.4)</u>	1.2	(0.4)
Interest Charges:	• • •	93.5	96.3
Interest on long-term debt	86.6	11.0	_12.1
Other innerest expense	98.4	104.5	108.4
Allowance for borrowed funds used			
during construction	_(2.9)	(3.5)	_(4.8)
Trail		101.0	103.6
Net Income	238.4	227.0	200.8
Dividends on Professed Stock	_11	<u>9.7</u>	10.1
Net Income After Dividends on			
Professed Stack	8.232.6	8 217.3	\$ 190.7

# EXHIBIT (d)-2

Merrill Lynch Money Markets Letter Agreement dated November 18, 1997

# November 18, 1997

This letter (the "Amendment") will amend the letter agreement (the "Agreement") dated December 23, 1985, between Merrill Lynch Money Markets, Inc. ("MLMMI") and Florida Power Corporation ("Company").

The maximum amount of notes to be outstanding at any one time, as stipulated in paragraph one of the Agreement, is amended to be \$500 million.

If the amendment is in accordance with your understanding of this amendment, please sign and return to us a counterpart hereof, where upon this Amendment, along with the Agreement and all counterparts will become a binding agreement among the parties hereto in accordance with its terms.

Very truly yours,

FLORIDA POWER CORPORATION

Authorized Signatory

MERRILL LYNCH MONEY MARKETS, INC.

Vice President

# EXHIBIT (d)-4

First Chicago Letter Agreement dated December 4, 1997



December 4, 1997

Mr. Ed Austin
First Chicago Trust Company of New York
c/o The First National Bank of Chicago
One North State Street
Suite 0126, 9th Floor
Chicago, IL 60602

Dear Mr. Austin:

This letter is to confirm the following increase in the commercial paper program of Florida Power Corporation.

On August 21, 1997, the Board of Directors of Florida Power Corporation authorized an increase in its commercial paper program from \$400 million to \$500 million. The Company will increase its program to \$500 million and will maintain a \$500 million line of credit to back up the commercial paper.

Enclosed is a list of banks participating in this line and a certified copy of the Resolutions of the August Board meeting.

If you have any questions regarding these changes, please feel free to contact me.

Sincerely.

K. E. McDonald

KEMJak

Enclosures

k westery ben gratery fronts in